

UNIVERSITY OF NAIROBI

DEPARTMENT OF EARTH AND CLIMATE SCIENCES

IMPACT OF CLIMATE CHANGE AND LAND USE/LAND COVER DYNAMICS ON SMALL-SCALE FARMING SYSTEM IN GEDAREF STATE, SUDAN

By

MAYSOON ABDELHAMEID ABDELHAI OSMAN

(185/53732/2018)

A thesis submitted in partial fulfilment of the requirements for the degree of Doctor of Philosophy in Climate Change and Adaptation in the department of Earth and Climate Sciences of the University of Nairobi

June 2024

PLAGIARISM STATEMENT

I, Maysoon Abdelhameid Abdelhai Osman, declare that this thesis is my original work and has not been previously submitted by me or any other person for a degree to this or any other tertiary institution.

Signature:_____ Date: ...29/06/2024.....

Name of Student: Maysoon Abdelhameid Abdelhai Osman

This thesis has been submitted to the department of Earth and Climate Sciences in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Climate Change and Adaptation of the University of Nairobi with our approval as supervisors.

Signature:Date: 29/6/2024.....

Dr. Joshua Orungo Onono

Department of Public Health, Pharmacology and Toxicology, University of Nairobi; P.O Box 30197-00100, Nairobi, Kenya



....Date: 29/6/2024.....

Dr. Muna Mohamed Elhag

Associate professor, Water Management and Irrigation Institute University of Gezira; P.O. Box 20, Wad Medani, Sudan

Dr. Lydia Olaka

Lecturer, Department of Geoscience and the Environment, The Technical University of Kenya P.O. Box 52428- 00200, Nairobi, Kenya

DEDICATION

I dedicate this achievement to my beloved mother, late father, brothers, and sisters, who have always supported and loved me unconditionally. Above all, I thank Almighty Allah, the source of my strength throughout this journey.

ACKNOWLEDGEMENTS

I am very grateful to my supervisors, Dr. Joshua Orungo Onono, Dr. Lydia Olaka of the University of Nairobi, Dr. Muna Mohamed Elhag of the University of Gezira, and Dr. Elfatih M. Abdel-Rahman of the International Centre of Insect Physiology and Ecology (*icipe*) for supervising this project, encouragement, support and exceptional academic guidance throughout my study. Special thanks to Dr. Elfatih M. Abdel-Rahman; words can hardly describe my thanks and appreciation to you. You have been my source of inspiration, support, and guidance. You have taught me to be unique, determined, to believe in myself, and to always persevere.

I am greatly indebted to the Regional Universities Forum for Capacity Building in Agriculture (RUFORUM), University of Nairobi and University of Gezira for the PhD scholarship. I also highly acknowledge the International Centre of Insect Physiology and Ecology (*icipe*). More specifically, I am grateful to Dr. Henri E. Z. Tonnang, the head of the Data Management, Modelling, and Geo-Information (DMMG) Unit, for his valuable advice and continuous support, without it, this study would not have been possible. I am also grateful to Mawazo Institute for the scarce skills fellowship and the financial support I received. I also acknowledge various institutions that provided data for the study.

My sincere appreciation goes to Tasneem Moawia Abdalla Osman - friend, 'sister,' 'co- traveller' and colleague who continuously provided encouragement and was always willing and enthusiastic to assist in any way she could throughout the research project. I would also like to express my thanks, gratitude and appreciation to my lifelong friend and brother, Dr. Abdelmutalab Gesmalla Ahmed Azrag, who never-failing encouragement and was always there for me, even on the tough days, supporting and offering deep insight into this study. Lastly, I would like to thank my family,

my friend Tasabeh Babiker Ahmed Dafa Allah, colleagues at *icipe* and University of Nairobi, and Dr. Samira Abuelgasim Mohamed for their unwavering moral support during my study.

ABSTRACT

Rainfed agriculture is generally the principal activity for the economy of Sudan. However, this sector is severely affected by climate change due to shifting in rainfall patterns, rising temperatures, and extreme weather events such as floods and droughts. Consequently, this has resulted in reduced crop production and increased hunger. Specifically, climate change coupled with changes in land use/ land cover (LULC), as a proxy for landscape structure, land fragmentation, degradation, and reduction in soil fertility have threatened small-holder livelihood in Sudan. Therefore, the aim of this study was to investigate the impacts of climate trends and LULC changes on crop production of small-holder farmers in Gedaref state, Sudan. To achieve this, four objectives were defined; i) to determine the relationship between climate trends and the level of crop yields; ii) to determine farmers' perception of climate variability and change and their choice of adaptation measures; iii) to assess and quantify LULC changes and their intensities between 1988 and 2018, and to project LULC structure in 2028 and 2048; iv) to assess the local farmers' perception of LULC change trends; and determine their drivers in Gedaref state. Historical records of rainfall and temperature, crop (sorghum, sesame, millet, cotton and sunflower) production, and population data for Gedaref state were obtained from Sudan Meteorological Authority, Ministry of Agriculture, and Sudan Central Bureau of Statistics, respectively. Semi-structured questionnaire, key informant interviews and focus group discussions were used to collect data on small-holder farmers' perception of climate and LULC trends. These datasets were subjected to various analysis including Mann-Kendall trend test, multiple linear regression, correlation and multinomial regression model. In addition, satellite-based data were used to map LULC changes between 1984 and 2018 using the machine learning random forest algorithm and future LULC for 2028 and 2048 utilizing the cellular automata-artificial neural network model. Trends in annual maximum and minimum temperatures significantly (p < 0.0001) increased in Gedaref state by 0.03°C and 0.05°C per year, respectively, with fluctuations in the amount of rainfall. The rainfall amount and duration of the rainy seasons were the only climatic factors that positively affected crop yields. The small-scale farmers were aware about climate change, and they used crop rotation, early cultivation and cultivation of short-maturing crop varieties as adaptation measures to climate change and variability. The analysis of LULC change showed a decline in forest and grassland cover and an increase in cropland and settlement. Local

land users perceived a similar trend and ranked firewood collection, agricultural expansion, and charcoal production as the top proximate drivers, while poverty and rapid population growth were the most vital drivers for LULC changes in Gedaref landscape structure. However, future prediction of LULC showed an overall increase in cropland and settlement areas at the expense of forest and grassland areas by 2048. These findings have shown that crop yields are affected by climate change and variability and changes in LULC would significantly affect future crop production. Therefore, there is a need to raise awareness among different stakeholders, especially policymakers to provide sustainable interventions for small-holder farmers against climate and LULC changes in Gedaref state and other similar farming systems.

Table of Contents

DEDICATION	II
ACKNOWLEDGEMENTS	III
ABSTRACT	V
LIST OF TABLES	XI
LIST OF FIGURES	XIII
LIST OF ABBREVIATIONS	XVI
CHAPTER ONE	1
INTRODUCTION	1
1.1. Background	1
1.2. Problem statement	4
1.3. Justification	6
1.4. Objective of the study	7
1.4.1. General objective	7
1.4.2. Specific Objectives	7
1.5. Research questions	7
1.6. Scope and limitations of the study	8
1.6.1. Scope	8
1.6.2. Limitations	8
1.7 Structure of the thesis	9
CHAPTER TWO	10
LITERATURE REVIEW	
2.1 Introduction	10
2.2 Empirical review of farming system in Sudan	10
2.3 Empirical review on climate variability and change	13
2.4 Impact of climate trends on crop yields	15
2.5 Farmers' perception of climate change and determinants of adaptation measures	17
2.6 Land use and land cover change	19
2.7 Land use and land cover mapping	21

2.8 Land use and land cover change in Sudan2	2
2.9 Summary	5
CHAPTER THREE	7
STUDY AREA AND METHODS	7
3.1 Introduction	7
3.2 Location	7
3.2.1 Population in Gedaref state2	9
3.3 Biophysical setting in Gedaref state2	9
3.3.1 Climate	9
3.3.2 Soil types in Gedaref state3	0
3.3.3 Agriculture as a livelihood source in Gedaref state3	0
3.3.4 Land use/ land cover (LULC) in Gedaref state3	1
3.4 Conceptual framework for the study	1
3.5 Methods	3
3.5.1 The study design	3
3.5.2 Determination of the relationship between climate trends and the level of crop yields3	3
<i>3.5.3 Determination of farmers' perception of climate variability and change and the choice of adaptation measures.</i>	2
3.5.4 Assess and quantify LULC changes and their intensities for 30 years (1988-2018) and project LULC in 2028 and 2048	4
3.5.5 Assess the local farmers' perception of land use/ land cover (LULC) change trends; and determine LULC drivers in Gedaref state5	5
CHAPTER FOUR	0
CLIMATE VARIABILITY AND CHANGE AFFECT CROPS YIELD UNDER RAINFEI)
CONDITIONS: A CASE STUDY IN GEDAREF STATE, SUDAN	0
4.1 Introduction	0
4.2. Results	0
4.2.1. Estimation of annual and seasonal temperature trends in Gedaref state	0
4.2.2. Determination of annual rainfall trends in Gedaref state	2
4.2.3. Assessment of annual crop yield trends in Gedaref state6	5
4.2.4. Estimation of temperature and rainfall variability indices	6
4.2.5. Characteristics of rainy seasons in Gedaref state6	8
4.2.6. Assessment of the relationship between climatic variables and crop yield in Gedaref7	1

4.3. Discussion	74
4.4. Conclusions	79
4.5. Summary	80
CHAPTER FIVE	
SMALL-HOLDER FARMERS' PERCEPTIONS OF CLIMATE VARIABILITY AN OBSERVED CLIMATE TRENDS AND ADAPTATION MEASURES IN GEDAR	ND CHANGE, EF STATE,
SUDAN	
5.1. Introduction	
5.2. Results	82
5.2.1. Demographic and socio-economic characteristics of small-holder farmers	82
5.2.2. Small-holder farmers' perception of climate variability and change	84
5.2.3. Access and source of climate information	87
5.2.4. Trends in climatic variables (rainfall and temperature) of the metrological s	station88
5.2.5. Small-holder farmers' adaptation measures	90
5.2.6. Factors influencing small-holder farmers' choice of adaptation measures	91
5.3. Discussion	94
5.4 Conclusions	
CHAPTER SIX	
MAPPING, INTENSITIES AND FUTURE PREDICTION OF LAND USE/	LAND COVER
DYNAMICS USING GOOGLE EARTH ENGINE AND CA- ARTIFIC	CIAL NEURAL
NETWORK MODEL	
6.1 Introduction	101
6.2 Results	
6.2 L and use/land cover (LULC) classification	
6.2.2 Accuracy of land use/land cover (LULC) classification	
6.2.2 Land use/land source (LULC) shares detection	
6.2.5 Latanzita anglazia in lan daga (lan dagana (LULC) tangaiti ang	
6.2.5 Intensity analysis in land use/ land cover (LULC) transitions	
6.2.6 Future prediction of land use/ land cover (LULC)	
6.3 Discussion	
6.4 Conclusions	
6.5 Summary	
CHAPTER SEVEN	

INTEGRATING SATELLITE REMOTE SENSING DATA AND SMALL-SCALE
FARMERS' PERCEPTIONS TO DETERMINE LAND USE/ LAND COVER CHANGES
AND THEIR DRIVING FACTORS IN GEDAREF STATE, SUDAN 123
7.1. Introduction
7.2 Results
7.2.1 Land use/ land (LULC) maps and trends
7.2.2 Socio-economic and demographic characteristics of sampled households
7.2.3 Perception of small-holder farmers of the drivers of land use / land cover (LULC) change in Gedaref state, Sudan
7.2.4 Multinomial logistic regression results of perceived drivers of land use/ land cover (LULC) changes
7.2.5 Comparison between small-holder farmers' perception and land use / land cover (LULC) classification
7.2.6 The actual trend of cultivated area, crop yield and production135
7.2.7 Relationship between climatic variables, population and land use/ land cover (LULC) classes
7.3 Discussion
7.4 Conclusions147
7.5 Summary
CHAPTER EIGHT 151
SYNTHESIS, GENERAL CONCLUSIONS AND RECOMMENDATIONS 151
8.1 Introduction151
8.2 Synthesis
8.3 Conclusions154
8.4 Recommendations
References 158
Appendices

LIST OF TABLES

Table 3.1: Mathematical symbols used to calculate different intensities as illustrated in equations
3-7 as described by Aldwaik and Pontius (2012)52
Table 4.1: Estimated Sen's slope values for the annual and seasonal temperature (°C) trends at
Gedaref state, Sudan between 1984 and 2018 61
Table 4.2: Estimated Sen's slope values for annual rainfall (mm) trends reported at different
stations in Gedaref State, Sudan between 1980 and 201863
Table 4.3: Estimated slope of linear regression for the annual yield trends for five crops (sesame,
sorghum, cotton, millet, and sunflower) in Gedaref state, Sudan from 1970–201865
Table 4.4: Estimated length of the rainy season (1984–2018) for Gedaref State, Sudan69
Table 4.5: Multiple linear regression terms and R^2 values for estimating yield, as a function of
climatic variables, of five major crops grown in Gedaref state, Sudan73
Table 5.1: Socio-economic characteristics of surveyed households in the study area ($N = 400$) 83
Table 5.2: Estimated Sen's slope values for annual rainfall (mm) and mean annual temperature
(°C) trends in Gedaref State, Sudan, between 1984 and 201889
Table 5.3: Adaptation measures used by the respondents to adapt to climate change $(N = 400)-90$
Table 5.4: Multinomial logistic regression estimates for the factors that influencing the choice of
adaptation measures by small-holder farmers in Gedaref state92
Table 6.1: Area (ha) and percent cover (%) of each land use/ cover (LULC) class in Gedaref state
for the years 1988, 1998, 2008, and 2018 estimated using multi-date Landsat images and random
forest classification algorithm 103
Table 6.2: Overall and individual class accuracies of land use/ land cover (LULC) maps of Gedaref
state for the years 1988, 1998, 2008 and 2018 103

Table 6.3: Land use/ land cover (LULC) change estimates (area and percentage) for Gedaref state
for the 1988–1998, 1998–2008, 2008–2018 and 1988–2018 105
Table 6.4: The proportion of predicted land use/ land cover (LULC) categories at Gedaref state
112
Table 7.1: Summary of household characteristics for small holders farming communities in
Gedaref state, Sudan (N = 400) 127
Table 7.2: Perceived proximate drivers of land use/ land cover (LULC) changes by local land
users in Gedaref state, Sudan 129
Table 7.3: Perceived underlying drivers of land use/ land cover (LULC) changes in Gedaref state,
Sudan130
Table 7.4: Significant ($p \le 0.05$) socio-economic factors influencing respondents' perceptions on
the drivers of land use/ land (LULC) cover changes in Gedaref state, Sudan 131
Table 7.5: Estimated Sen's slope values for cropland, forest, and settlement (ha) trends in Gedaref
State, Sudan between 1988 and 2018 135
Table 7.6: Estimated Sen's slope values for culitvated area of sorghum, sesame, millet, sunflower
and cotton (ha) trends in Gedaref State, Sudan between 1988 and 2018 137
Table 7.7: Estimated slope of linear regression for the annual crop production trends for five crops
(sesame, sorghum, cotton, millet, and sunflower) grown in Gedaref state, Sudan from 1988–2018.
139
Table 7.8: The multiple linear regression terms and coefficients of determination (R^2) estimating
the change land use/ land cover (LULC) areas as a function of rainfall, temperature, and population
in Gedaref state, Sudan 142

LIST OF FIGURES

Figure 3.1: Location of Gedaref State in Sudan (Source: Author) -----28

Figure 3.2: Boxplot showing the monthly rainfall in Gedaref state, Sudan between 1984 2018.
The red arrows show the main crop cultivation period. Median rainfall values are shown by the
bold lines within the boxes and the circles show outliers30
Figure 3.3: Conceptual framework of the study32
Figure 3.4: Histogram, normal distribution, and boxplot fitted for minimum (Tmin) and maximum
(Tmax) temperatures (1984-2018) and rainfall (1980-2018) data obtained from the Gedaref
meteorological station, Sudan. Circles in some boxplots represent individual outlier observations.
Figure 3.5: Histogram, normal distribution, and boxplot fitted for crop yield data obtained from
the Ministry of Agriculture, Gedaref, Sudan36
Figure 3.6: Methodological flowchart for the land use/ land cover (LULC) mapping, transition,
intensity and future prediction46
Figure 3.7: Predictor variables used for land use/ land cover (LULC) future predictions55
Figure 4.1: Annual and seasonal temperature trends in Gedaref state, Sudan between 1984 and
2018. (A): maximum temperature, (B): minimum temperature, and (C): diurnal temperature range.
Figure 4.2: Annual rainfall trends in five locations in Gedaref state, Sudan between 1980 and
2018. (A): Gedaref, (B): El gadabalea, (C): Am Senat, (D): Samsam, (E): El hawata, and (F):
Average rainfall for all five locations64
Figure 4.3: Annual crop yield trends for the major crops grown in Gedaref State, Sudan from
1970–2018, (A): sorghum, (B): sesame (C): cotton, (D): millet and (E): sunflower yield trends.
66

Figure 4.4: Standardized anomaly	index for mean	temperature (A),	and annual	rainfall (B) in
Gedaref State, Sudan from 1984–20	18			68

Figure 4.5: Relationships between the length of the rainy season and yield of five major crops grown in Gedaref state, Sudan. R² is coefficient of determination. -----71 **Figure 4.6:** Associations, as assessed by a Pearson's correlation analysis, among climatic variables and yield of five major crops grown in Gedaref State, Sudan. r is correlation coefficient.----72

Figure 5.1: Farmers' perception of climate change85
Figure 5.2: Indicators that support farmers' perceptions of fluctuations in rainfall85
Figure 5.3: Indicators that support farmers' perceptions of increasing temperatures in summer and
winter seasons87
Figure 5.4: A) small-holder farmers' access to climate information and B) source that farmers
used to access climate information88
Figure 5.5: Annual rainfall and mean temperature trends in Gedaref state, Sudan, between 1984
and 201889
Figure 6.1: Classified Land use/ land cover (LULC) maps of Gedaref state for the years 1988,
1998, 2008 and 2018 produced using multi-date Landsat images and random forest classification
algorithm102
Figure 6.2: Land use/ land cover (LULC) transitions of Gedaref state during 1988–2018, 1998–
2018 and 2008–2018 106
Figure 6.3: Interval level intensity of land use/ land cover (LULC) change for 1988 1998, 1998–
2008 and 2008–2018. A) the percent of the area that changed over each interval and B) the
percentage of the area that annually changed during each interval 107
Figure 6.4: Land use/ land cover (LULC) category level intensity for 1988–1998, 1998–2008,
2008 and 2018 108
Figure 6.5: Transition level intensity of land use/ land cover (LULC) for the period 1988–1998.
(A) cropland, (B) forest, (C) grassland, (D) water and (E) settlement (gains on the right and losses
on the left) 110
Figure 6.6: Transition level intensity of land use/ land cover (LULC) for the period 1998–2008.
(A) cropland, (B) forest, (C) grassland, (D) water and (E) settlement (gains on the right and losses
on the left 110

Figure 6.7: Transition level intensity of land use/ land cover (LULC) for the period 2008–2018.
(A) cropland, (B) forest, (C) grassland, (D) water, and (E) settlement (gains on the right and losses
on the left) 111
Figure 6.8: Predicted land use/ land cover (LULC) in Gadaref state for 2028 and 2048 112
Figure 7.1: Land use/ land cover change trend (A) and rate (B) of Gedaref state, Sudan 125
Figure 7.2: Net change in land use/ land cover (LULC) classes of Gedaref state, Sudan between
1988 and 2018 126
Figure 7.3: Farmers' perception towards changes in (A) cropland area; (B) forest area; (C)
settlement area; and (D) crop production in Gedaref state, Sudan 134
Figure 7.4: Satellite-based land use/ land cover change trends in Gedaref state, Sudan between
1988 and 2018. (A) cropland; (B) forest; and (C) settlement 134
Figure 7.5: Annual actual cultivated area trends for sorghum, sesame, millet, sunflower and cotton
in Gedaref state, Sudan between 1988 and 2018 136
Figure 7.6: Annual crop production trends for sorghum, sesame, millet, sunflower and cotton in
Gedaref State, Sudan from 1988–2018 138

Figure 7.7: Pearson's correlation assessed the relationship between cultivated area (ha) and yield of sorghum, sesame, millet, sunflower and cotton in Gedaref state, Sudan. The dashed lines above and below the trend line are the confidence intervals. ------ 140

Figure 7.8: Pearson's correlation analysis showing the relationship between: (A) cropland area and rainfall; (B) cropland area and temperature; (C) cropland area and population; (D) forest area and rainfall; (E) forest area and temperature; (F) forest area and population; (G) settlement area and population. The shaded lines above and below the trend line are the confidence intervals 142

LIST OF ABBREVIATIONS

AD	Allocation Disagreement
ANN	Artificial Neural Network
ASTER	Advanced Spaceborne Thermal Emission and Reflection Radiometer
CA	Cellular Automata
CA-ANN	Cellular Automata-Artificial Neural Network Model
CV	Coefficient of Variation
DSSAT	Decision Support System for Agrotechnology Transfer
DTR	Diurnal Temperature Range
ENVI	Environment for Visualising Images
ERDAS	Earth Resources Data Analysis System
ETM+	Enhanced Thematic Mapper Plus
FAO	Food and Agriculture Organization
FGDs	Focus group discussions
GDP	Gross Domestic Product
GEE	Google Earth Engine
GMT	Greenwich Mean Time
ENVI	Environment for Visualising Images
IPCC	Intergovernmental Panel on Climate Change
LULC	Land Use and Land Cover
ML	Maximum Likelihood Classifier
NGOs	Non-Governmental Organisation

OA	Overall Accuracy
OLI	OLI Operational Land Imager
PA	Producer's Accuracy
QD	Quantity Disagreement
RF	Random Forest
SPSS	Statistical Package for Social Sciences
TD	Transdisciplinarity Approach
ТМ	Thematic Mapper

CHAPTER ONE

INTRODUCTION

1.1. Background

In the last century, the global air temperature has risen by 2–6°C due to greenhouse gas emissions, resulting in what is known as climate change (IPCC, 2014). With the continuous increase of greenhouse gases in the Earth's atmosphere, it is estimated that by the end of the twenty-first century the mean atmospheric temperature will increase by 1.4 to 5.8°C (IPCC, 2014). This temperature increase is expected to significantly affect crop production, resulting in food insecurity, particularly in developing countries (Zewdie, 2014). Therefore, climate change is rapidly becoming one of the most threatening and toughest global challenges.

Changing temperatures, shifting rainfall, sea-level rise, and increased experience of extreme climate events will highly decrease global food production in this century, unless an effort is made to address these adverse effects of climatic change (Ewert *et al.*, 2015). Furthermore, it has been projected that by 2050, the world will require to raise agriculture production by 60–110% to meet the ever growing human population and increasing demand for food (Ray *et al.*, 2013). Consequently, this will also exert pressure on available natural resources such as water, energy, and land. Therefore, increasing agricultural production in the near future will be one of the biggest challenges facing humanity (Godfray *et al.*, 2010; Licker *et al.*, 2010). Water availability for agricultural production is a critical factor of the many dimensions of this challenge (Fereres *et al.*, 2011). Although it is argued that agricultural output needs to be increased to meet the growing demand for food and to ensure the sustainability of natural resources, agricultural expansion is not

a choice, but rather a strategy that need to be considered (i.e., increasing crop efficiency and closing yield gaps on underperforming land) (Elagib *et al.*, 2019).

Besides the impact of climatic change, human activities also influence ecosystem services through continuous land use/ land cover (LULC) changes. The combination of human population increase and competitive land usage due to climate change results in land scarcity, conversion of natural areas to agricultural lands, and other uses such as settlements (Kanianska, 2016). Indeed, humans have modified approximately one-third to half of the Earth's surface, and the extent of land use is expected to grow to meet the increasing demand for land, especially for agricultural production (Brovkin *et al.*, 2013; Ellis, 2011). Increased agricultural intensity puts a strain not just on land resources but also on the whole ecosystem. Accordingly, many scientists have stressed the significance of incorporating LULC studies into climate change research (de Chazal & Rounsevell, 2009; Mahmood *et al.*, 2010).

In Africa, millions of people depend on agricultural activities; thus, land is considered the most critical livelihood source. However, climate change has become a concerning issue due to its effects on agricultural production and land use, threatening livelihoods and food security (Akudugu *et al.*, 2012). Sub-Saharan Africa is one of the poverty-stricken areas facing considerable variation and decrease in food production due to climate variability and change (Schmidhuber & Tubiello, 2007). Yet, agricultural production is still tiny compared to the tremendous growth in agricultural output in many regions of the world over the past decades (Pretty *et al.*, 2011). Therefore, to meet the high demand for food in changing climate, the agricultural area has been dramatically expanded in sub-Saharan Africa (Kleemann *et al.*, 2017). This expansion led to the loss of bio(geo)diversity, severe erosion and land degradation, soil infertility and deforestation (Akinyemi & Speranza, 2022). On the other hand, the increase in the

human population as well as other non-agricultural activities and urbanization, has resulted in a loss of productive agricultural lands in sub-Saharan Africa (Creutzig *et al.*, 2019). The rapid increase in human population coupled with a dramatic change in LULC, and this might aggravate climate change impact in the future.

In Sudan, agriculture remains a vital sector of the economy as the primary source of food, raw materials, income, and foreign exchange (Mahgoub, 2014). It supports livelihood for about 60% to 80% of the population in the country (Hussein *et al.*, 2022). It also plays a vital role in economic growth through the industrial sector and trade, thus, it contributes 27% of the country's Gross Domestic Product (GDP) (Hussein et al., 2022). The agricultural sector in Sudan is split into two sub-sectors: irrigated and rainfed agriculture. The rainfed sector is the most important in Sudan and it occupies around 90% of the total cultivated land in the country and provides livelihoods to most of the population, especially in the rural areas. However, Sudan is highly vulnerable to climate variability and change, as like other African countries. This situation is exacerbated by several pressures at different levels, such as poverty, institutional deficiencies, and restricted access to finance, market and infrastructure. Indeed, in the last decades, the country has witnessed rainfall fluctuations, rising temperatures, increasing floods, and recurrent droughts, which highly affected agricultural production (Siddig et al., 2020). This has affected the livelihood of millions of people in rural areas as the substantial rainfed cultivation is the primary source of their livelihood and income.

Gedaref state is the most important and largest rainfed agricultural area in Sudan, located in the eastern part of the country (Yousif & Babiker, 2015). This state is characterized by a semi-arid climate with medium to low rainfall and high temperature, which are the most critical direct factors to agricultural production (Mohmmed *et al.*, 2018). The rainfed sector in Gedaref State is divided

into semi-mechanized farming and traditional farming, which is practiced only by small-holder farmers, who largely depend on farming activities for their livelihoods and contribute much to agricultural outputs (Ayoub, 1999). Degraded soils, erratic climate conditions, and weak adaptive capability enhances the vulnerability of small-scale farmers to climate variability and change (Mohmmed *et al.*, 2018). Furthermore, LULC changes at local scales have a major role in climate variability and change (Deng *et al.*, 2013). Therefore, LULC change analysis at the local scale over an extended timescale helps to reveal important basics that can explain future projections of new LULC changes (Sulieman & Elagib, 2012).

1.2. Problem statement

In Sudan, the agriculture sector remains the mainstay of the country's economy, where rainfed agriculture is the most important sector. However, climate change, evidently seen by rising temperatures and erratic rainfall in the country has significantly affected the agricultural sector, especially rainfed agriculture (Ayoub, 1999; Elagib & Elhag, 2011; Eldredge *et al.*, 1988; Siddig *et al.*, 2020). Given the projection that the population of Sudan (approximately 43 million people) will be doubled by the year 2050 (Nath *et al.*, 2001), hence the agricultural production, especially in rainfed areas such as Gedaraef state, needs to be doubled as well to ensure food and nutrition security. This increment will create pressure on available natural resources to produce sufficient agricultural output to meet the rising demand for food. Indeed, Gedaref state is the main rainfed agricultural area in Sudan, where thousands of people practice traditional small-holding farming, which their livelihoods mainly depend on. However, many farmers in this state have experienced different climate change effects over the last decades, which sometimes appear as increased intensity and recurrence of extreme climate events like floods, droughts, long dry spells, and

unpredictable onset of the rainy season. As a result, agricultural productivity has been largely affected. Nevertheless, there is a lack of science-led evidence and assessment of climate effects on crop yield in the Gedaref region. Understanding the responses of crops to the shifts in inter-and intra-seasonal rainfall patterns is useful for agricultural planning and designing adaptation measures. Furthermore, how small-holder farmers in Gedaref state perceive climate and LULC changes and what adaptation measures they use to cope with the negative impacts of these changes are questions that have not yet been evaluated.

The immigration of many people within the country as well as from the neighbouring countries to Gedaref has led to expansion of agricultural land and settlements, affecting other natural resources. Indeed, Gedaref area has witnessed a remarkable large-scale land degradation indicated by reduced vegetation coverage, and loss of soil fertility (Glover & Elsiddig, 2012). This has aggravated the negative impacts of climate variability and change on rainfed farming (Morgan, 2019), coupled with LULC changes, which are triggered mainly by malpractices such as the removal of a large area of natural forest by people who engages in charcoal burning and illegal tree cutting and wood harvesting (Abdalla, 2015). However, LULC changes and factors that drive these changes for the whole Gedaref state are still largely unknown. Therefore, the emerging challenges of climate variability and change and LULC changes should not be considered independently; instead, their integrated effect should be evaluated in a holistic manner to support policy mainstreaming on mitigation of climate impacts. Hence, more attention on the impact of climate and LULC changes on the traditional rainfed agricultural sector is needed to minimize the negative impacts and achieve food and nutrition security in the Gedaref area and Sudan at large.

1.3. Justification

In Sudan, agriculture is one of the significant pillars of the country's economy. Notwithstanding, in recent years, rainfed agriculture has been affected significantly by climate variability and change and land degradation. This might result in a reduction in agricultural production by reducing suitable agricultural areas and quantities. In the country, one of the main states that rely on agriculture as the main the agriculture activity is Gadaref state. Agriculture in the state provides employment and livelihood for about 80% of the population (Mahgoub, 2014). Although Gadaref state is the main rainfed area in Sudan that is characterised by a semi-arid climate, little attention has been paid to assessing the effect of climate change on small-scale crop farming. It is speculated that fluctuation in the timing and duration of the rainy season could have an important negative influence on crop yield (Murenzi, 2019). Therefore, improved understanding of climate change impacts and land use changes can lead to the development of better farming practices. Also, introduction of climate-resilient crops and efficient land use techniques can increase yield and income for small-scale farmers. Without addressing climate change impacts and land use dynamics, agricultural productivity may decline due to increased vulnerability to extreme weather events and land degradation. Farmers may also continue using unsustainable practices, leading to further soil depletion and lower yields. Therefore, this study will provide much-needed information for decision-makers, extension officers, researchers, non-governmental organizations (NGOs), and fiscal development planners to adopt measures that enable reduction of the vulnerability of small-scale farmers to climate change and variability and LULC changes.

1.4. Objective of the study

1.4.1. General objective

The general objective of this study was to investigate the impacts of climate trends and variability and LULC changes on small-holder farmers' crop production in Gedaref State, Sudan.

1.4.2. Specific Objectives

The specific objectives of this study were to:

1. Determine the relationship between climate trends and the level of crop yields,

2. Determine farmers' perception of climate variability and change and choice of adaptation measures,

3. Assess and quantify the LULC changes and their intensities between the years 1988 to 2018 and forecast future LULC outlook in 2028 and 2048, and

4. Evaluate the local farmers' perception of LULC change trends; and determine the main drivers of such changes.

1.5. Research questions

1) What are the climate trends in Gedaref state?

2) What are crop farmers' perceptions on LULC changes in Gedaref state?

3) What is the relationship between climate trends and crop yields in Gedaref state?

4) What is the trend of LULC changes in Gedaref state over the last thirty years?

5) What are the factors that trigger LULC changes in Gedaref state?

6) What are the factors that influence farmers' choices of adaptation measures for climate change?

1.6. Scope and limitations of the study

1.6.1. Scope

Gedaref state in Sudan was selected for this study due to its importance as Sudan's largest rainfed agricultural area, with a few large and many small-scale farmers. It is largely recognized as the country's land of sorghum and sesame, where about one-third of these two crops are produced under a rainfed mechanized farming system (Elagib *et al.*, 2019). Other crops such as cotton, groundnut, millet, and sunflower are also cultivated in the area under rainfed conditions. Sorghum and millet are grown for food consumption, with sorghum being the staple food in the entire country. Sesame and sunflower are grown mainly for oil production and export, while cotton is an industrial cash crop. However, climate variability and change may significantly impact crops yield and, consequently, livelihoods, especially for small-holder farmers and the entire state's economy. Additionally, LULC changes have aggravated the effects of climate variability and change on rainfed farming (Morgan, 2019) in Gedaref state. Therefore, food security and sustainable livelihoods are in danger for the highly vulnerable groups whose livelihoods depend largely on natural resources.

1.6.2. Limitations

The temperature data used for this study were obtained for one meteorological station in Gedaref state. This is due to the fact that Sudan meteorological authority from which these data were obtained has only one meteorological station in Gedaref state. In addition, the limited resources that could allow collection of ground LULC observations to train and test the models developed for detecting LULC changes and their intensities. Hence, on-screen reference LULC data was collected from the Google Earth Pro platform and used for training and testing the LULC models.

1.7 Structure of the thesis

This thesis contains eight chapters. The first chapter provides the background information about the study. The second chapter presents a detailed state-of-the-arts in relation to LULC and climate changes and their impact on crop production. The chapter also provides information in relation to farmers' perceptions of LULC and climate changes and their adaptation measures. A synthesis of the state-of-the-arts that shows the gaps in knowledge that needs to be filled was also highlighted. Chapter three describes the study area and provides a detailed methodological framework that employed to address each of the study objective. Chapter four provides results and discussion on climate and crop yield trends and the impact of climate variability and change on the main crops yield in Gedaref state. Chapter five provides results and discussion on farmers' perceptions of climate variability and change in Gedaref state, and the adaptation measures they use at the local scale to cope with the negative impact of climate change on crop production. Chapter six presents results and discussion on LULC dynamics and their intensities and future prediction of LULC outlook in Gedaref state. Chapter seven provides results and discussions on farmers' perception of LULC dynamics and the approximate and underlining drives that caused the changes in the landscape structure. Chapter eight presents the general conclusions for various objectives of the study and list of recommendations for further research, policy change and support for extension services.

CHAPTER TWO

LITERATURE REVIEW

2.1 Introduction

This chapter explores a relevant literature review on climate variability and change and its impact on crop yields, farmers' perception of climate change and adaptation measures they use to cope with the impacts of climate change in Sudan. It also reviews literature on land use/ land cover change and their driving factors.

2.2 Empirical review of farming system in Sudan

Farming systems face complex, dynamic, and interconnected changes in the production context as a result of climate change, rising food demand, scarcity of natural resources, volatile input and output prices, rising energy costs, and administrative regulation (Martin *et al.*, 2013). Farming systems differ from one location to another depending on many factors such as climatic conditions, water availability, level of farming and soil type. However, among these, water availability is the most important factor that plays a crucial role in determining the type of farming system in a specific location. Three farming systems namely rainfed, irrigation and supplementary irrigation are used in different parts of the world depending on the water supply for crop production (Pathak *et al.*, 2009). In Africa however, most of the farmers depend only on the amount of rainfall to cultivate their crops during the rainy season. Although farmers use different cropping systems such as maize mixed farming, agro-pastoral farming, cereal-root crop mixed farming, among others to ensure food security (Garrity *et al.*, 2012), the crop yields under rainfed conditions remain 50% lower than yields gained from the irrigated system (Jaramillo *et al.*, 2020). This is mainly due to

the poor infrastructure in Africa, uncertainty in the amount of rainfall and the lower supply of solar energy during the rainy season that limits photosynthesis, leading yield reduction (Jaramillo *et al.*, 2020). Indeed, the low crop yield under rainfed conditions is the key factor that causes hunger, unemployment, poverty, and migration to cities in developing countries (Jaramillo *et al.*, 2020) including SSA. Therefore, a sustainable food supply needs to be achieved through sustainable agriculture using irrigation systems and agricultural technologies to ensure food and nutrition security, especially under climate change.

Sudan is empowered with large cultivable land areas located between the White Nile and the Blue Nile, and between the Blue Nile and the Atbara River, beside the wide cultivated regions of Kordofan and Darfur. Arable land constitutes about one-third of the total area of the country, of which 21% is cultivated, with fluctuating productivity, but the output remains far below potential performance (Mahgoub, 2014). More than 40% of the country's area consists of forests and pastures (Mahgoub, 2014). Commercial production and small-crop farming are practiced for local consumption and export. Two types of crop cultivation systems are used in Sudan: irrigated and rainfed cultivation. The irrigated system is located along the River Nile and its tributaries, including the area between Blue and White Niles and Atbara River. The farmers in these areas use different irrigation forms, including floods, pumps, gravities, and from seasonal streams (N. M. E. Ahmed & Elsaied, 2017). The irrigated area covers approximately 1.8 million hactares and produces cotton, wheat, beans, lentil, faba bean, and a significant portion of sorghum and groundnuts (FAO, 2015b). Although it occupies only about 10% of the total cultivated area, it contributes about 14% of the added value of agriculture and 40% of the added value of the crop sector in the country (FAO, 2015b). Nevertheless, poor infrastructure, lack of agricultural technologies, lack of financial support, change of government policies, and poor maintenance of

irrigation channels in the major schemes such as Gezira, Al Rahad and New Halfa are the major challenges that are facing the irrigation farming system in Sudan. This has forced many farmers' especially, small-scale farmers to abandon crop cultivation or reduce the areas under crop cultivation.

On the other hand, rainfed cultivation is Sudan's widely practiced by small-scale farmers in Kordofan, Darfour, Gezira, Gedaref, Kasala, Blue and White Nile states covering around 80-90% of the total cultivated area (FAO, 2015b). In Sudan, a small-scale farmer typically refers to an individual or family who operates a relatively small piece of agricultural land, often less than 5 hectares. These farmers rely primarily on manual labor and traditional farming methods. The farming system in the rainfed sector is divided into semi-mechanised and traditional rainfed agriculture. The semi-mechanised covers nearly 6.7 million ha and is found in the central clay plains of Sudan in Gedaref, Blue and White Nile states and some parts of Gezira and South Kordafan states (Mahgoub, 2014). Mechanised activities are primarily limited to operations of tillage and harvesting. In contrast, the traditional rainfed farming system covers around 10 million ha; it is by far the biggest crop cultivation sector in Sudan (FAO, 2015b; Mahgoub, 2014). This farming system includes traditional practices for land preparation and sowing with rarely use of modern inputs. It is based on the use of hand-made tools that are locally made (Marizin et al., 2017). Overall, the rainfed agricultural system produced the largest quantity of sorghum (44%), the major staple food in the country (FAO, 2015b). The main crops that are cultivated under rainfed conditions include sorghum, sesame, short-staple cotton, sunflower, millet, groundnut and cowpeas (FAO, 2015b). Although rainfed farming is the most important sector in Sudan where approximately 65-70% of the population is involved, agricultural productivity remains low and the majority of the farmers have abandon farming activities and migrated to the cities. This is due to

fact that rainfed sector receive little attention from the local and national governments as well as the ongoing confilicts in Kordofan and Darfour has largely affect rainfed agricultural in Sudan. Furthermore, lack of technologies, inputs, access to credit, soil degradation and infertility and pests and disease remain the major challenges that are affecting rainfed cultivation in Sudan. This, coupled with climate variability and change, causes variations in the amount of rainfall leading to a huge impact on crop production.

2.3 Empirical review on climate variability and change

Climate variability and change pose a great challenge to natural resources in arid and semi-arid conditions (Singh & Chudasama, 2021). Variability in precipitation and increased frequency of droughts can lead to crop failures, affecting food security and the livelihoods of communities dependent on rainfed agriculture. Climate variability refers to a year-to-year fluctuation or the variations in the mean state of climate statistics on all spatial and temporal scales over a given period of time (e.g. a month, season or year). Variability may result from natural internal processes within the climate system or variations in human activities as external forces (Thornton *et al.*, 2014). While climate change refers to any change in climate over a long period of time (at least 30 years) or for an extended period, typically decades or longer, over a given region of the world, whether due to natural variability or human activities (Thornton et al., 2014). According to IPCC report, land is already under growing human pressure and climate change is adding to these pressures, affecting food and nutrition security, especially in developing countries (IPCC, 2014). Hence it is necessary to develop appropriate adaptation and mitigation measures to cope with the negative impact of climate change. Concurrently, keeping global warming to well below 2°C can be achieved only by reducing greenhouse gas emissions from all sectors, including land and food (IPCC, 2014). Therefore, climate change is considered to be one of the most significant difficulties

confronting mankind in the 21st century, and its impacts pose a tremendous and real threat to all people around the world.

The Intergovernmental Panel on Climate Change (IPCC) states that the average annual temperature will increase by 2–6°C in Africa under high emission scenarios (RCP8.5) by the end of the 21st century (IPCC, 2014). This increment is expected to be higher within the arid regions of the African continent compared to the rate of global average temperature increment (Almazroui *et al.*, 2020; Déqué *et al.*, 2017). The effect of rising temperature combined with fluctuating rainfall and scarcity of water for irrigation is argued to have a pronounced negative impact on crop yield, particularly in semi-arid and arid regions in Africa (Kangalawe & Lyimo, 2013; Raza *et al.*, 2019). Consequently, this threatens the food and nutrition security situation in agriculture-based economies, which is the main source of livelihood for the resource-poor communities, especially in sub-Saharan Africa, where millions of people depend on agriculture for their livelihood support. Africa's development and climate are closely linked: if climate change is unaddressed, it will endanger Africa's hard-won development achievements and its ambitions for poverty reduction and further growth (The World Bank Group, 2016).

Sudan is one of the most vulnerable countries to climate variability and change. In the last decades, extreme climatic events, such as floods, droughts, and heat and cold waves, have been reported in Sudan (Elagib & Elhag, 2011). A study by Elagib and Mansell (2000) showed an increase in annual temperature in Sudan between 1941 and 1996 by 0.076–0.20°C per decade. Another study by Elagib and Elhag (2011) reported that the amount of rainfall significantly decreased between 1975 and 2008, with frequent drought occurrences ranging from 44.1% to 70.6%. Future projections also showed an increase in temperature between 2.5–3.0°C by 2065, with an increasing rainfall trend (Chen *et al.*, 2013). Although these are clear indicators of climate change in Sudan, the

impact of climate variability and change on agricultural production and other sectors has been poorly documented. For instance, the seasonal temperature and annual rainfall trends in main rainfed agricultural areas like Gedaref have not been evaluated. Assessing and understanding the climate trend of a specific area is useful in determining the consequences of the social and economic impacts of climate change. Further, evaluating climate trends is essential for future agricultural production planning.

2.4 Impact of climate trends on crop yields

Climate variability and change severely threaten agriculture and food security (Rezaei et al., 2015). From the perspective of climate change predictions, both natural systems and humans are at risk. However, developing countries are more at risk because they depend primarily on natural resources for their livelihoods and have lower adaptive capacity (Mumtaz et al., 2019). Agriculture is critical to most sub-Saharan African economies, which provide about 70% - 80% of employment and contribute an average of 30% and 40% to GDP and exports, respectively (Calzadilla et al., 2013). However, the uneven distribution of rainfall patterns, reported high temperatures, and extreme weather events negatively impact agricultural production (Ahmed *et al.*, 2018). This is because an estimated 97% of existing arable lands in sub-Saharan African are under rainfed agriculture (You et al., 2011). In general, several studies have reported that crop yield under rainfed agriculture is considerably influenced by climate variables, with rainfall being associated with a higher impact on the level of yield. Specifically, Sawa and Adebayo (2018) argued that both frequencies of occurrence and length of dry spells are the main variables influencing crop yield under rainfed production systems. Furthermore, lack of sound adaptation measures coupled with the vulnerability of the rainfed agricultural sector in Sub-Saharan Africa has major effects on crop production (Mertz, Halsnæs, et al., 2009). Poor access to services, limited knowledge, and small

farm sizes have limited agricultural development in Sub-Saharan Africa. In this regard, urgent climate adaptation and mitigation measures are needed to minimise the risk and negative impact of climate change in Africa to achieve food and nutrition security.

In Sudan, changes in rainfall patterns and rising temperatures affect agricultural production of both rainfed and irrigated sectors, consequently impacting livelihoods and food security (Nelson et al., 2010). For example, following the severe drought in Sudan in the year 1984, crop production, particularly in the main crop production areas such as the Gedaref rainfed region, substantially decreased; consequently, the country was exposed to a severe famine (Eltohami, 2016). As a result of such a famine, it was reported that about 55 thousand people died, while the survivors were recorded to have suffered socio-economic loss, especially in communities that rely on crop farming and agro-ecosystem services (Elhag & Zhang, 2018). It has been documented that the optimal temperature for the production of most crops in Sudan, such as sorghum and millet, ranges between 26°C and 32°C (Mahgoub, 2014). However, in some parts of the country, temperature exceeds 47°C, causing stress and heat-related diseases to crops. This indicates that an increase in temperature trend not only affects the plant physiology but also increases crop pest and disease incidence. Furthermore, the rainfall fluctuation from the North to the South, besides its concentration into a short growing season, has increased the vulnerability of rainfed farming systems (Mahgoub, 2014). According to the world resources report 2011, the humid agroclimatic zones in Sudan will shift to the South, making northern areas increasingly unsuitable for farming (Lim et al., 2011). For example, millet production in Kordofan region is predicted to decline between 15% and 62%, sorghum between 29% and 71%, and gum Arabic between 25% and 30% under future climatic conditions between 2030 to 2060 (Lim et al., 2011). Though these are significant findings, how climate variability and change affect crop yield in the Gedaref state and Sudan at large has not been understood. It is speculated that the fluctuation in the timing and duration of the rainy season could have an important influence on crop yield (Murenzi, 2019). However, there is a lack of science-led evidence and assessment of climate effects on crop yield in the Gedaref region. This is due to the fact that determining the relationship between climatic factors and crop yield is rather difficult. Separating the effect of overlapping factors such as climate, agricultural inputs, soil fertility, technologies and management practices on crop yields remains the major challenge in assessing the impact of climate change on crop yield. Nevertheless, elucidating the relationships between climate variables such as temperature and rainfall patterns provides fundamental information that helps in agricultural planning and designing appropriate adaptation measures.

2.5 Farmers' perception of climate change and determinants of adaptation measures

Perception is defined as a process by which information is received and transformed to create psychological awareness (Qiong, 2017). People's perception of climate change varies from one place to another depending on their cultural and socio-economic differences that expose people to various values, interests, and attitudes (Ayal & Filho, 2017; Wolf *et al.*, 2013). Therefore, farmers' perceptions of climate change and its impacts are influenced by socio-economic and psychological differences that limit their response to climate change. For farmers to choose adaptation measures, they must first perceive that the climate is changing and acknowledge that it poses a challenge to their farming activities and consequently their livelihoods (Bryan *et al.*, 2013). Recently, studies on farmers' perceptions of climate change have elicited significant research interest. Literature has shown that there are different variables like age, farm size, and annual income influence farmers' perceptions of climate change. In one such study, Maddison (2007) states that farmers' perception of climate change depends on their farming experience and accessibility of extension guidance

related to climate change. In line with this finding, Foguesatto and Machado (2021) concluded that the help provided by extension offers can be like a proxy for different kinds of information. Therefore, getting extension services has significant positive impacts on the farmers' perception of climate change. In Ethiopia, the study by Tesfahunegn et al. (2016) revealed that some biophysical and institutional factors positively influence farmers' perceptions regarding climate change. Similarly, in Chile, Roco et al. (2015) found that age, income, and weather information from mass media and the internet influenced farmers' perceptions.

In Sudan, a thorough literature search on farmers' perception of climate variability and change yielded only one related study in Bahar Alarab locality of East Darfur state (Younis *et al.*, 2022). The study revealed that 69% of the respondents perceived climate change indicated by increased temperature, while 49.4% indicated a decrease in the amount of rainfall (Younis *et al.*, 2022). This clearly shows the huge gap in climate research in Sudan, particularly climate change perception. The ongoing conflicts, coupled with fragile governments and poor institutional capacity, have contributed to the lack of climate research in Sudan. Therefore, it is necessary to evaluate how farmers perceive climate change and its impact on their farming activities in order to enhance their adaptive capacity. However, perception is not enough condition for adaptation since farmers who have recognised the change in climate might not adapt or the type of their adaptation response may differ as a result of a complex interaction among environmental, social, economic, and institutional factors (Maharjan *et al.*, 2011; Mertz, Mbow, *et al.*, 2009). Thus, it is crucial to evaluate the adaption measures that farmers use in their localities to cope with climate change.

Adaptation to climate change includes all adjustments in behaviour or economic structure that reduce society's vulnerability to climate change impacts (Gyampoh *et al.*, 2009). According to
Shongwe and Manyatsi (2014), adaptation measures include changing crop varieties, shifting crop planting dates, selecting different cropping technologies, and changing irrigation systems. Some studies reported that soil conservation practices, the use of improved crop varieties, agroforestry, and adjusting planting dates are the most important adaptation measures by small-holder farmers in East Africa (Asrat & Simane, 2018; Bryan et al., 2013; Deressa et al., 2008). Although Elramlawi et al. (2020) stated that soil and water management are important adaption options for sorghum production in Gedaref state in Sudan, other adaptation measures that farmers use to cope the negative impact of climate change are still unknown. Mohmmed et al. (2018) reported that farming communities in Alfushqa and Alfaw regions in Gedaref state have a low adaptive capacity to climate change, especially for drought vulnerability. These regions are characterized by low crop diversity, productivity, and agricultural insurance (Mohmmed et al., 2018). Shifting in the onset of the rainy season coupled with a short duration of rain could be the critical factors for the farmers to adapt to climate change. However, the adaptation measures that farmers use could vary from one location to another depending on the socio-economic characteristics of the farmers. Indeed, understanding local perceptions and adaptive behavior provide relevant information for policies that help to address the challenge of sustainable agricultural development in the face of climate change (Simane et al., 2016).

2.6 Land use and land cover change

Land use and land cover (LULC) patterns on the Earth reflect the relationship between human activities and the natural environment (Alonso-Pérez *et al.*, 2003). Competitive land use, together with human population growth, causes land scarcity, turning wildlands into farmland and other uses (Kanianska, 2016). Land use and land cover change is one of the main drivers of climate

change globally and is of major concern due to its effects on different economic sectors (Wondie *et al.*, 2011). These changes occur temporally over different times, such as months or years and spatially, such as the size of the area and the land-use intensity (Houghton, 1994). In light of this, long-term changes probably are the most significant factors for global environmental change and are useful in evaluating natural resource sustainability (Lambin & Ehrlich, 1997; Tian *et al.*, 2019) . Hence, understanding LULC dynamics provide a vital factor for developing strategies for monitoring, evaluating, and conserving natural resources that are required for sustainable development (Lambin & Meyfroidt, 2011; Twisa & Buchroithner, 2019).

Historically, LULC changes are linked to variation in the biophysical environment, while recent changes are primarily associated with anthropogenic factors (Verburg *et al.*, 2004). Consequently, an area's climate pattern plays a major role in controlling human land use and land cover (Matlhodi *et al.*, 2019). Globally, human activities have been considered as a driver of environmental changes with an unprecedented rate, magnitude, and spatial size (Seto *et al.*, 2011). These changes are related to rapid population growth, economic development, and technological progress, while human land use links to different cultivation forms, grazing, protected land, timber extraction, and settlement (Matlhodi *et al.*, 2019). In developing countries, understanding land-use change dynamics is critical for sustainable land resource management, especially in sub-Saharan Africa, where most people rely on natural resources for their livelihoods.

In Africa, LULC has been highly affected by severe and recurrent droughts, anthropic/human activities, and armed conflicts, among others (Barnieh *et al.*, 2020; Mbaabu *et al.*, 2019). Particularly, the sub-Saharan Africa region is projected to be highly susceptible to the effects of LULC changes, where many parts of the region experienced a diverse pattern of LULC dynamics, with significant transformations of forest and grassland into cropland (Näschen *et al.*, 2019;

Petersen *et al.*, 2021). In addition, high levels of poverty, harvesting of fuelwood, charcoal production, agricultural expansion, settlements, unfavourable climatic events and land degradation in various agro-ecological zones are considered to be the major contributors to LULC changes in sub-Saharan Africa (Kamwi *et al.*, 2015; Mekuyie *et al.*, 2018; Munthali *et al.*, 2019). Therefore, more research on the location, extent, magnitude and rate of LULC dynamics is still needed in sub-Saharan Africa, where the population is growing rapidly, coexisting with soil infertility and overuse of nature-based resources such as forests and water.

2.7 Land use and land cover mapping

Detection of LULC changes using satellite remote sensing is one of the approaches that has been intensively used to assess and understand various land-use dynamic forces at various spatiotemporal scales (Dubovik *et al.*, 2021). Despite the recent advancement in remote sensing and geospatial tools, there is an inconsistency and a lack of standards in LULC mapping and detection products, particularly at global and regional scales (Chang *et al.*, 2018). At a local scale, several studies have successfully assessed the dynamics of LULC and their drivers (i.e., Mertz et al. 2009b; Rahman et al. 2017; Bonilla-Moheno and Aide 2020). Most of these studies have utilized commercial geospatial analytical tools like ArcGIS, Environment for Visualising Images (ENVI) and Earth Resources Data Analysis System (ERDAS), among others, for mapping and detecting LULC changes. However, in resource-limited countries, such analytical tools might not be feasibly used. Also, these tools require computers with high-performance capability to analyze 'big' satellite data. That also comes with time and cost implications in many developing countries. Moreover, most commercial geospatial tools do not allow automation of satellite data acquisition, processing and analysis. To overcome these challenges, cloud-based remote sensing and geospatial analytical tools like Google Earth Engine (GEE) have recently been introduced as freely available platforms for providing terabytes of images and advanced machine learning and artificial intelligence analytical tools (e.g., random forest (RF)) (Floreano & de Moraes, 2021). This could allow the development of relatively accurate semi- or fully automated LULC change detection approaches.

Furthermore, accurate LULC change layers could be efficiently used to predict future LULC patterns, which are also helpful for forecasting the vulnerability of ecosystems to, for instance, climate change. This requires a different set of tools that use artificial intelligence to simulate and mimic such future LULC dynamics. One of these tools is the cellular automata (CA) model, which has a high potential to effectively perform nonlinear spatially complex LULC change processes (Qiang & Lam, 2015). Cellular automata is a valuable approach for understanding LULC dynamics and their integral systems, especially when combined with other machine learning techniques, such as artificial neural networks (ANN) (Abbas *et al.*, 2021; Basse *et al.*, 2014). The CA-ANN is an artificial intelligence algorithm commonly used for simulating LULC change patterns and works on what-if scenarios (Abbas *et al.*, 2021; Baig *et al.*, 2022). In spite of the complexity of LULC set up in any ecosystem, the CA-ANN model provides comparatively accurate future predictions that could deliver to stakeholders and policymakers future LULC outlooks for informed planning (Buğday & Erkan Buğday, 2019; Qiang & Lam, 2015; Saputra & Lee, 2019).

2.8 Land use and land cover change in Sudan

In Sudan, the natural resources are continuously diminishing, where forests and natural woodlands are lost as agricultural land expands (Arfat, 2010). Also, about 80% of the energy consumed in Sudan is produced from biomass, i.e., fuelwood, charcoal, and crop residues (Hassan *et al.*, 2009). These activities have primarily reformed LULC structure, especially in the main agricultural areas.

Many studies attempted to map LULC changes in different regions of Sudan (i.e., Mohamed 2006; Zakaria 2010; Dafalla et al. 2014; Mohmmed et al. 2018). However, these studies have mainly utilized the maximum likelihood (ML) classifier with Landsat and Advanced Spaceborne Thermal Emission and Reflection Radiometer (ASTER) datasets to characterize LULC dynamics in rainfed agricultural areas. Subsequently, LULC changes in these areas were estimated using different periods. For instance, in the Northern Kordofan region, LULC change was assessed between the periods 1973 – 2001 (Dafalla *et al.*, 2014; Mohamed, 2006) and 1972 – 2007 (Zakaria, 2010). In West Kordofan, LULC changes were mapped for 2000 – 2005 (Mohmmed *et al.*, 2018). Negative vegetation cover changes were reported in the two regions, mainly due to desertification and socio-economic effects.

Gedaref state has been exposed to large-scale land degradation indicated by reduced vegetation coverage, and loss of soil fertility, among others (Biro *et al.*, 2013; E. K. Glover & Elsiddig, 2012). This is basically due to unsuccessful land-use policies and practices used, such as sorghum monocropping system and inappropriate methods of soil preparation and conservation (Biro *et al.*, 2013; Glover, 2017; Glover & Elsiddig, 2012). Additionally, the expansion of rainfed mechanized agricultural schemes in Gedaref has played a significant role in LULC changes, which resulted in land degradation, environmental deterioration, and a decline in agricultural productivity (Sulieman and Elagib, 2012). As a consequence, livelihood in this region has been highly affected. For instance, many pastoralists have lost their livestock or are forced to abandon livestock-rearing activities due to the loss of a considerable proportion of the traditional grazing lands (Sulieman and Elagib, 2012). Therefore, mapping LULC changes in Gedaref state could enable the quantification of trends in agriculture, grassland, forest cover, and freshwater resources. This can help manage agro-natural systems and improve land use policies (Midekisa *et al.*, 2017). Although many studies have been conducted in Gedaref state on LULC dynamics, these studies were done in specific areas and did not cover the whole Gedaref state, which is the country's food basket. For example, a study by Sulieman (2010) detected a large expansion of mechanised agriculture in Southern Gedaref using Landsat satellite imagery between 1972 and 2003. Issa (2018) reported a decrease in grassland and frost areas by 80% and 2.9%, respectively, in Qala El-Nahal locality in Gedaref state between 1972 and 2018. Similarly, Adam (2019) reported a decline in forest area in El-Rawashda locality in Eastern Gedaref between 1988 and 2018. The studies found drastic changes in natural vegetation, mainly due to the areal extent of mechanized rainfed farming in Gedaref state. Despite the relatively high LULC classification accuracy obtained in the previously-mentioned studies using maximum likelihood classifier, the transferability of such a parametric mapping approach to other points in space and time could be hindered by overfitting due to limited training dataset; studies have yet to predict the future LULC changes in high productive rainfed agricultural areas in Sudan like Gedaref. Moreover, no study has utilized a machine learning algorithm to classify LULC in rainfed agricultural areas in Sudan. In this regards, it is crucial to evaluate LULC change for the whole Gedaref state using such tools to assess LULC trends and their intensities to determine the underlining dynamics that caused the change in landscape structure. A survey conducted by Adam (2019) showed that the local communities identified the collection of firewood, charcoal and building materials as the main drivers that have negatively affected El-Rawashda forest in Gedaref state. However, other factors such as rapid population growth, agricultural expansion and climate change should be considered in such studies for a better understanding LULC drivers. Furthermore, future prediction of LULC change based on natural and anthropogenic drivers could help in developing land use polices and a better resource management in Gedaref state. Therefore, LULC in Gedaref state and their underlining drivers need to be evaluated using a holistic approach including satellite imagery, ground truth data and local land users' perception.

2.9 Summary

The literature review in this chapter has demonstrated the challenges of rainfed farming system in Gedardef state, Sudan. This includes climate variability and change, change in LULC structure that affect natural resources. Although these are important topics that are directly linked to food and nutrition security, they have received little priority in research, especially climate change studies and its impact on crop yield. Few studies have assessed climatic trends (temperature and rainfall) in Sudan (Elagib & Elhag, 2011; Elagib & Mansell, 2000; Sulieman & Elagib, 2012). However, the impact of climate change on crop yield is still has not been quantitatively measured, especially in the main rainfed areas of Sudan. Despite the fact that the traditional rainfed farming is the biggest crop cultivation sector in Sudan (FAO, 2015a; Mahgoub, 2014), it has been neglected by researchers compared to the mechanized sector. Traditional rainfed farming is occupied mainly by small-holder farmers, who are the most vulnerable group to the impact of climate change. Hence, it is important to assess their perceptions of climate change and the adaptation measures they use to minimize the negative impact on their farming activities. Assessing the local perceptions and adaptive behavior of small-holder farmers is essential in providing relevant information for developing policies to address the challenge of sustainable agriculture, food security and livelihood under climate change. On the other hand, LULC change is one of the main drivers of climate change and is of major concern due to its effects on different economic sectors (Wondie et al., 2011). Many studies have assessed LULC changes in Gedaref state in specific areas focusing on the mechanized agricultural expansion (Sulieman, 2010), change in vegetation cover (Issa, 2018; Sulieman, 2008; Yagoub et al., 2017), and the impact of LULC changes on pastoral

communities (H. M. Sulieman & Elagib, 2012). However, less attention has been given to the underlining dynamics of LULC changes, factors that drive these changes and future prediction of LULC structure. In other words, studies of LULC change should involve a holistic approach, including satellite imagery, ground truth data and local land users' perception. Understating LULC change and the approximate and underlining factors that drive these changes could help develop land use policies and better resource management in Gedaref state. To this end, it was critical to conduct this study to investigate the impacts of climate trends and variability and land use/land cover changes on crop production of small-holder farmers in Gedaref State, Sudan. The findings would provide a clear picture of the impact of climate change on crop yield, historical and future LULC change and the approximate and underlining drivers that caused LULC dynamics in Gedaref state.

CHAPTER THREE

STUDY AREA AND METHODS

3.1 Introduction

This chapter provides a comprehensive description of the study site that includes the location of the study area, biophysical and socio-economic settings, and the study's conceptual framework. It also presents a detailed overarching methodological description for the study objectives including methods of data collection, data management, analytical approaches, algorithms and tools used for analysis.

3.2 Location

Gedaref is one of the states in the Eastern part of Sudan; it is located between $33 - 37^{\circ}$ E Longitudes and $12 - 16^{\circ}$ N Latitudes, with an area of about 78.228 km² and its average altitude is 600 meters above sea level (Figure 3.1). It is bordered on the South by Blue Nile state, in the North by Kassala and Khartoum states, in the West by Gezira and shared a border with Ethiopia from the East. Gedaref state is the main rainfed agricultural area in Sudan and is largely recognized as the land of sorghum (primary staple food) and sesame in Sudan, where about one-third of sorghum and sesame produced are cultivated in this region. Other crops such as cotton, groundnut, millet, wheat, and sunflower are also cultivated. Gedaref state was selected as a study area because it is Sudan's hub of rainfed crop production. Also, the state has a long history (about 77 years) of a wellestablished rainfed mechanized farming system where secondary data on crop yield and climate variables can be readily obtained.



Figure 3.1: Location of Gedaref State in Sudan (Source: Author)

3.2.1 Population in Gedaref state

The population in Gedaref state was estimated to be about 2,208,385 in 2018 (Idreas, 2015). Due to the introduction of semi-mechanized rainfed farming in 1944, Gedaref state has become an important economic market, attracting more people in the area. In early 1940's Gedaref area was estimated to have a population of less than 20,000 people, while in 1968 the population of the state increased to 483,032 people. Generally, in this state, the population grows by a rate of 4.7 % annually, which is even higher than the national growth rate of 2.2 % (Idreas, 2015). In addition, the migration of people from other states in Sudan as well as refugees from the neighboring countries due to political conflicts has increased the population in Gedaref state.

3.3 Biophysical setting in Gedaref state

3.3.1 Climate

Gadaref state is characterized by semi-arid climatic conditions. The rainy season starts from June and ends in October, with a peak in August and September (Figure 3.2), while the dry season extends from November to May (Elagib *et al.*, 2019). The mean annual rainfall ranges between 200-800 mm (Idreas, 2015). However, the amount of rainfall increases from the North toward the South parts of the state with an average annual rainfall amount of 175, 570, and 650 mm in North, Center, and South Gedaref, respectively. The average annual temperature in Gedaref area is 30°C, with the hottest season occurring between the months of April and May (Yagoub *et al.*, 2017). Rainfall is considered as a very important climatic variable in Gedaref state because the majority of the population rely on rainfall to grow food and cash crops and support livestock production. The growing season in the state extends from June to October, which corresponds to the rainy season in Sudan (Figure 3.2). The planting dates range between from June to mid of July while the harvesting time range from October to November, depending on the type of the crop.



Figure 3.2: Boxplot showing the monthly rainfall in Gedaref state, Sudan between 1984 2018. The red arrows show the main crop cultivation period. Median rainfall values are shown by the bold lines within the boxes and the circles show outliers.

Source: Author

3.3.2 Soil types in Gedaref state

The area is dominated by Vertisols soil, which is characterized by dark and heavy clay. Its clay content about 60%, which it tends to increase from South to East, coinciding with the pattern of rainfall increment. The fertile clay soils coupled with and vast suitable land for crop production led to the development of the well-established and highly profitable rainfed grain cultivation system in the state (Sulieman & Elagib, 2012).

3.3.3 Agriculture as a livelihood source in Gedaref state

Agriculture is the major economic activity in Gedaref State. It provides employment and supports livelihood for around 80% of the people in the state (Mahgoub, 2014). Agricultural systems are generally rainfed, which are divided into semi-mechanized and traditional farming systems. The semi-mechanized farming systems depend mainly on machines for land preparation and sowing

and it was introduced in Gedaref state in 1944. It is practiced by large-scale farmers with farm sizes larger than 420 ha, while traditional farming dominated by small-scale farmers with an average farm size of approximately 2 ha (Mohmmed *et al.*, 2018). The principal crops cultivated in Gedaref region are sesame, sorghum, cotton, groundnut, millet, and sunflower. Sorghum and millet are grown for food consumption with sorghum being the stable food in the entire country. Sesame and sunflower are grown mainly for oil production as well as export, while cotton farming is primarily grown as an industrial and cash crop.

3.3.4 Land use/ land cover (LULC) in Gedaref state

The land use in Gedaref state is mainly dominated by agriculture, followed by livestock keeping in villages in traditional seasonal transhumance patterns. Also, as a recent feature, the large-scale mechanized merchant-farmers keep cattle as the main livestock investment enterprises. Some traditional forms of economic activity in the state are trading forest products like gum tapping and charcoal burning. Therefore, people obtain their income from a set of three different major systems of land use in Gedaref state; farming, grazing, and forestry (H. M. Sulieman & Elagib, 2012). While the main land cover features in Gaderef area according to the Food and Agricultural Organization (FAO) of the United Nations are trees, shrubs, herbaceous, cropland, bare rocks and soil, water bodies, and urban areas.

3.4 Conceptual framework for the study

The conceptual framework (Figure 3.3) developed for his study shows how climate variability and change, LULC changes, adaptation measures interact with each other and how they affect small-holder rainfed farming in Gedaref State. Therefore, this interaction can impact the livelihood and

food security of small-scale farmers in the enter State as rainfed farming being the main source of income and food in the region.



Figure 3.3: Conceptual framework of the study

Source: Author

3.5 Methods

3.5.1 The study design

This study used a transdisciplinarity (TD) approach involving collection of both qualitative and quantitative datasets. This includes impact of climate change, LULC change, crop production and yield and the perceptions of the farmers of the drivers of such changes in the study area. The TD approach is holistic and robust method in solving research problems as it allows an in-depth analysis of the problem under study. This is done through collaboration between different groups of experts and the societies that are affected by problem to find sustainable solutions to such complex challenges like climate and LULC changes (Klenk & Meehan, 2015). The present study involved both primary and secondary datasets. The primary data were obtained through 400 questionnaires for the household survey, 16 focus group discussions (FGDs), and key informant interviews. These data were supplemented with secondary datasets, which consisted of climatic data (temperature and rainfall), crop production and yield for five major crops cultivated in Gedaref state (sorghum, sesame, sunflower, millet and cotton) and remotely sensed Landsat images to address the objectives of this study.

3.5.2 Determination of the relationship between climate trends and the level of crop yields

3.5.2.1 Desktop studies and secondary data

Daily minimum (Tmin) and maximum (Tmax) temperature, which refer to the lowest and highest values of the daily temperature records, respectively, as well as daily rainfall data from 1984 to 2018 were obtained from Gedaref meteorological station (latitude 14.03° N; longitude 35.40° E;

altitude 600 m). The Tmin and Tmax were estimated every three hours daily using an alcohol thermometer. On the other hand, the daily rainfall was measured and recorded every six hours daily using rain gauges where a rainy day starts from 8:01 am and ends on the following day at 8:00 am local time (Greenwich Mean Time: GMT + 2). In addition, annual rainfall data for the same period (i.e., 1980 to 2018) were obtained from different locations in Gedaref state, namely Elghadambliya (latitude 14.023° N; longitude 35.012° E; altitude; 497 m), Um Seinat (latitude 12.845° N; longitude 35.864° E; altitude; 572 m), Samsam (latitude 12.838° N; longitude 35.733° E; altitude; 527 m), and Elhawata (latitude 13.431° N; 34.629° E; altitude; 444 m), where rainfall data were also measured using rain gauges. Data on yield for five major crops, viz., sorghum, sesame, cotton, sunflower, and millet at the Gedaref state scale were obtained from the Ministry of Agriculture, Gedaref. Basically, the crop yield data were collected from some representative farmers' fields. Crop production (ton) data and harvested area (ha) were obtained for the period 1970–2018, except for millet and sunflower, which were only available between 1982 and 2018, and 1987 and 2018, respectively. Annual yield (kg ha^{-1}) for each crop was calculated by dividing the total crop production by the harvested area.

3.5.2.2 Data analysis

A) Data quality assessment

Complete records were obtained for climate and crop yields, and these datasets were measured during the above-mentioned periods with no missing values. Since these are secondary datasets, we subjected them to a descriptive statistical analysis (skewness coefficient, mean, range, and confidence interval) and Shapiro–Wilk test (*p*-value and *W* value) of inferential statistics prior to trend and regression analyses to evaluate their quality. A summary of descriptive statistics provides

useful information about the data quality. For instance, a mean value that describes the center of a dataset distribution can be an indicator of data quality since there is prior knowledge (e.g., expert knowledge) about the dataset itself. Additionally, a close to zero skewness value indicates a moderately skewed dataset that can fit a normal distribution, which is a good indicator of data quality (Klayman, 2022). Normally distributed data imply that the observations were collected with less bias and errors that hinder the quality of the data (Klayman, 2022). Among the climatic datasets, rainfall was normally distributed (skewness = 0.33, W = 0.97 and p = 0.629), while Tmax (skewness = 0.199, W = 0.98 and p = 0.001) and Tmin (skewness = -0.34, W = 0.98 and p = 0.001) slightly deviated from a normal distribution (Figure 3.4). On the other hand, all crop yield datasets were normally distributed (sorghum: skewness = 0.31, W = 0.97 and p = 0.242, cotton: skewness = 0.46, W = 0.97 and p = 0.242, millet: skewness = 0.59, W = 0.95 and p = 0.227, and sunflower: skewness = 0.30, W = 0.95 and p = 0.126), except sesame: skewness = 1.97, W = 0.85, and p = 0.1260.000 (Figure 3.5). Furthermore, the climate and crop yield datasets had a few outliers in the Tmin, rainfall, and sesame, millet, and sunflower yield observations. These statistical metrics ensured the quality of the data that met the assumption for the present study.



Figure 3.4: Histogram, normal distribution, and boxplot fitted for minimum (Tmin) and maximum (Tmax) temperatures (1984–2018) and rainfall (1980–2018) data obtained from the Gedaref meteorological station, Sudan. Circles in some boxplots represent individual outlier observations.



Figure 3.5: Histogram, normal distribution, and boxplot fitted for crop yield data obtained from the Ministry of Agriculture, Gedaref, Sudan. The crop yield data were collected in 1970–2018, except for millet and sunflower, which were only available between 1982–2018 and 1987–2018, respectively. Circles in some boxplots represent individual outlier observations.

B) Trend Analysis

The average annual Tmax, Tmin, and diurnal temperature range (DTR: which refers to the range between Tmax and Tmin) were calculated from the daily temperature records of Gedaref metrological station for each year (1984–2018) and season of the year (i.e., summer: March to May, winter: February to November, and autumn: June to October). Afterward, Tmax, Tmin, and annual rainfall and temperature variables were subjected to Mann–Kendall trend analysis to assess the positive or negative trend in temperature and rainfall. A Mann–Kendall trend test is a nonparametric test (distribution-free) that is not sensitive to outliers (Kendall, 1948; Mann, 1945), and is widely used for analyzing time-series historical data to assess if there is a significant increasing or decreasing trend in variables of interest (e.g., temperature and rainfall) over time. In the present study, the Mann–Kendall trend test was computed using Equation (1):

$$\sum_{i=1}^{n-1} \sum_{j=1+i}^{n} \operatorname{sign} = (x_j - x_i)$$
(1)

where: x_j = the data value at time j; x_i = the data value at time i; n = the length of the time-series data; sign ($x_j - x_i$) = the sign function which can be defined as follows:

$$Sign(x_{j} - x_{i}) = \begin{cases} 1 \ if \ (x_{j} - x_{i}) > 0\\ 2 \ if \ (x_{j} - x_{i}) = 0\\ -1 \ if \ (x_{j} - x_{i}) < 0 \end{cases}$$
(2)

The increasing trend is explained by a very high positive value of the Mann–Kendall test, while the decreasing trend is explained by a very low negative value. To statistically quantify the significance of the trend in the temperature and rainfall data, Sen's slope estimator test was applied (Sen, 1968). In specific, the slope (Q), which is the change in the temperature and rainfall as a function of time (i.e., years), was calculated for all possible data pairs as follow (Equation 3):

$$Q = \frac{x_j - x_i}{j - i}$$
(3)

where x_i and x_i are the data value in the time series at time j and i, respectively.

Since there are a number of Qs (N), they are ranked from smallest to largest in the time-series data, then Sen's slope was estimated as the median of these values (Equation 4):

$$Q_{i} = \begin{cases} Q\left(\frac{N+1}{2}\right) & \text{if } N \text{ is odd} \\ \frac{1}{2}\left(Q \frac{N}{2} + Q \frac{N+2}{2}\right) & \text{if } N \text{ is even} \end{cases}$$
(4)

On the other hand, simple linear regression was employed to assess the trend of crop yield over time. This is because crop yield data depend on many factors rather than climatic variables; a reason that does not meet the assumption of the Mann–Kendall trend analysis test (Kendall, 1948; Mann, 1945). In addition, the yield data were normally distributed (Figure 3.5), hence linear regression was preferred over a Mann–Kendall trend analysis.

C) Temperature and rainfall variability analysis

To assess the annual temperature and rainfall variabilities in the study area over 35 years (1984–2018), the standardized anomaly index was calculated. Standardized anomaly index is a common index used to indicate temperature and rainfall fluctuations in regional climate change studies (Koudahe *et al.*, 2017). The standardized anomaly index was calculated as follows (Equation 5):

$$SAI = \frac{X - X_i}{\sigma} \tag{5}$$

where *X* is the mean temperature or rainfall of a year, X_i is the mean value over the long-term, and σ is the standard deviation value over the long-term. Years with an above long-term average were indicated as the most warming periods, while the years with values below the long-term average

were considered as cold periods. Similarly, years with above long-term average rainfall were indicated as the years with surplus rainfall, while years with below long-term average rainfall were indicated as the years with deficit rainfall.

Moreover, the coefficient of variation (*CV*) was calculated as follows (Equation 6) to assess the variability (fluctuation) in each temperature and rainfall variable and crop yield data:

$$CV = \frac{\sigma}{\mu} * 100 \tag{6}$$

where *CV* is the coefficient of variation; σ is standard deviation; μ is the mean of the time series data set of each climatic variable (i.e., Tmin, Tmax, DTR, and rainfall) and crop yield.

D) Characteristics of rainy season

The onset and cessation dates and length of the rainy season were analysed using R INSTAT software version 0.6.6 (Stern *et al.*, 2021). This was done by applying the threshold procedure to determine variations in the rainy season characteristics such as onset, cessation, and the length of the rainy season. The threshold for a rainy day was set at 0.85 mm (Ngetich *et al.*, 2014; Ntirenganya, 2018) and the mean rainfall was calculated for every 5 days (pentads) of the rainy season. This threshold value is appropriate for agricultural purposes in the tropical region because the accumulation of such an amount of rain contributes greatly to soil moisture (Ngetich *et al.*, 2014; Ntirenganya, 2018; Ojara *et al.*, 2020). Furthermore, the onset of the rainy season was defined as a "date when the amount of rainfall accumulation is 20 mm in 1 or 2 days within 3 dekads (dekad = 10 days), but not followed by more than 10 consecutive dry days in the next 3 dekads" (Sibanda *et al.*, 2020). This amount of rainfall is sufficient for germination and growth of the crop during the first month after planting. The cessation of the rainy season has several

definitions. Here, the approach proposed by (Tadross *et al.*, 2005) was adopted for this study, which defines the cessation of the rainy season as the date when the rain is less than 20 mm within 3 dekads followed by 2 dekads of dry days. This approach was widely used to determine the end of the rainy season (Moyo *et al.*, 2017; Setiawan, 2020). Rainfall is the most significant factor affecting crop growth and yield, particularly in semi-arid regions where rainfall is limited to only a few months per year (Laux *et al.*, 2009). Therefore, simple linear regression analysis was used to determine the relationship between the length of the rainy season and crop yield. The following steps were taken to determine this relationship: (1) the first difference approach was used to generate anomalies in crop yield and length of the rainy season and (2) in a second step, simple linear regression analysis was employed to test the impact of the length of the rainy season on crop yield anomalies.

E) Analysis of relationships between the climate variables and crop yield

Due to the effect of non-climatic factors such as crop management practices and new cultivars on crop yield, some statistical methods such as the first difference approach (Nicholls, 1997) and crop simulation models such as decision support system for agro-technology transfer (DSSAT) (Jones *et al.*, 2003) are used to evaluate the effect of climate change on crop yield (Ding *et al.*, 2021). However, models such as DSSAT require some settings and parameters, such as crop genetic coefficient, that might not have been estimated to simulate crop yield under Gedaref climatic conditions. Hence, in this study, the first difference approach was used to remove the effects of non-climatic factors on crop yield. The first difference approach was initially introduced by Nicholls (1997) and thereafter adopted by studies that evaluated the effect of climate change on crop yield (El-Maayar & Lange, 2013; Peltonen-Sainio *et al.*, 2010; Zhang *et al.*, 2014). The first

difference values for crop yield and climatic variables (i.e., anomalies) for the period 1984–2018 were generated as follows (Equation 7):

$$\Delta Y = Y_t - Y_{t-1}$$

$$\Delta X = X_t - X_{t-1}$$
(7)

where ΔY is the yield difference in two consecutive years; that is the yield in year t and t - 1, respectively, while ΔX is the difference in the climatic variable in two consecutive years; that is the climatic variable during crops growing season (June–October) in year t and t - 1, respectively.

To estimate quantitative relationships between climate variables and crop yield, the anomalies generated from the first difference for climate variables and crop yield were subjected to a Pearson's correlation analysis to determine the association between the crop yield and climatic variables. In addition, a multiple linear regression model was used to quantify the impact of climate change on crop yield using the anomalies of the first difference (Poudel & Shaw, 2016). The following linear equation (equation 8) was used to determine such a relationship for each crop:

$$Y = a + b_1 * Tmin + b_2 * Tmax + b_3 * DTR + b_4 * Rainfall$$
(8)

where, *Y* is the observed change in yield (kg ha⁻¹) due to climatic variables, *a* is the intercept of the regression model, and b_1 , b_2 , b_3 , and b_4 are the regression coefficients of Tmin, Tmax, DTR, and rainfall, respectively.

F) Models validation

A leave-two-out cross-validation procedure was used to validate the simple and multiple linear regression models. In specific, the data were divided into k samples (k = total number of crop yield samples) and then samples were removed two-by-two. Yield predictive models were fitted k times using all k data points, except for the removed ones, and validated using these omitted (holdout)

ones. A cross-validated R^2 between the observed and predicted yield data was then calculated to test the certainty of the models.

3.5.3 Determination of farmers' perception of climate variability and change and the choice of adaptation measures.

3.5.3.1 Desktop studies and secondary data

Temperature and rainfall data are presented in section *3.5.2.1*. For this section, the mean annual temperature and rainfall were used for further analysis on farmers' perceptions and choice of adaptation measures.

3.5.3.2 Household survey, key informant interview and focused group discussion

A) Household data collection

To assess farmers' perception of climate change, face-to-face interviews were conducted in Gedaref state, Sudan, between January and February 2021, using a semi-structured questionnaire. The questionnaire was divided into two parts; the first part of the questionnaire included open and closed-ended questions to collect in-depth data on the local communities' perception of climate change and the adaptation measures they use. The second part consisted of questions of LULC perceptions. Data collected in the first part of the questionnaire were used for this objective. The respondents were selected randomly for the interviews using random sampling techniques. A total of 400 respondents were selected for the interview using Yamane's, (1967) equation:

$$n = \frac{N}{1 + N(e)^2} \tag{9}$$

Where n = the selected sample size for the study, N = the total number of small-holder farmers in Gederaf state (total population) e = the standard error.

The questionnaire was tested with 20 respondents in Gedaref state who were not included in the study. The questionnaires' responses enabled to make the necessary adjustments before the actual interviews were conducted. Additionally, the selected respondents for the interview had the following criteria (i) small-scale farmers (ii) the respondents are at least 20 years old and above, and (iii) had resided in the Gedaref state for ≥ 10 years. Each respondent was interviewed for about 40 to 60 minutes.

B) Key informant interviews and focus group discussions

In addition, focus group discussion (FGD) and key informant interviews were also conducted to confirm the information generated from the household survey. This helped in gathering the necessary information and understanding the perception of the local community on climate variability and change, its impacts on their agricultural activities, and the adaptation measures used to adapt to these changes in Gedaref state. Two FGDs were held in each study village (n = 8) at the same time as household interviews. A total of 16 FGDs were conducted in the eight villages; all participants in the FGD were not included in the household interviews. The FGDs were facilitated in accordance with Hennink (2007), and were guided by a checklist of questions about climate variability and change, its impacts on their agricultural activities, and the adaptation measures used to adapt to these changes. Each FGD lasted between 120 and 180 minutes and involved between 10 and 15 people. Key informants were identified using a purposive sampling method based on their experience and knowledge of the study area. Interviews with the key informants included elders, researchers and officers from the ministry of agriculture,

meteorological authority, environmental and research institutions, natural resource conservation and agro-dealers at the state level.

3.5.3.3 Data management and analysis

The trend analysis for mean annual temperature and rainfall was conducted as describe above in section *3.5.2.2 (B)*. This trend of the metrological station was compared with farmers' perception to understand whether the local communities in Gedaref state perceive climate change correctly. The data generated from the questionnaire were coded, entered, and analyzed using Statistical Package for Social Sciences (SPSS) version 20. Descriptive statistics were used to describe the socio-economic characteristics, institutional characteristics, farmers' perceptions of climate variability and change, and their adaptation measures. A multinomial logistic regression model (equation 10) was used to examine the factors influencing farmers' choice of adaptation measures. The socio-economic characteristics of the small-holderfarmers were designated as independent variables in the model, while the adaptation measures were the dependent variables. The following equation was used:

$$logit(Y) = \alpha + \beta_1 X_1 + \beta_2 X_2 + \dots + \beta_n X_n$$
(10)

where Y = dependent variable, $\alpha =$ the intercept, $\beta_1 \dots \beta_n n =$ coefficients of associated independent variables, while $X_1 \dots X_n =$ independent variables.

3.5.4 Assess and quantify LULC changes and their intensities for 30 years (1988-2018) and project LULC in 2028 and 2048

3.5.4.1 Desktop studies and secondary data

A) Satellite remotely sensed data acquisition and pre-processing

Figure 3.6 shows the methodological approach used in this study. Landsat multispectral images are the most widely used for time series analysis of LULC classification due to the long historical data that are readily available (Qu *et al.*, 2021). Multi-date Landsat imagery were used for the years 1988, 1998, 2008 and 2018 acquired by Landsat 5 Thematic Mapper (TM), Landsat 7 Enhanced Thematic Mapper Plus (ETM+) and Landsat 8 Operational Land Imager (OLI) sensors from the freely available data catalog in Google Earth Engine (GEE) at a spatial resolution of 30 m in the World Geodetic System (WGS84). Standard image pre-processing, including cloud filtering, topographic, atmospheric, and geometric corrections, layer stacking and re-sizing was performed in GEE. A yearly (from 1st January to 31st December) median value was used to create a composite image for the selected years (i.e., 1988 and 1998 for Landsat 5, 2008 for Landsat 7, and 2018 for Landsat 8) (Griffiths *et al.*, 2013; Hermosilla *et al.*, 2015).



Figure 3.6: Methodological flowchart for the land use/ land cover (LULC) mapping, transition, intensity and future prediction.

B) Training and testing data

Five of the defined Intergovernmental Panel on Climate Change (IPCC) classes (Penman et al., 2003), namely cropland, forest land, grassland, water, and settlements were used for the classification experiment. Polygon-based training data for each class were obtained through onscreen digitization using historical high-resolution images on the Google Earth Pro platform. This method has been widely used and reported in the literature for LULC analysis, as it provides high and reliable classification accuracy (De Sousa et al., 2020; Hansen et al., 2008; Phan et al., 2020). A total of 1000 samples were selected for each year (1988, 1998, 2008 and 2018) to train and validate the LULC maps, 70% of the samples collected were used for training the classification model (700 polygons) and 30% were used for testing the model (300 points). The defined polygons, which were relatively small in size, each containing many comparatively homogeneous pixels of a specific LULC type, were used as a training dataset to reduce the influence of spatial autocorrelation. The polygons also capture the colour gradient of each LULC type (e.g., deep water against shallow water; high, medium, and low grassland coverage) to avoid confusion between the classes. On the other hand, point-based data were utilized to test the LULC maps accuracy. These points were randomly selected at a minimum distance of 100 m from the nearest training polygon to minimize overfitting and spatial multi-correlation (Phan et al., 2020).

3.5.4.2 Data analysis

A) Landsat image classification

There are many advanced non-parametric machine-learning classification algorithms in GEE for supervised classification, such as random forest (RF), support vector machines, and classification

and regression tree, among others (Lin et al., 2020). Also, a meta-analysis of more than 300 peerreviewed articles published in the last ten years before 2018 showed that the most used classifier for classifying satellite data in GEE is the RF (Tamiminia et al., 2020). The RF classification algorithm was employed in GEE to classify and distinguish among the LULC classes of Gedaref state in Sudan using the multi-date Landsat images. Many studies have reported that RF algorithm achieved higher classification accuracy and reliability compared to other machine learning algorithms (Pelletier et al., 2016; Tong et al., 2020; Zurqani et al., 2018). This is because RF is a user-friendly algorithm that requires settings and optimization of two parameters only. It also can handle large and noisy data as well as outliers, and reduce overfitting. The algorithm can simulate missing values through the calculation of proximity among samples (Feng et al., 2020). This algorithm is a combination of learning methods, which includes many individual decision trees (Breiman, 2001; Fonseka et al., 2019). Each single decision tree (ntree) has many splits (mtry, i.e., number of randomly selected variables) and nodes that predict the final class label based on the large number of votes from all decision trees. Considering the recommendations of other studies (Cánovas-García et al., 2017; Ghimire et al., 2012), 100 trees (ntree = 100) were used, and a default mtry value (the square root of the number of predictor variables). The strength of RF is that it can efficiently process a huge number of input variables without being affected by outliers and noise in the data, and is highly robust against overfitting (Ge et al., 2019; Lin et al., 2020; Phan et al., 2020).

B) Assessment of classification accuracy

The reliability of a thematic LULC map relies on the overall and individual accuracies of the map and the individual classes, respectively (Warrens, 2015). Commonly, several metrics like kappa coefficient, producer's accuracy (PA), user's accuracy (UA) and overall accuracy (OA) are utilized to validate the accuracy of the thematic maps. In this study, these accuracy metrics were calculated, except kappa coefficient, to assess the accuracy of the LULC classification experiment. Subsequently, a class-wise accuracy metric was developed by applying the F1-score formula (equation 11). This score combines PA and UA into a single fused accuracy measure ranging from 0 to 100% (Mudereri *et al.*, 2022), and it was calculated using the equation below.

$$(F1)_i = \frac{2 \times PA_i \times UA_i}{PA_i + UA_i} \tag{11}$$

Additionally, due to many concerns regarding the use of kappa coefficient in evaluating the reliability of thematic maps (Foody, 2020), two more suitable measures of disagreement were performed *viz.*, quantity disagreement (QD) and allocation disagreement (AD) that were proposed by (Pontius & Millones, 2011). The QD measures the difference between the observed and modelled class instances, whereas AD assesses the variance in the localities of the observed class samples.

C) Detection of Land use/ land cover (LULC) changes

LULC patterns of different time periods in the study area were assessed to detect the change in each class category. The change (%) for various LULC types in different point in time was calculated according to Anand and Oinam, (2020) as expressed in following equation:

$$\frac{c_2 - c_1}{c_1} \times 100$$
 (12)

where C_1 and C_2 are LULC class areas during the first (1988) and last of the study time period (2018), respectively.

The transition matrix produced in this analysis provided a general overview of the LULC stocks (amount and composition). Also, the transfers among LULC categories every 10 years during the study period, 1988–1998, 1998–2008 and 2008–2018 were evaluated.

D) Land use/ land cover (LULC) transitions mapping

In order to visually and quantitatively examine the nature of LULC transitions in Gedaref state and the transformation of each LULC class (Barnieh *et al.*, 2020; B. Nath *et al.*, 2018), we used the Semi-Automatic Classification plugin that embedded in QGIS software version 2.18.15. The thematic LULC maps between 1988 and 2018 were used to create LULC transitions maps and their corresponding transition matrixes, where we used the LULC maps for the years 1988, 1998, and 2008 as reference layers to detect the transitions in each class in 30, 20 and 10 years (i.e., till 2018), respectively.

E) Intensity analysis in land use/ land cover (LULC) transitions

Overall, the thematic LULC maps do not mimic the pattern and magnitude of the change that cause the landscape transformation (Asante-Yeboah *et al.*, 2022). To address this, Aldwaik and Pontius (2012) proposed the so-called 'intensity analysis', which is a qualitative approach for better understanding the magnitude of the transformation in landscape structure. LULC intensity analysis was performed using the contingency table for each period to look at the extent and intensity of change at various scales; interval, category, and transition. The analysis of the interval level computes the rate and size of change over a specific point in time. Whereas the categorical level analysis examines differences in the intensity of change across LULC classes. Lastly, the analysis of the transition level emphasises on the magnitude and direction of the change between the LULC categories in each time interval (Asante-Yeboah et al., 2022; Zhou et al., 2014).

The uniform intensity lines provided by all three levels of analysis depict a theoretical situation in which uniform transformation takes place across all LULC classes. The predicted class area from the interval level experiment defines the period that has annual fast or slow changes compared to the uniform intensity line. When the intensity of a category exceeds the uniform line, it is called an active category; when it falls lower than the uniform line, it is called a dormant category. Similarly, in the transition intensity, a targeted class is the one that its loss or gain exceeds the uniform intensity line. On the other hand, if a category does not reach the uniform intensity line, it is regarded as avoided (Asante-Yeboah et al., 2022). Initially, transition matrices of the periods 1988-1998, 1998-2008, and 2008-2018 were generated for the thematic LULC maps. Thereafter, a tool developed by Aldwaik and Pontius (2013) was used to compute the three intensity levels at different time intervals using the following equations and their description provided in Table 3.1: Firstly, the transitions at the interval level were computed (Equation 13), through dividing the magnitude of change by the length of time interval, generating percentage of spatial extent. The categorical annually gross loss intensity in a time interval was calculated by dividing the size of the category's annual gross loss by the size of the category at the beginning of each interval (Equation 14). On the other hand, the category's annually gross gain intensity in a time interval was calculated by dividing the size of the category's annually gross gain with the size of the category at the final stage of each time interval (Equation 15). The common hypothesis for each interval's category level proposes that all categories experience gross loss and gross gain with the same annually intensity. This sum is equal to the transition rate in the interval (S_t) . If $L_{ti} < S_t$, the loss of i, is paused during the interval t. In contrast, if $G_{tj} < S_t$, the gain of j is withheld during the interval t. In the case $L_{ti} > S_t$, loss of i is considered to be active during the interval t; similarly,

if $G_{tj} > S_t$, gain of *j* is considered active during that time interval. Equation (16) computes the annual transition intensity of the gain in a specific category *n* from other categories*i*, that is the amount of the annually transition to the specific category *n* from the other category divided by amount of another category at the beginning of each interval. The hypothesis at the level of transition for intervals states that particular category *n* moves to all other categories with a comparable annual intensity. This amount is calculated by dividing the size of the yearly gain of category *n* by the total quantities of sizes of all other categories at the beginning time of intervals (Equation 17). Hence, if $R_{tin} < W_{tn}$, the gain of *n* pause *i* during the interval *t*. If $R_{tin} > W_{tn}$, the gain of *n* targets *i* within interval *t*.

Table 3.1: Mathematical symbols used to calculate different intensities as illustrated in equations 3-7 as described by Aldwaik and Pontius (2012).

Symbol	Description
Т	number of time points
γ_t	year at time point t
t	index for the initial time point of an interval
	$[\gamma_t - \gamma_{t+1}]$, where t ranges from 1 to $T - 1$
J	number of categories
i	index for a category at the initial time point of an interval
j	index for a category at the latter time point of an interval
n	index of the gaining category for the selected transition
C _{tij}	size of transition from category <i>i</i> to category <i>j</i> during interval $[\gamma_t - \gamma_{t+1}]$
S _t	annual change during interval $[\gamma_t - \gamma_{t+1}]$
G_{tj}	intensity of annual gain of category j during interval $[\gamma_t - \gamma_{t+1}]$ relative to size of
	category j at time $t + 1$
L _{ti}	intensity of annual loss of category i during interval $[\gamma_t - \gamma_{t+1}]$ relative to size of
	category i at time t
R _{tin}	intensity of annual transition from category <i>i</i> to category <i>n</i> during interval $[\gamma_t - \gamma_{t+1}]$
	relative to size of category i at time t
W _{tn}	uniform intensity of annual transition from all non- n categories to category n during
	interval $[\gamma_t - \gamma_{t+1}]$ relative to size of all non- <i>n</i> categories at time <i>t</i>

$$S_{t} = \frac{Change\ during\ [\gamma_{t},\gamma_{t+1}]}{(Duration\ of\ [\gamma_{t},\gamma_{t+1}])(Extent\ Size)} 100\% = \frac{\sum_{j=1}^{J} [(\sum_{i=1}^{J} C_{ij}) - C_{ij}]}{(\gamma_{t+1},\gamma_{t})(\sum_{j=1}^{J} \sum_{j=1}^{J} C_{ij})} 100\%$$
(13)

$$L_{ti} = \frac{Annual \ loss \ o \ f \ i \ during \ [\gamma_t, \gamma_{t+1}]}{Size \ o \ f \ i \ at \ \gamma_t} 100\% = \frac{[\Sigma_{i=1}^J \ C_{tij}] - (\gamma_{t+1} - \gamma_t)}{\Sigma_{j=1}^J \ C_{tij}} 100\%$$
(14)

$$(4)G_{tj} = \frac{Annual \ gain \ of \ j \ during \ [\gamma_{t}, \gamma_{t+1}]}{Size \ of \ j \ at \ \gamma_{t+1}} 100\% = \frac{[(\sum_{i=1}^{J} c_{tij}) - c_{tij}]/(\gamma_{t+1} - \gamma_{t})}{\sum_{i=1}^{J} c_{tij}} 100\%$$
(15)

$$R_{tin} = \frac{Annual \ transition \ from \ i \ to \ n \ during \ [\gamma_t, \gamma_{t+1}]}{Size \ of \ i \ at \ \gamma_t} 100\% = \frac{C_{tin/}(\gamma_{t+1} - \gamma_t)}{\sum_{i=1}^J C_{tij}} 100\%$$
(16)

$$W_{tn} = \frac{Annual \ gain \ of \ n \ during \ [\gamma_t, \gamma_{t+1}]}{Size \ of \ non-n \ at \ \gamma_t} 100\% = \frac{[(\sum_{i=1}^{J} C_{tin}) - C_{tnn}]/(\gamma_{t+1} - \gamma_t)}{\sum_{j=1}^{J} [(\sum_{i=1}^{J} C_{tij}) - C_{tnj}]} 100\%$$
(17)

F) Future land use/ land cover (LULC) prediction and validation

After generating LULC maps from Landsat data for the period 1988 to 2018 with 10 years intervals, future simulation of LULC change was performed using Cellular Automata Artificial Neural Network (CA-ANN) algorithm in MOLUSCE plugin that embedded in QGIS software version 2.18.15. Studies have shown that the CA-ANN model is more powerful and robust in simulating future LULC as compared to other models like linear regression and Marcov (Abbas *et al.*, 2021; El-Tantawi *et al.*, 2019; Rahman *et al.*, 2017). Moreover, the MOLUSCE plugin effectively processes LULC change analyses and is suitable for evaluating spatio-temporal LULC changes and predicting future scenarios (Gismondi, 2013; Muhammad *et al.*, 2022). For future LULC predictions, the same resolution was retained (30 x 30 m) and WGS 84 coordinate system.

To simulate future LULC, it is recommended that a number of predictor variables, which play a major role in LULC change and transition should be considered (McCarthy *et al.*, 2001). Based on LULC change drivers that were reported in previous studies (Gounaridis *et al.*, 2019; Kafy *et al.*, 2021; Wu *et al.*, 2022), and the availability of such factor datasets, 8 predictor variables (Figure

3.6) were selected to describe the LULC change processes that occurred in Gadaref state between 1988 – 2018. These predictors include topographic variables such as slope, aspect and elevation; and human disturbance variables like distance from Gedaref state center, towns, highways, roads and the railway line (Figure 3.7). These variables are frequently used to predict LULC because they provide reproducible data on the natural and human disturbances in LULC processes (Muhammad *et al.*, 2022).

Prediction of future potential LULC for a prospective project can be only reliable if the simulation outcome is validated using existing datasets. Accordingly, in the first step, we predicted LULC for the year 2018 using the transition matrix generated from the thematic maps of the years 1998 and 2008 and the selected predictor variables that are presented in Figure 3.7. Thereafter, the validation process was performed using a comparative analytical procedure of the overall correctness percentage and kappa coefficient in the MOLUSCE plugin. Specifically, to validate the performance of CA-ANN model, the simulated LULC map for 2018 that was generated using CA-ANN algorithm was compared with one that generated for the same year using the multi-date Landsat images and RF classifier. After obtaining adequate validation metrics, LULC data from 2008 and 2018 maps (herein referred to as prediction data) were utilized to simulate future LULC in 2028 and 2048.


Figure 3.7: Predictor variables used for land use/ land cover (LULC) future predictions **Source:** Author

3.5.5 Assess the local farmers' perception of land use/ land cover (LULC) change trends; and determine LULC drivers in Gedaref state.

3.5.5.1 Desktop studies and secondary data

Remote sensing and geospatial data have been identified as trustworthy sources for determining and understanding LULC change drivers of homogenous and heterogeneous landscapes (Daba & You, 2022). Therefore, the LULC maps for the years 1988, 1998, 2008, and 2018 generated using the methodology of objective 3, were used to determine the land cover trend. The change in the LULC classes (area in ha) for each year was extracted from the classified maps and then compared with farmers' perceptions.

Temperature and rainfall data for the years 1988 to 2018 obtained in section *3.5.2.1* of objective 1 were used for this objective. However only annual mean temperature and rainfall were used for the analysis. The total cultivated area, production, and yield for the five main crops, namely sorghum, sesame, millet, sunflower and cotton, were acquired from the Ministry of Agriculture in Gedaref state. While population datasets for the state were acquired from the Sudan Central Bureau of Statistics. Sudan has undertaken three population censuses before and during the study period (1983, 1993 and 2008). The census population for the years 1988, 1998 and 2018 were estimated from the nearest census data of Gedaref state and annual population growth rates in Gedaref state based on the following equation adopted by Kindu et al. (2015) and Munthali et al. (2019):

$$P_2 = P_1 e^{rt} \tag{18}$$

Where P_1 = total population size of the nearest census; P_2 =the estimated total populations for the given year; e = exponential constant (2.718); t = number of years between the initial census (nearest census P_1) and estimated enumeration (P_2); and r = the rate of the annual population growth in Gedaref state (4.7%).

3.5.5.2 Household survey, key informant interview and focused group discussion.

A) Household data collection

The second part of the questionnaire that was described in section 3.5.3.2 of objective 2, was basically contained questions to assess farmers' perception of LULC change and the underlying factors that driver these dynamics in Gedaref state. This part of the questionnaire included

questions that were designed to collect in-depth data on the local land users' perception of LULC dynamics, the driving factors of these changes and crop yield in Gedaref state during the study period (1988–2018). The same 400 respondents that were selected in section *3.5.3.2* were interviewed.

B) Key informant interview and focus group discussion

A total of 16 FGDs were conducted in the eight villages using the same procedure described in section *3.5.3.2.2* of objective 2. These FGDs were done separately from the ones that were done in objective 2. The FGDs were conducted to generate the necessary information and understanding the perception of the local land users on LULC changes that occurred in Gedaref state and the underlying causes that have contributed to these changes. Discussions with the key informants including elders, researchers and officials from agricultural, environmental institutions, and natural resource conservation at the state level.

3.5.5.3 Data management and analysis

The generated data from the questionnaires on the perception of LULC were entered, coded and cleaned before the analysis. Thereafter, descriptive statistics was used to describe the socioeconomic and demographic characteristics of the respondents. In addition, the perceived LULC change drivers by the local land users were ranked based on their weighted average using the ranking index adopted by Musa et al. (2006) and Solomon et al. (2017):

Index =
$$\frac{R_n C_1 + R_{n-1} C_2 \dots + R_1 C_n}{\sum R_n C_1 + R_{n-1} C_2 \dots + R_1 C_n}$$
 (19)

Where R_n = value assigned to the lowest rank (for instance, if the lowest rank is the 5th, then R_n = 5, $R_{n-1} = 4$, $R_1 = 1$; C_1 = counts of the lowest ranked level (in the above example, the count of the 5th rank = C_5 , and the count of the 1st rank = C_1).

Multinomial logistic regression model (equation 10 in objective 2) was used to determine the key factors that drive LULC dynamics in Gedaref state at the household level. The identified perceived drivers for LULC changes were the dependent variables for the model, while socio-economic variables such as sex, sex of the household, age, education level and land tenure were the independent variables.

This study only focused on three LULC classes, namely cropland, forest, and settlement. Therefore, the cropland, forest and settlement areas estimated from the satellite images and the actual total cultivated area for the five main crops that are cultivated in Gedaref state were subjected to Mann-Kendall trend analysis to assess the increase or decrease trends in their coverage based on Sen's slope value. In addition, we employed simple linear regression to assess the trends in the production of the five crops over time. This is due to the fact that crop production is largely related to crop harvested area rather than climatic and other variables. Hence, the Mann-Kendall trend analysis is not appropriate for such data as crop production data do not meet the assumption of this analysis (Kendall, 1948; Osman et al., 2021). Moreover, the association between each of the LULC drivers (temperature, rainfall and population) and the change (area in ha) in LULC classes (cropland, forest and settlement) was determined using Pearson's correlation test. The same correlation analysis was also employed to assess the association between the actual total cultivated areas and crop yields for the five major crops during the studied period (1988-2018). This analysis was done to get insights into whether crop acreage and yield were changed simultaneously in Gedaref state. Furthermore, a multiple linear regression model was used to

determine the effect of LULC drivers (temperature, rainfall and population) on the area (in ha) of LULC classes (cropland, forest and settlement).

CHAPTER FOUR

CLIMATE VARIABILITY AND CHANGE AFFECT CROPS YIELD UNDER RAINFED CONDITIONS: A CASE STUDY IN GEDAREF STATE, SUDAN

4.1 Introduction

This chapter presents results on an assessment of the trends in minimum and maximum temperatures (Tmin and Tmax), diurnal temperature rang (DTR) and rainfall in Gedaref state, Sudan, between 1984 and 2018. It also characterizes the onset, cessation and length of the rainy season in Gedaref state for the same period. In addition, this chapter also presents an assessment of the trends in crops yield of five major cultivated crops in Gedaref state: sorghum, sesame, sunflower, millet and cotton. Furthermore, the chapter investigates the impact of rainfall and temperature-based variables (Tmin, Tmax and DTR) on the yield of the main crops in Gedaref state.

4.2. Results

4.2.1. Estimation of annual and seasonal temperature trends in Gedaref state

The Mann–Kendall and Sen's slope trend analysis showed that the annual Tmax significantly (p < 0.01) increased in Gedaref state by 0.03°C per year, between the years 1984 and 2018, with a confidence interval ranged between 0.014 and 0.043°C (Table 4.1; Figure 4.1A). Similarly, the annual Tmin significantly (p < 0.0001) increased by 0.05°C per year, with a confidence interval ranging between 0.031 and 0.061°C (Table 4.1; Figure 4.1B). Between the years 1984 and 2018,

the Tmin in Gedaref state ranged between 20.81 and 23.26°C per year, while the values of the Tmax ranged from 36.10 to 38.34°C (Table 4.1). The annual DTR had significantly (p < 0.01) decreased by 0.02°C per year, with a confidence interval between -0.234 and 0.142°C for the study period (Table 4.1; Figure 4.1C). The seasonality trend showed that the Tmax increased by 0.02 and 0.04°C per year in winter and summer, respectively, with no change reported in autumn (Table 4.1; Figure 4.1A), while the Tmin in winter, summer, and autumn has significantly (p < 0.001) increased by 0.05°C, 0.06 °C, and 0.04 °C per year, respectively (Table 4.1; Figure 4.1B). The highest decrease in DTR was reported in winter, with an estimated value of 0.03 °C per year (p < 0.001).

Table 4.1: Estimated Sen's slope values for the annual and seasonal temperature (°C) trends at Gedaref state, Sudan between 1984 and 2018.

Parameter	Range		Son's Slong	95% Confidence	Coefficient	
	Minimum	Maximum	-sen s slope	interval	(CV %)	<i>p</i> value
Annual Tmin	20.817	23.266	0.045	0.031-0.061	2.65	< 0.0001
Annual Tmax	36.100	38.343	0.030	0.014-0.043	4.33	0.001
Annual DTR	14.064	16.090	-0.023	-0.234-0.142	6.82	< 0.003
Winter Tmin	17.248	21.283	0.048	0.025-0.070	4.29	< 0.001
Winter Tmax	34.687	37.548	0.017	-0.001-0.039	1.74	0.069
Winter DTR	15.661	18.350	-0.037	-0.207-0.192	3.38	< 0.001
Summer Tmin	22.957	25.838	0.056	0.036-0.071	2.83	< 0.0001
Summer Tmax	38.809	41.771	0.039	0.014–0.63	3.19	0.003
Summer DTR	15.034	17.174	-0.009	-0.304-0.246	3.19	0.288
Autumn Tmin	21.349	23.637	0.038	0.023-0.051	2.33	< 0.0001
Autumn Tmax	33.228	36.676	0.002	-0.026-0.034	2.58	0.809
Autumn DTR	11.073	15.207	-0.026	-0.001-0.312	6.81	0.045

Tmin, Tmax and DTR are minimum temperature, maximum temperature and diurnal temperature range, respectively.



Figure 4.1: Annual and seasonal temperature trends in Gedaref state, Sudan between 1984 and 2018. (A): maximum temperature, (B): minimum temperature, and (C): diurnal temperature range.

4.2.2. Determination of annual rainfall trends in Gedaref state

Mann–Kendall and Sen's slope trends for the annual rainfall data for the five different stations are presented in Table 4.2 and Figure 4.2, respectively. Overall, there was a variation in the annual rainfall trends between 1980 and 2018, with no significant (*p* ranged between 0.131 and 0.841) increase or decrease in the amount of rainfall within this period in the five locations. Rainfall decreased in Gedaref and Samsam by about 0.3 mm per year and increased in the other three locations by about 2.5–3.1 mm per year (Table 4.2) during the study period, with high variability in El hawata (CV = 26.57%) and low variability in Samsam (CV = 20.59%). The highest decrease in rainfall was recorded at the Samsam location, fluctuating from 420 to 1023 mm (Table 4.2;

Figure 4.2D). The overall trend of the average rainfall recorded in the five locations revealed that the rainfall has increased by ≈ 1.0 mm in Gedaref state between 1980–2018, with *CV* of 14.73% (Table 4.2; Figure 4.2F).

Table 4.2: Estimated Sen's slope values for annual rainfall (mm) trends reported at different stations in Gedaref State, Sudan between 1980 and 2018.

Location	Range		Sen's	95% Confidence	Coefficient of	n-Value	
Location	Minimum	Maximum	slope	interval	%)	P^{-1} and	
Gedaref	322.000	871.000	-0.323	-3.967-3.636	21.29	0.841	
El gadabalea	285.000	755.000	2.571	1.032-5.533	21.63	0.217	
Am Senat	435.000	1070.000	3.100	-1.250-7.091	22.35	0.150	
Samsam	420.000	1023.000	-3.222	-7.757-0.769	20.59	0.131	
El hawata	222.000	809.000	2.611	-1.679-7.152	26.57	0.183	
Mean for all five locations	425.4000	753.8000	0.9627	-34.200-41.080	14.73	0.4135	



Figure 4.2: Annual rainfall trends in five locations in Gedaref state, Sudan between 1980 and 2018. (A): Gedaref, (B): El gadabalea, (C): Am Senat, (D): Samsam, (E): El hawata, and (F): Average rainfall for all five locations.

4.2.3. Assessment of annual crop yield trends in Gedaref state

The trend analysis showed a decrease in the annual yield of the five studied crops between 1970 and 2018, with the exception of sunflower and sesame, which had a yield increment (Table 4.3; Figure 4.3). However, there was a significant yield change for sorghum (p< 0.01) and sunflower (p< 0.001) only (Table 4.3; Figure 4.3). In particular, sorghum and cotton, respectively recorded the highest (0.41 kg ha⁻¹ per year) and lowest (0.02 kg ha⁻¹ per year) yield decrement, while sunflower recorded a yield increment of about 0.61 kg ha⁻¹ per year (Table 4.3; Figure 4.3).

Table 4.1: Estimated slope of linear regression for the annual yield trends for five crops (sesame, sorghum, cotton, millet, and sunflower) in Gedaref state, Sudan from 1970–2018.

Crop Vield	Range		Slope	95% Confidence	Coefficient	n Voluo
Crop Tielu	Minimum	Maximum	Slope	interval	(CV %)	<i>p</i> -value
Sorghum	185.718	1000.020	-0.409	-0.6760.141	37.85	0.003
Sesame	111.907	780.016	0.163	-0.127-0.453	40.78	0.263
Cotton	58.096	1190.500	-0.018	-0.311-0.276	55.49	0.905
Millet	216.671	642.870	-0.025	-0.368-0.318	29.29	0.883
Sunflower	238.100	833.350	0.607	0.310-0.903	30.16	0.000



Figure 4.3: Annual crop yield trends for the major crops grown in Gedaref State, Sudan from 1970–2018, (**A**): sorghum, (**B**): sesame (**C**): cotton, (**D**): millet and (**E**): sunflower yield trends.

4.2.4. Estimation of temperature and rainfall variability indices

The temperature and rainfall anomalies that occur in Gedaref state for the period 1984–2018 were described by mean annual temperature and rainfall. Standardized anomaly index for the mean

annual temperature in Gedaref state was characterized by the below long-term average between 1984 and 2000, indicating cold years (Figure 4.4A). The cold years started in 1985 and continued with only one warm year (i.e., 1990) until 2000. Afterward, there were warm years which continued until 2018, which are characterized by being above the long-term average with two breaks of cold years (Figure 4.4A). Standardized rainfall anomaly index for long-term annual rainfall of Gedaref for the period 1984–2018 was used to identify the years with rainfall deficit and the years with surplus rainfall. The coefficient of variation for annual rainfall was 23%, which indicates that there was no high variation in the amount of rainfall between the years. The year 1984 experienced the highest rainfall deficit occurrence, followed by 1990, 1991, 2013, 2011, and 1987, in decreasing order of magnitude (Figure 4.4B). In contrast, the years with the highest surplus rainfall were 1999, 2014, 2003, and 2002, as can be observed in Figure 4.4B.



Figure 4.4: Standardized anomaly index for mean temperature (**A**), and annual rainfall (**B**) in Gedaref State, Sudan from 1984–2018.

4.2.5. Characteristics of rainy seasons in Gedaref state

On average, the starting dates of the rains in Gedaref state were estimated to range between the second and third dekad of June, which corresponds to the 166–176 days of the year. The cessation dates estimated between the second and third dekad of September and the first and second dekad

of October, which falls between the 250–287 days of the year (Table 4.4). The length of the rainy season in Gedaref state ranged between 57 days (2013 season) and 117 days (2016 season) (Table 4.4). Some exceptions are noticed in the years 1991, 1998, and 2013 when onset dates occurred within the second and third dekad of July, while cessation date exceptions are 2007 and 2012 in the first dekad of September and the year of 2017 in the third dekad of August (Table 4.4). The total amount of rainfall ranged between 286 and 820 mm, with an average of 539.7 mm per season (Table 4.4).

Linear regression analysis showed that an increase in the length of the rainy season significantly increased the yield of sesame (p < 0.001 and $R^2 = 0.47$), sorghum (p < 0.001 and $R^2 = 0.43$), sunflower (p < 0.001 and $R^2 = 0.49$), and cotton (p < 0.05 and $R^2 = 0.22$) (Figure 4.5). However, the yield of millet was not significantly (p = 0.432 and $R^2 = 0.02$) affected by the length of the rainy season (Figure 4.5).

Season	Onset date	Day of the year for onset	Cessation date	Day of the year for cessation	Length of the rainy season (Day)	Total rain (mm)
1984	7-July	189	16-September	260	76	286
1985	20-June	171	12-October	285	114	669
1986	29-June	180	3-October	275	96	525
1987	19-June	170	12-October	285	115	445
1988	29-June	181	18-September	262	81	532
1989	22-June	173	19-September	262	89	682
1990	15-June	166	25-September	268	102	335
1991	13-July	129	1-October	274	80	308
1992	4-July	186	13-October	287	101	520
1993	18-June	169	28-September	271	102	693

Table 4.2: Estimated length of the rainy season (1984–2018) for Gedaref State, Sudan.

Season	Onset date	Day of the year for onset	Cessation date ¹	Day of the year for cessation	Length of the rainy season (Day)	Total rain (mm)
1994	17-June	168	24-September	267	99	579
1995	17-June	168	19-September	262	94	499
1996	22-June	174	30-September	274	100	576
1997	25-June	166	24-September	267	91	505
1998	17-July	198	10-October	283	85	552
1999	20-June	171	8-October	281	110	766
2000	24-June	176	1-October	275	99	621
2001	23-June	174	7-October	280	106	430
2002	10-July	191	21-September	264	73	629
2003	21-June	172	2-October	275	103	820
2004	20-June	172	10-October	284	112	579
2005	22-June	173	20-September	263	90	504
2006	19-June	170	28-September	271	101	626
2007	24-June	175	7-September	250	75	575
2008	23-June	175	12-September	256	81	528
2009	2-July	183	15-September	258	75	510
2010	23-June	174	14-October	287	113	544
2011	19-June	170	13-September	256	86	408
2012	23-June	175	28-August	241	66	511
2013	21-July	202	16-September	259	57	418
2014	24-June	175	4-October	277	102	787
2015	9-July	190	28-September	271	81	399

Table 4.4(cont.):

18-June

18-June

29-June

13-October

6-September

29-September



Figure 4.5: Relationships between the length of the rainy season and yield of five major crops grown in Gedaref state, Sudan. R^2 is coefficient of determination.

4.2.6. Assessment of the relationship between climatic variables and crop yield in Gedaref

The result of the relationships among crop yield and temperature variables as well as rainfall, as assessed using Pearson's correlation, are presented in Figure 4.6. There were negative relationships among the temperature-based variables (Tmin, Tmax, and DTR) and crop yield, whereas the relationships among rainfall and crop yield were positive. The associations among all studied climate variables and most of the crop yields were significantly correlated (p < 0.01), with correlation coefficient values of above 0.5, except for millet and Tmin and rainfall, cotton and all climate variables, and sesame and DTR (Figure 4.6).



Figure 4.6: Associations, as assessed by a Pearson's correlation analysis, among climatic variables and yield of five major crops grown in Gedaref State, Sudan. r is correlation coefficient.

The results of the multiple linear regression analysis estimating the changes in yield of each of the five crops as a function of temperature-based and rainfall predictor variables are presented in Table 4.5. When all the climate variables were combined in one linear model, the result showed that 50% to 70% of sorghum, millet, and sunflower yield variabilities could be explained by the studied climatic variables. This was also confirmed by the cross-validated R^2 , which ranged between 0.54 and 0.69, attesting to the certainty of the crop yield estimate models (Table 4.5). However, the regression coefficients were only significant (p < 0.05) for Tmin and DTR on sorghum yield changes. In addition, the coefficients of Tmin and Tmax were significant (p < 0.05) for the change in millet yield model, and rainfall in the sunflower yield model (Table 4.5). For sesame and cotton, only 41% and 6% of their yield variability, respectively, could be explained by the climatic predictor variables, indicating a very weak relationship between these variables and the change in yield of these two crops.

Cı	rop	Intercept	Tmin (°C)	Tmax (°C)	DTR (°C)	Rainfall (mm)	R ²	Cross- Validated R ²
Secome	Coefficient	-0.734	-23.337	-32.206	10.891	0.239	0.41	0.38
Sesame	<i>p</i> -value	0.96	0.59	0.22	0.64	0.07	0.41	0.58
Sorahum	Coefficient	33.980	-260.213	40.452	-132.345	0.110	0.70	0.60
Sorgnum	<i>p</i> -value	0.293	< 0.001	0.42	< 0.05	0.65	0.70	0.09
A (*11 -	Coefficient	-8.578	217.319	-191.497	102.271	-0.021	0.54	0.54
Millet	<i>p</i> -value	0.61	< 0.05	< 0.05	0.19	0.84	0.34	0.34
Sunflower	Coefficient	-16.746	9.8571	-16.042	-25.435	0.536	0.61	0.62
	<i>p</i> -value	0.43	0.87	0.65	0.49	< 0.01	0.01	0.62
a	Coefficient	-3.610	-93.661	-17.070	0.000	-0.277	0.00	0.00
Cotton	<i>p</i> -value	0.93	0.40	0.67	-	0.33	0.06	0.08

Table 4.3: Multiple linear regression terms and R^2 values for estimating yield, as a function of climatic variables, of five major crops grown in Gedaref state, Sudan.

4.3. Discussion

This study provides information on climate trends and their effect on the yield of five main crops grown in Gedaref state, which is the most important rainfed agricultural area in Sudan that lies in the semi-arid region. The crops were grown under rainfed conditions during the rainy season (June–October). The results showed that there is an increasing trend in annual temperature over the last 35 years. These results agree with the findings of Loh et al. (2020), who reported an increase in the mean annual temperature trend by 0.06°C per year for the period between 1985– 2015 in the Eastern part of Sudan. A similar trend was also reported for semi-arid regions of Iran where Tmin and Tmax increased over the last 50 years (Bannayan et al., 2020). However, this study showed that the DTR annual trend decreased by 0.23° C per decade, which was indicated by a narrowing range of Tmin and Tmax. The trends of Tmin for all seasons of the year (winter, summer, and autumn) increased, while the DTR trend decreased between 1984 and 2018. The results of this study are also following the findings of Elagib (2010), who reported a rise in temperature in Sudan with a warming rate of 0.424, 0.357, and 0.451°C per decade for summer, winter, and autumn seasons, respectively, between 1941 and 2005. Yet, it has been argued that an increase in mean temperature by 2.6°C should be expected in the study area in 2070 (Platts1 et al., 2015) and that might have a serious impact on crop production. The rainfall trend analysis for the five locations in Gedaref state showed variation in annual rainfall, which could be explained by the fact that rainfall varies from one location to another. Nevertheless, coefficient of variation for rainfall amongst the five locations was almost similar except for El hawata, where rainfall variation was slightly higher. The trend of the overall mean for the five locations showed an increase in rainfall trend in Gadaref by ≈1 mm for the years between 1980 and 2018. However, the projected regional climatic model showed that the annual amount of rainfall in Gadaref state might decrease by 50 mm by the end of this century (Platts1 *et al.*, 2015).

The standardized anomalies of annual mean temperature revealed that there was a cold period between 1984 and 2000 in the study area, which is indicated by anomalies below the long-term temperature average. However, after the year 2000, temperatures above the long-term average were detected, indicating a warm period. In fact, temperature had increased by 1.5°C above average in the years spanning 2009, 2010, 2015, and 2016, and this warming continued until the year 2018. The year-to-year departures were not reported for temperature anomalies, suggesting that the changes for some years were slightly around the mean (Koudahe *et al.*, 2017). Similarly, the standardized anomalies of annual rainfall indicated that the year 1984 took the first highest position of rainfall deficit occurrence, followed by the years 1990, 1991, 2013, 2011, and 1987, respectively. These anomalies in annual rainfall showed that the years with rainfall deficit corresponded to the true occurrence of droughts in Sudan. For example, the rainfall deficit in 1984 led to drought throughout the country, which resulted in famine and the death of thousands of people within the country (Hamid & Eltayeb, 2017). Overall, it is clear that the mean annual temperature between 1984 and 2018 was characterized by variability at a decadal scale, unlike the annual rainfall series, which is characterized by variability between the years. The results showed that an increase in the length of the rainy season significantly increased the yield of sesame, sorghum, sunflower, and cotton. These results agree with the findings of Murenzi (2019), who showed a positive relationship between the length of the rainy season and maize yield in Rwanda. However, the relationship between the length of the rainy season and cotton yield should be interpreted with caution as it could have been affected by the outliers when the change in the rainy season was above 36 days and cotton yield was more than 600 kg ha⁻¹ (Figure 4.5). In addition,

there was no clear relationship between the length of the rainy season and the yield of millet. This could be attributed to the fact that millet is a drought-tolerant crop with low water requirements, a characteristic that enables the crop to tolerate the terminal drought that usually occurs towards the end of the growing season during the grain-filling stage (Tadele, 2016).

The yield trend of sorghum, millet, and cotton decreased by 0.409 kg ha⁻¹, 0.025 kg ha⁻¹, and 0.018 kg ha^{-1} , respectively; however, the decreases in yields have fluctuated over the years. Obviously, the fluctuating rainfall trend is associated with the fluctuation and decrease in yield of sorghum, millet, and cotton. This suggests that there is a meaningful trend in the association between these crops' yield and rainfall. In the case of sorghum, the results were consistent with a previous study in the Gedaref region, which showed that rainfall shortage or flood can lead to a reduction in sorghum yield (Khalifa, 2016). The findings of this study are similar to those reported by Rowhani et al. (2011), who found a positive correlation between rainfall and sorghum yield in Tanzania, with intra-seasonal variability having a negative impact on yield. Likewise, the results agree with previous results from a study by Ibrahim (2015), who concluded that the fluctuations in the amount of rainfall were the most important factor that influenced the yield of sesame in Sudan. In the case of millet, the yield trend decreased with the decrease in rainfall trend, as reported by Traore et al. (2017). However, the trend showed that the yield reduction of millet was only -0.025 kg ha⁻¹ compared to the other crops that require a high amount of water, such as sorghum. Similarly, correlation analysis revealed that the association between the amount of rainfall and millet yield was weak and not significant, which, again, confirms the low water requirements of this crop. The fluctuation in cotton yield was previously attributed to the variation in the amount of rainfall (Sultan et al., 2008) when the crop is grown under rainfed conditions. Indeed, in the present study, rainfall amount was positively correlated with the yield of all crops except cotton.

When compared to other crops, water requirement by cotton depends on the length of the growing period and the favorite climatic conditions. In addition, continuous rain during the flowering and boll opening stages of cotton crop impairs pollination and thus the final crop yield (Cetin & Basbag, 2010). The trend analysis also showed an increase in sunflower yield over time. This could be attributed to expansion in sunflower cultivation as an industrial crop. The crop is also relatively drought-tolerant and can be grown in various soil types, making it a viable option for areas where other crops may not perform well.

In this study, temperatures were negatively correlated with crop yield regardless of the crop type. A study by Rowhani et al. (2011) in Tanzania reported that the variability in the mean seasonal temperature had a negative effect on sorghum yield. This previous finding agrees with the results of this study, which showed a negative correlation between sorghum yield and the studied temperature-based variables. It has also been reported by Hammer et al. (2015) that high temperature shortens the development time of sorghum, but it also leads to a significant reduction in plant height, pollen viability, and seed set, and, as a consequence, a reduction in crop yield. Therefore, temperature rise, in light of global warming, might reduce the yield of sorghum in Sudan and this may have a serious consequence on food and nutrition security countrywide. The present study also shows that temperature-based variables (Tmin, Tmax, and DTR) were negatively correlated with sesame yield in Gedaref state. This is consistent with the report of Nath et al. (2001), who showed that the ambient temperature of 30°C can negatively affect sesame yield in India. In addition, Kumazaki et al. (2008) showed that day and night temperatures of 23 and 18°C, respectively, affected the stem growth of sesame and that flowering of the crop also did not occur under these unfavorable conditions. Similarly, the temperature had a negative effect on the

yield of millet, sunflower, and cotton. It is anticipated that this could be due to the effects of increased temperature on vegetative growth, flowering, and grain or boll-filling stages.

Although the multiple linear regression results showed few significant relationships among climate variables and crop yield, the regression coefficients can be used to determine the effects of the studied climatic variables on the yield changes of the five crops (Poudel & Shaw, 2016). For example, an increase in Tmin by 1°C led to a reduction in the yield of sesame and sorghum by 23.3 and 260.2 kg ha⁻¹, respectively. In addition, the sign of the regression coefficients in the regression model can indicate the direction of change in the yield versus climate variable changes (Nicholls, 1997). The multiple linear regression model captured between 6% and 70% variability in crop yield as a function of climatic factors. This indicates that the variation in the yield is well explained by climatic variables, except for cotton and sesame, where the model captured only 6% and 41% of their yield variations. The rest of the variations in yield that the model could not capture as a function of the climatic variables could be explained by the variance that is due to other factors, such as fertilizer and pesticide application and weed control, among other confounding factors. Also, it has been demonstrated that plant density on the farm is one of the most important factors that influence sesame yield (Öztürk & Şaman, 2012). A study by Ali et al. (2020) in Gedaref showed that planting sesame at 5 cm between rows could increase its yield by 210.18 kg ha⁻¹. For sorghum, the results showed that the climatic variables were responsible for 70% of the variation in its yield. This finding is in agreement with several studies, which revealed that climatic variables are the most important factors affecting sorghum yield, particularly rainfall (Msongaleli et al., 2017; Mundia et al., 2019). Indeed, the model showed that an increase in rainfall amount by 1 mm led to an increase in sorghum yield by 0.11 kg ha⁻¹. Also, the results depicted 54% in millet yield change, which is explained by the studied climatic variables. For the sunflower, the results proved that 61% of its yield change variation could be due to climatic variables. This result corroborates the findings of Mijić et al. (2012), who indicated that rainfall before and during the vegetation period has a great effect on sunflower yield.

Overall, the present study has utilized secondary data and no ground survey was conducted for primary data collection. As assessed by the descriptive statistical and normal disruption analyses, the quality of the secondary data met the hypothesis that secondary data should not be highly skewed, with a few outliers, and somewhat fit a normal distribution (Figures 3.4 and 3.5). However, some data, such as Tmin and yield of sesame, were either slightly skewed or deviated from a normal distribution. It is worth noting that the long-term (\geq 35 years) secondary data were consistently collected with no missing values. This study promotes the movement of open data science, data sharing, and re-use for addressing further research questions. However, the study focused only on climatic factors rather than incorporating other factors that influence yields, such as soil properties and farming practices. Such factors could be included in other modelling approaches such as DSSAT and production function. Hence, the crop yield estimate models should be interpreted with some caution, as their certainties (Cross-validated R²) were not high. In this context, it is recommended that the crop yield estimate models should further be assessed using an independent test dataset collected at different points in time (e.g., 2019–2021).

4.4. Conclusions

In conclusion, this study shows that the annual Tmin and Tmax had increased by 0.04°C and 0.03 °C per year in the period between 1980 and 2018, while DTR decreased by 0.02°C per year in Gedaref state. Furthermore, the state had cold and warm years between 1984 and 2000 and 2001 and 2018, respectively, and the length of the rainy season in Gedaref state ranged between 57 and

117 days. The trend of annual yield for sorghum had significantly decreased, while sunflower yield had increased in the period between 1970 and 2018. Temperature variables had a negative relationship with the yield of all crops, while an increase in the amount of rainfall significantly increased the yield of sorghum, sesame, and sunflower. Moreover, the increase in the length of the rainy season significantly increased the yield of sesame, sorghum, sunflower, and cotton. There was high variability in crop yields, for example, over 50% variability in the yield of sorghum (R^2 = 0.70 and cross-validated R^2 = 0.69), millet (R^2 = 0.54 and cross-validated R^2 = 0.54) and sunflower (R^2 = 0.64 and cross-validated R^2 = 0.62), which could be related to climatic variables. These findings could be used to support awareness creation amongst different stakeholders and policymakers on the impacts of climate variability and change on crop production and the need for resource allocation to support the uptake of adaptation practices that ensure resilience amongst agricultural communities within the state.

4.5. Summary

It is projected that, on average, annual temperature will increase between 2° C to 6° C under high emission scenarios by the end of the 21^{st} century, with serious consequences in food and nutrition security, especially within semi-arid regions of sub-Saharan Africa. This study aimed to investigate the impact of historical long-term climate (temperature and rainfall) variables on the yield of five major crops *viz.*, sorghum, sesame, cotton, sunflower, and millet in Gedaref state, Sudan over the last 35 years. Mann–Kendall trend analysis was used to determine the existing positive or negative trends in temperature and rainfall, while simple linear regression was used to assess trends in crop yield over time. The first difference approach was used to remove the effect of non-climatic factors on crop yields. On the other hand, the standardized anomaly index was calculated to assess the variability in both rainfall and temperature over the study period (i.e., 35

years). Correlation and multiple linear regression (MLR) analyses were employed to determine the relationships between climatic variables and crops yield. Similarly, a simple linear regression was used to determine the relationship between the length of the rainy season and crop yield. The results showed that the annual maximum temperature (Tmax) increased by 0.03°C per year between the years 1984 and 2018, while the minimum temperature (Tmin) increased by 0.05°C per year, leading to a narrow range in diurnal temperature (DTR). In contrast, annual rainfall fluctuated with no evidence of a significant (p > 0.05) increasing or decreasing trend. The yields for all selected crops were negatively correlated with Tmin, Tmax (r ranged between -0.09 and -0.76), and DTR (r ranged between -0.10 and -0.70). However, the annual rainfall had a strong positive correlation with yield of sorghum (r = 0.64), sesame (r = 0.58), and sunflower (r = 0.75). Furthermore, the results showed that a longer rainy season had significant (p < 0.05) direct relationships with the yield of most crops, while Tmax, Tmin, DTR, and amount of rainfall explained more than 50% of the variability in the yield of sorghum ($R^2 = 0.70$), sunflower ($R^2 =$ 0.61), and millet ($R^2 = 0.54$). These results call for increased awareness among different stakeholders and policymakers on the impact of climate change on crop yield, and the need to upscale adaptation measures to mitigate the negative impacts of climate variability and change. The next chapter will present findings on small-holder farmers' perceptions of climate variability and change, observed climate trends and adaptation measures of climate change in Gedaref state, Sudan.

CHAPTER FIVE

SMALL-HOLDER FARMERS' PERCEPTIONS OF CLIMATE VARIABILITY AND CHANGE, OBSERVED CLIMATE TRENDS AND ADAPTATION MEASURES IN GEDAREF STATE, SUDAN

5.1. Introduction

This chapter presents results on assessment of farmers' perceptions of climate variability and change in Gedaref state, Sudan, and the adaptation measures they use at the local scale to cope with the negative impact of climate change on their agricultural activities. The perception of the small-holder farmers was compared with the meteorological data (rainfall and temperature) to determine how farmers' perceptions mirror climatic trends. The chapter also presents results on the socio-economic factors that influence the choice of climate adaptation measures by the small-holder farmers in Gedaref state, Sudan.

5.2. Results

5.2.1. Demographic and socio-economic characteristics of small-holder farmers

The summary of the demographic and socio-economic characteristics of the sampled respondents is presented in Table 5.1. Out of the 400 sampled small-holder farmers, 58.2% were males, while 41.7% were females. The average age of the respondents was 43.4 years old, with an average of 38.9 years for the respondents live in the village. The average family size of the households comprised six members, with most households headed by men (79.7%) and an average land size of 5.2 hectares. The marital status of the respondents indicated that 73.0% were married, 20.3%

were single, while 6.0% were widowed, separated, or divorced. Among the total sampled respondents, 27% had never attended school, 34.7% had primary school level, 20.5% had attained secondary school education and 17.5% went to college or university education. Farming was the primary source of income and livelihood for 61% of the respondents, followed by 18.5% of whom are in salaried employment, 18.3% were self-employed, and 2.3% were selling forest products. In addition, the average farming experience for the respondents was 19.2 years.

Household characteristics	Statistics
Mean age of the respondent (year)	43.35
Mean of the years of the respondent live in the villages	38.98
Mean family size of the respondents	06.00
Mean land size of the respondent (hectare)	05.21
Gender (%)	
Male	58.25
Female	41.75
Gender of the head of the household (%)	
Male	79.75
Female	20.25
Marital status (%)	
Married	73.00
Single	20.25
Divorced	02.75
Separated	01.00
Widowed	03.00
Education (%)	
Illiterate	27.00
Primary	34.75

Table 5.1: Socio-economic characteristics of surveyed households in the study area (N = 400)

Table 5.1 (cont.):	
Education (%)	
Secondary	20.75
Graduate	17.50
Main sources of income (%)	
Farming	61.00
Salaried employment	18.50
Self-employed	18.25
Selling of forest produce	02.25
Experience in agriculture (years) (%)	
<11	40.75
11 - 20.99	24.25
21 - 40.99	26.50
41 and above	08.50

5.2.2. Small-holder farmers' perception of climate variability and change

More than 90% of the respondents had perceived climate variability and change, while 5.75% disagreed that climate change had not occurred in Gedaref state (Figure 5.1). Among those who perceived that climate change had occurred, 31.4% of them indicated fluctuations/erratic patterns in rainfall trend, while 16.9%, 12% and 11%, respectively, indicated a decrease in rainfall amount, an increase in rainfall amount and late onset of the rainy season (Figure 5.2). The least perception by the farmers was for frequent drought, increase in the length of the dry period and frequency of flooding (Figure 5.2).



Respondents (%)

Figure 5.1: Farmers' perception of climate change



Respondents (%)

Figure 5.2: Indicators that support farmers' perceptions of fluctuations in rainfall.

Figure 5.3 presents farmers' perceptions of temperatures during the summer and winter seasons. About 61% of the respondents indicated an increase in daytime temperature, while 17.7% testified to a decrease in nighttime temperature during the summer (Figure 5.3). Parallel to this, 42% and 28.4% of the respondents perceived an increase in daytime temperature and a decrease in nighttime temperature, respectively during winter (Figure 5.3). The FGDs and key informant interview results revealed that the local communities in Gedaref state were aware of climate change and variability, which were indicated by unpredictable rainfall patterns and amounts, shifting of the onset of the rainy season, reoccurring of dry spells, decrease in rainfall amount, and rise in temperatures.



Figure 5.3: Indicators that support farmers' perceptions of increasing temperatures in summer and winter seasons

5.2.3. Access and source of climate information

Figure 5.4 represents the percentage of respondents who had access to climate information and sources of information within the community. The results showed that 62% of the respondents had access to climate information, while 38% had no access to climate information (Figure 5.4A). Among those who had access to climate information, the majority of them (66.2%) relied on the radio to access the information (Figure 5.4B). While 14.5% of respondents obtained their climate information from different channels like Televion sets, newspapers, indigenous knowledge and opinion leaders. This was followed by 11.9% who obtained their climate information through

mobile phones (Figure 5.4B). During the FGDs, farmers stated that they did not receive advice on climatic conditions and agricultural practices through government extension services to help them to adapt to climate change.



Figure 5.4: A) small-holder farmers' access to climate information and **B**) source that farmers used to access climate information

5.2.4. Trends in climatic variables (rainfall and temperature) of the metrological station

Mann–Kendall and Sen's slope trends for the annual rainfall and mean temperature data for Gedaref state between 1984 and 2018 are presented in Table 5.2 and Figure 5.5. The metrological data showed that annual rainfall trends increased by 0.96 mm per year. However, this increment was not significant (p = 0.41), indicating fluctuation in the amount of rainfall over the years (confidence interval 34.200 – 41.080). In contrast, the annual mean temperature significantly increased by 0.04°C per year, with a confidence interval ranging between 0.02 and 0.05°C (p = 0.05).

Climatic variable	Range		Sen's slope	95% Confidence	<i>p</i> -Value	
	Minimum	Maximum		Interval	I	
Rainfall	425.400	753.8000	0.9627	-34.200 - 41.080	0.41	
Mean temperature	28.458	30.609	0.038	0.022-0.052	0.05	

Table 5.2: Estimated Sen's slope values for annual rainfall (mm) and mean annual temperature (°C) trends in Gedaref State, Sudan, between 1984 and 2018.



Figure 5.5: Annual rainfall and mean temperature trends in Gedaref state, Sudan, between 1984 and 2018.

5.2.5. Small-holder farmers' adaptation measures

Farmers in Gedaref state used different adaptation measures to adapt to climate variability and change (Table 5.3). The most used adaptation measure was crop rotation (18.6%), followed by early cultivation (17.2%), mixing farming (14.5%) and cultivation of short-maturing crop varieties (14.2%). The use of underground water, rainwater harvesting, and practice of supplementary irrigation were the least adaptation measures used by the small-holder farmers in Gedaref state (Table 5.3).

Adaptation measures		onses	% of
Adaptation measures	Ν	%	Cases
Early cultivation	305	17.2	76.4
Delayed cultivation	115	6.5	28.8
Cultivation of short-maturing varieties	253	14.2	63.4
Intercropping	19	1.1	4.8
Crop rotation	330	18.6	82.7
Soil conservation	103	5.8	25.8
Practice supplementary irrigation	4	0.2	1.0
Use of technologies as fertilizers and pesticides	130	7.3	32.6
Introduce drought-tolerant varieties	106	6.0	26.6
Use of improved varieties	58	3.3	14.5
Rainwater harvesting	9	0.5	2.3
Cover the soil around the plant with straw, stone, plastic and/or crop residues to facilitate water infiltration and decreased water evaporation	78	4.4	19.5
Mixing farming	258	14.5	64.7
Use of underground water	10	0.6	2.5

Table 5.3: Adaptation measures used by the respondents to adapt to climate change (N= 400)
5.2.6. Factors influencing small-holder farmers' choice of adaptation measures

Factors influencing farmers' choice of adaptation measures to adapt to climate variability and change are presented in Table 5.4. The result depicted that the sex of the respondent significantly influenced the choice of early cultivation (p < 0.05), cultivation of short maturation variety, soil conservation, use of technology such as fertilizer and pesticide, introducing drought tolerant variety and mixing farming (Table 5.4). Farming experience positively affected soil conservation, introducing drought tolerant varieties and the use of improved varieties. The number of years that the respondent lived in the village significantly influenced the probability of adopting early cultivation and introduction of drought tolerant as adaptation measures (p < 0.05).

Table 5.4: Multinomial logistic regression estimates for the factors that influence the choice of adaptation measures by small-holder farmers in Gedaref state

Adaptation Option	Independent Variable	Estimate	Std Frror	Wald	<i>p</i> -Value	95% Confidence Interval	
	independent variable	Listimute	5			Lower Bound	Upper Bound
Early cultivation	Sex	0.740	0.327	5.111	0.024	1.103	3.982
	Number of years living in the village	-0.025	0.011	5.437	0.020	0.955	0.996
Delay cultivation	Education	0.347	0.125	7.681	0.006	1.107	1.809
Cultivation of short maturation	Sex	-0.862	0.354	5.931	0.015	0.211	0.845
variety	Head of the household	2.287	0.394	33.650	0.000	4.545	21.313
Intercropping	Age	-0.048	0.023	4.357	0.037	0.911	0.997
	Sex	2.379	0.371	41.181	0.000	5.221	22.336
Soil concernation	Farming experience	0.053	0.013	15.471	0.000	1.027	1.082
Son conservation	Main sources of income	0.461	0.162	8.144	0.004	1.155	2.177
	Family size	-0.088	0.043	4.146	0.042	0.841	0.997
	Sex	-1.007	0.316	10.166	0.001	0.197	0.678
	Head of the household	1.315	0.369	12.696	0.000	1.807	7.672
Use of technology such as fertilizer and pesticides	Education	-0.275	0.116	5.590	0.018	0.605	0.954
-	Main sources of income	0.340	0.140	5.867	0.015	0.541	0.937
	Access to climate information	0.730	0.233	9.843	0.002	1.315	3.276

Adaptation Option	Independent variable	Estimate	Std Frror	Wald	<i>p</i> -Value	95% Confidence Interval	
	independent variable	Listimute	544. 11101			Lower bound	Upper bound
	Sex	1.581	0.340	21.634	0.000	2.496	9.460
Introduce drought tolerant variety	Number of years living in the village	-0.030	0.016	3.646	0.056	0.941	1.001
	Age	0.034	0.018	3.604	0.058	0.999	1.073
	Farming experience	0.027	0.012	4.998	0.025	1.003	1.053
	Main sources of income	0.401	0.156	6.634	0.010	1.101	2.028
Use of improved varieties	Farming experience	0.042	0.016	6.686	0.010	1.010	1.076
Rainwater harvesting	Age	0.204	0.106	3.717	0.054	0.997	1.508
Cover soil around the plant	Family size	0.119	0.048	6.096	0.014	1.025	1.237
	Sex	-0.987	0.311	10.059	0.002	0.203	0.686
Mixing farming	Land tenure	0.664	0.188	12.512	0.000	1.345	2.807
	Family size	-0.084	0.039	4.562	0033	0.851	0.993

5.3. Discussion

This study assessed small-holder farmers' perceptions of climate variability and change in Gedaref state and the factors that influence their choice of a specific adaptation measure. The results indicate that the local communities in Gedaref state perceived changes in climate variables (rainfall and temperature) over time. About 31.4% of the farmers were aware of rainfall fluctuation, while 11% perceived late onset of the rainy season over the last decades. These perceptions in rainfall align with the actual annual rainfall trend between 1984 and 2018, which showed a fluctuation in the amount of rainfall ranging from 425.4 to 753.8 in Gedaref region. These results agree with the findings of Glover and Elsiddig (2012) who reported that annual rainfall in Gedaref state was erratic and scarce in the past decades, ranging between 400 and 800 mm. However, 16.9% and 12%, respectively, indicated a decrease and increase in rainfall, which could be attributed to the variation in the rainfall amount from one location to another. Apart from the rainfall, farmers also correctly perceived increased temperature during the summer and winter seasons, confirming the results from the annual mean temperature trends of the metrological records in Gedaref between 1984 and 2018. These results were also supported by Issa (2018) who reported a significant increase in temperature in the study area between 1972 and 2017. This was further confirmed by key informant interviews and FGDs who highlighted similar rainfall and temperature trends in the study area. The FGDs also pointed out that these changes negatively affected agricultural activities in Gedaref state.

The majority of respondents who were aware of climate change (92%) had adopted some adaptation practices to minimize the impacts of climate change on their agricultural activities. These farmers used one or more adaptation measures to cope with the negative impact of climate change in Gedaref state. The majority (18.6%) of the farmers practiced crop rotation as an

adaptation measure. Crop rotation improves soil nutrients, increasing the biomass that enhances soil health water holding capacity as well as interrupts the life cycles of the pests. Marini et al. (2020) reported that crop rotation is one of the adaptation measures that increase cereal yields under a changing climate. In Gedaref state, sorghum is the most dominant cultivated crop and the main staple food crop for the entire country. This may be one factor that leads small-holder farmers to mostly practice crop rotation, compared to other adaptation measures to increase yield (Marini et al., 2020). The use of crop rotation might also be attributed to the fact that small-holder farmers had limited access to agricultural technologies and inputs that were unaffordable to most of them. Apart from crop rotation, 17.2% of the farmers applied early cultivation as a measure to adapt to climate change. These findings are in line with the earlier result of Mohmmed et al. (2018), who reported that the farmers in Gedaref state had changed their planting dates as an adaptation option to deal with climate change and variability. In addition, 14.5% of respondents practiced mixed farming, which includes crop farming and livestock raising, as an adaptive strategy. The farmers during the FGDs explained that keeping livestock was a buffer against crop losses during drought or flooding events. For example, they sell milk or chicken that could support the household in overcoming a poor crop harvest as well as contributes to household food security. Moreover, 14.2% of the farmers indicated that adoption of the cultivation of short-maturing varieties as a climate adaptation measure. This could be explained by rainfall being limited to a short period in Gedaref state (Osman et al., 2021).

Although farmers listed various adaptation measures to climate change, the choice of a specific adaptation option by a farmer was dependent on several factors. In this study, the choice of the cultivation of short maturation variety and the use of technology such as fertilizer and pesticide were significantly influenced by the sex of the head of the household. This mainly could be

attributed to the fact that most farmers (79.8%) were male-headed. These findings agreed with the study by Abdalla et al. (2013) who stated that male-headed households were the predominant group in Sudan. The FGDs have highlighted that most key decision-makers in Gedaref were men on when what and where to plant and use during the cultivation season. Also, the results of the present study indicated that the age of respondents was strongly related and affected the choice of intercropping, introducing drought tolerant variety and rainwater harvesting. This suggests that older farmers were more likely to adopt these adaptation measures, which could be explained by long farming experiences that allow them to perceive environmental changes. Deressa et al. (2008) reported that the age of the farmers could be used to detect the farming experience in the Nile Basin of Ethiopia. The authors also indicated that the age of the farmers had a positive relationship with taking up adaptation measures. In this study, the number of years farmers lived in their villages significantly influenced the choice of early cultivation and the use of drought tolerant variety. Farmers with a higher number of years living in the villages had adequate knowledge about the area and the nature of rainfall. This could help them select an appropriate adaptation measure depending on the onset of the rainy season.

In this study, 62% of the respondents had access to climate information and more than half of them (66.2) received climate information through the radio. This is due to the fact that radio was the most accessible and popular medium of communication in rural areas because even illiterate farmers can access and understand climate information in their local language. This result confirmed the finding of Ndavula and Lungahi (2018) who concluded that radio played a crucial role in many African countries as an effective information dissemination means. The authors further stated that rural populations favor radio and consider it to be the most significant and accessible media.

Furthermore, farmers with more extended farming experience were more likely to adopt soil conservation, use of drought tolerant varieties and improved varieties. Many previous studies had reported similar funding. For instance, Adimassu and Kessler (2016) concluded a significant positive relationship between years of experience in agriculture and farmers' adoption of improved agricultural technologies such as drought tolerant and improved varieties. While Alhassan et al. (2019) and Mwungu et al. (2020) reported that farmers with more farming experience were more likely to use adaptation methods to increase their resilience under environmental changes. The mono-cropping of sorghum in Gedaref state coupled with fluctuation in the amount of rainfall might be the major factors that contributed to the adoption of soil conservation, the use of drought tolerant varieties and improved varieties by small-holder farmers. The main source of income was significantly and positively related to soil conservation, use of technology such as fertilizer and pesticide and introducing drought tolerant variety. About 60% of the respondents occupied farming as a primary source of their income, and this could be the reason of the use of these adaptation measures to increase their crop yield and consequently, income. The results of the present study also showed that education level significantly influence the adoption of delayed cultivation and the use of technology such as fertilizer and pesticides. These results were consistent with a study in Kenya where the farmers' education had a significant relationship with adopting practices to adapt to climate change (Nyang'au et al., 2021). This is due to the fact that more educated farmers had more access to information about agricultural inputs such as fertilizers and pesticides as well as climate information that could help them to adapt accordingly.

In this study, family size significantly influenced farmers' choice of soil conservation and covering the soil around the plant with straw, stone, plastic and crop residues to facilitate water infiltration and decrease water evaporation. This could be linked to the fact that soil conservation needs sufficient family labor (Tesfaye *et al.*, 2014). Therefore, households with enough family labor to execute soil conservation measures were more likely to participate in the implementation of soil conservation as an adaptation measure. The result of this study depicted that land tenure had a significant positive relationship with adopting mixed farming to adapt to climate change and variability. This is mostly because land tenure was critical to adaptation due to the fact that landowners embraced new technology faster than renters (Fosu-Mensah *et al.*, 2012). Furthermore, the results reveal that access to climate information significantly influenced the use of technology such as fertilizer and pesticides as adaptation practices. Zamasiya et al. (2017) reported similar results that access to climate information increased small-holder farmers' adoption of climate change adaptation measures. During the key informant interviews and FGDs, the local community highlighted that applying manure to their farms as a climate change adaptation measure to conserve soil fertility and moisture because of the high cost of fertilizers, which they could not afford.

5.4 Conclusions

The findings of this study revealed that farmers in Gedaref state, Sudan are aware of climate variability and change, and they used different adaptation measures to cope with its negative impacts on their farming activities. The most used adaptation measures by the farmers were crop rotation, early cultivation, mixed farming and cultivation of short-maturing varieties. Age, sex of the respondent, education level, sex of the head of the household, number of years living in the village, farming experience, main sources of income, family size, access to climate information and land tenure were the main factors that significantly influenced the choice of adaptation measures. Therefore, government policies that promote climate adaptation strategies in Gedaref state should consider these factors to increase the likelihood of smallholder farmers' adoption of climate adaptation measures to cope with the negative impact of climate change on crop yield. In

addition, the findings of this study will be useful in implementing and monitoring adaptation measures through the integration of farmers' perceptions with the scientific findings to improve crop yield in Gedaref state and other rainfed agricultural areas in Sudan.

5.5 Summary

Climate projections indicate that sub-Saharan Africa will experience significant climatic changes, including extreme drying and warming, particularly in arid and semi-arid regions. Although many studies have reported the impact of climate change and variability on agricultural systems, few studies have examined perceptions of local communities on climate change and variability, and their roles in strategic decision making and developing appropriate adaptation measures for mitigation of local impacts on producers. Therefore, this study aimed to assess farmers' perceptions of climate change and their adaptation measures at the local level in Gedaref state; compare smallholder farmers' perceptions with the meteorological records to determine how farmers' perceptions mirror climatic trends. Four-hundred respondents were interviewed using semi-structured questionnaire to gather data on the local community's perception of climate variability and change, its impacts on agricultural activities, and adaptation measures employed to mitigate such impacts. This was followed by key informant interview and 16 focus group discussion. More than 90% of the respondents perceived that Gedaref state had experienced climate variability and change. About 61% of the respondents indicated an increase in daytime temperature, which was confirmed with meteorological records, which showed that annual mean temperature in Gedaref state had increased by 0.04°C per year. The adaptation measures practised by most farmers in Gedaref state were crop rotation, early cultivation, mixed farming and cultivation of short-maturing varieties. The main factors that significantly influenced choice of adaptation measures included age, sex of respondent, education level, sex of head of household,

number of years live in villages, farming experience, main sources of income, family size, access to climate information and land tenure. Findings from this study are useful for policy formulation on mitigation of climate change impacts and enhancing farmers' resilience to climate change, consequently increasing crop productivity and sustained livelihoods.

CHAPTER SIX

MAPPING, INTENSITIES AND FUTURE PREDICTION OF LAND USE/ LAND COVER DYNAMICS USING GOOGLE EARTH ENGINE AND CA-ARTIFICIAL NEURAL NETWORK MODEL

6.1 Introduction

This chapter presents results on evaluation of LULC changes in the Gedaref state, Sudan for the period between 1988–2018 using Landsat imageries and the random forest classifier. It also determines the underlying dynamics that caused the changes in the landscape structure using intensity analysis. Moreover, this chapter predicts the future LULC outlook for the years 2028 and 2048 in Gedaref state using cellular automata-artificial neural network (CA-ANN) algorithm.

6.2 Results

6.2.1 Land use/ land cover (LULC) classification

The classified LULC maps and associated area statistics under each class category for 1988, 1998, 2008 and 2018 are presented in Figure 6.1 and Table 6.1. Among the LULC categories, cropland was the most dominant in 1988, followed by grassland; each occupying 78.64%, and 19.64% of the total area, respectively (Table 6.1). Whereas forest, water and settlement covered less than 2% of the studied landscape (Table 6.1). A similar trend was observed for the other studied years. Nevertheless, there was a distinct LULC change in Gedaref during the study period, where an



expansion in cropland and settlement area and a decline in forest and grassland areas were observed (Figure 6.1 and Table 6.1). Water area increased in all study years except in 2008.

Figure 6.1: Classified Land use/ land cover (LULC) maps of Gedaref state for the years 1988, 1998, 2008 and 2018 produced using multi-date Landsat images and random forest classification algorithm

Table 6.1: Area (ha) and percent cover (%) of each land use/ cover (LULC) class in Gedaref state for the years 1988, 1998, 2008, and 2018 estimated using multi-date Landsat images and random forest classification algorithm

1988 Year			1998		2008		2018		
	Area		Area		Area		Area		
LULC	ha	%	ha	%	ha	%	ha	%	
Cropland	5023958.04	78.64	5289566.13	82.8	5566904.46	87.14	5723662.45	89.59	
Forest	0069053.13	01.08	0037217.16	00.58	0035406.99	00.55	0030056.44	00.47	
Grassland	1254479.85	19.64	1012649.94	15.85	0734874.57	11.51	0548998.87	08.59	
Water	0034132.05	00.53	0039363.12	00.62	0035961.84	00.56	0057366.22	00.89	
Settlement	0006739.02	00.11	0009565.74	00.15	0015214.23	00.24	0028278.13	00.44	
Total	6388362.09	100	6388362.09	100	6388362.09	100	6388362.09	100	

6.2.2 Accuracy of land use/ land cover (LULC) classification

The accuracy evaluation metrics of the classified LULC maps generated from the confusion matrix is presented in Table 6.2. The overall accuracy of the 1988, 1998, 2008 and 2018 LULC maps was 81.75%, 83.28%, 85.15%, and 87.70%, respectively. While F1 score value for all LULC classes in all years ranged between 80% and 90% (Table 6.2). Additionally, the results of quantity disagreement for all the classified maps ranged between 3% and 4%, whereas allocation disagreement varied from 9% to 14% (Table 6.2).

Table 6.2: Overall and individual class accuracies of land use/ land cover (LULC) maps of Gedarefstate for the years 1988, 1998, 2008 and 2018.

Class	Producer's accuracy (%)	User's accuracy (%)	F1 score (%)	Overall accuracy (%)	Allocation disagreement (%)	Quantity disagreement (%)
			1988			
Cropland	88.88	77.41	82.75			
Forest	81.13	81.39	81.39	81.75	14	4
Grassland	85.29	80.55	82.85			

Table 6.2 (cont.):

Class	Producer's accuracy (%)	User's accuracy (%)	F1 score (%)	Overall accuracy (%)	Allocation disagreement (%)	Quantity disagreement (%)
Water	81.13	81.13	81.13	81 75	14	4
Settlement	73.13	89.09	80.32	01.72		
			1998			
Cropland	83.07	83.07	83.07			
Forest	83.13	87.34	85.18			
Grassland	84.48	77.77	80.99	83.28	14	3
Water	86.66	81.25	83.87			
Settlement	78.78	86.66	82.53			
			2008	1	I	
Cropland	95.23	82.19	88.23			
Forest	86.07	85.00	85.53			
Grassland	76.56	87.50	81.63	85.15	11	4
Water	88.88	87.67	88.27	-		
Settlement	76.92	83.33	80.00			
			2018	l		
Cropland	89.28	80.64	84.74			
Forest	89.07	88.33	88.70			
Grassland	84.78	81.25	82.97	87.70	9	3
Water	79.20	94.11	86.02			
Settlement	92.40	88.48	90.40			

6.2.3 Land use/ land cover (LULC) change detection

Forest and grassland categories were considerably decreased by about 46% and 19%, respectively, in the first 1998 – 2008 interval (Table 6.3). On the other hand, there was an expansion in cropland

area during the whole study period, which ranged between about 2.8% and 5% in the three intervals. Similarly, the settlement areas dramatically increased by 41.94% and 59.04% in the first and second intervals, respectively, with the third interval being drastically greater (85.86%) than that of the first and second intervals. Water class increased by 15.32% in the first period of the study and decreased during the second time period, with a sharp increase (85.86%) in the third interval compared to the increase in the first interval.

Table 6.3: Land use/ land cover (LULC) change estimates (area and percentage) for Gedaref state for the 1988–1998, 1998–2008, 2008–2018 and 1988–2018.

Year	1988 - 1	998	1998 - 2	008	2008 - 2018		1988 - 2	018
LULC	Area	L	Area Area		Area		Area	
LeLe	ha	% ha % ha %		%	ha	%		
Cropland	265608.00	5.28	277338.30	05.24	156758	2.81	699704.35	13.92
Forest	-31835.97	-46.10	-1810.17	-04.86	-5350.55	-15.11	-38996.70	-56.47
Grassland	-241829.91	-19.27	-277775.37	-27.43	-185875.71	-25.29	-705480.98	-56.23
Water	5231.07	15.32	-3401.28	-8.64	21404.38	59.52	23234.17	86.07
Settlement	2826.72	41.94	5648.49	59.04	13063.9	85.86	21539.11	319.61

6.2.4 Land use/ land cover (LULC) transitions mapping

The results in Figure 6.2 indicate the transformation of each of the five LULC classes during 1988–2018, 1998–2018 and 2008–2018. The major LULC transition that took place over the study period (1988–2018) were forest to cropland, grassland to cropland, cropland to grassland, water to cropland, cropland to water, cropland to settlement and grassland. In particular, the dominant transition in Sothern and Western parts of Gedaref sate across the three transition periods (1988–2018, 1998–2018 and 2008–2018) was forest to cropland. Whereas the transition from grassland to cropland was mainly observed in Northern and Northeast of Gedaref state (Figure 6.2). The

transition from cropland to settlement was mainly occurred in Central, Northwest, and Southern parts of Gedaref state. $\underbrace{\overset{34^{\circ}0'0''E}{\underbrace{35^{\circ}0'0''E}} \underbrace{35^{\circ}0'0''E}{\underbrace{1ULC \text{ transition 1988 - 2018}} \underbrace{\overset{2}{\underbrace{5}} \underbrace{34^{\circ}0'0''E}{\underbrace{5}} \underbrace{34^{\circ}0'0''E}{\underbrace{1ULC \text{ transition 1998 - 2018}} \underbrace{\overset{2}{\underbrace{5}} \underbrace{1ULC \text{ transition 1998 - 2018}}_{\underbrace{5}} \underbrace{ULC \text{ transition 1998 - 2018}}_{\underbrace{$



Figure 6.2: Land use/ land cover (LULC) transitions of Gedaref state during 1988–2018, 1998–2018 and 2008–2018

6.2.5 Intensity analysis in land use/ land cover (LULC) transitions

Interval level intensity results showed the changing intensity over each time period (Figure 6.3A) and the annual change between intervals (Figure 6.3B). Interval analysis revealed that the period 1988–1998 experienced fast LULC transitions and annual change rates. In the second interval, the observed and annual transitions were relatively equal to the uniform line but lower than in the third interval indicating slow LULC transitions.



Figure 6.3: Interval level intensity of land use/ land cover (LULC) change for 1988 1998, 1998–2008 and 2008–2018. A) the percent of the area that changed over each interval and B) the percentage of the area that annually changed during each interval

Figure 6.4 illustrates that the LULC classes experienced dormant or active changes during the

study period. Moreover, the figure shows that the active LULC classes were the ones that their gain or loss is crossing the uniform line. In contrast, the dormant categories are those of gain or loss that do not reach the uniform line. During the three intervals, forest, grassland, water and settlement categories were active gainers with relatively higher gains in the settlement, forest and water, respectively. However, cropland category was the dormant gainer throughout the study period. Three categories, i.e., forest, grassland and water, were active losers during the three intervals with relatively higher losses in forest and grassland respectively. Whereas settlement was an active loser during the first interval and a dormant loser during the second and third intervals. The cropland category was the dormant loser during the three intervals.



Figure 6.4: Land use/ land cover (LULC) category level intensity for 1988–1998, 1998–2008, 2008 and 2018

Figures 6.5–6.7 demonstrate the results of the transition level analysis for each of LULC category. The vertical dashed lines on both sides of the chart represent hypothetical uniform intensity lines. The left side of the uniform line explains the theoretical uniform value in the transition intensity that accounted for the losses in the specific LULC classes. While the side on the right represents the gains in transition intensity. The intensity showed that the expansions in cropland in 1998 targeted forest only and losses in cropland targeted both forest and settlement (Figure 6.5A). The gain and loss in forest areas targeted water and cropland (Figure 6.5B). Likewise, the transition to grassland in 1998 targeted settlement and avoided the other LULC categories, while the losses in grassland targeted water (Figure 6.5C). Losses in water targeted cropland and forest, whereas the gains in the water category targeted forest, grassland, and settlement (Figure 6.5D). The reductions and expansions in cropland between 1998 and 2008 followed a similar trend to that of 1988-1998, where this category targeted forest areas (Figure 6.6A). On the other hand, the expansions in settlements in the same period targeted cropland, but reductions in settlement area targeted grassland and water equally (Figure 6.6E). Gains in water targeted forest and slight grassland, and losses targeted forest (Figure 6.6D).



Figure 6.5: Transition level intensity of land use/ land cover (LULC) for the period 1988–1998. (A) cropland, (B) forest, (C) grassland, (D) water and (E) settlement (gains on the right and losses on the left)



Figure 6.6: Transition level intensity of land use/ land cover (LULC) for the period 1998–2008. (A) cropland, (B) forest, (C) grassland, (D) water and (E) settlement (gains on the right and losses on the left.

The final ten years of the studied period (2008–2018), cropland had different transition intensity trends from the first and second periods and relatively similar transition intensity trends for the forest. In contrast, during this period, water experienced an expansion with transition intensity targeting forest and grassland (Figure 6.7). The intensity of water gained from grassland was more significant than for forest. Settlements expansion over this period targeted water and marginally cropland (Figure 6.7). Losses in settlement targeted water and grassland with marginal avoidance in cropland. In both cases of losing and gaining in the settlement, the highest transition intensity was from water and to water (Figure 6.7).



Figure 6.7: Transition level intensity of land use/ land cover (LULC) for the period 2008–2018. (A) cropland, (B) forest, (C) grassland, (D) water, and (E) settlement (gains on the right and losses on the left)

6.2.6 Future prediction of land use/ land cover (LULC)

The CA-ANN predicated future LULC for Gedaref state for the years 2028 and 2048 (Figure 6.8) based on the LULC maps of the years 2008 and 2018 that were generated from Landsat images

(Figure 6.8). The model predicted a slight increase in cropland area from 89.59% to 90.43% and a considerable decrease in forest area (0.47% to 0.41%) between 2018 and 2048 (Table 6.4). The model also predicted a marginal decrease in grassland (8.59 % to 7.78%) and an increase in settlement area from 0.44% to 0.50%. Whereas water area was predicted to be relatively consistent (0.89% to 0.88%) (Table 6.4) between 2018 and 2048.



Figure 6.8: Predicted land use/ land cover (LULC) in Gadaref state for 2028 and 2048

Vear	2018		2028 204			8	
i cai	Area		Area		Area		
LULC	ha	%	ha	%	ha	%	
Cropland	5723662.45	89.59	5731266.73	89.71	5777105.53	90.43	
Forest	0030056.44	0.47	0029882.92	0.46	0025895.38	0.41	
Grassland	0548998.87	8.59	0541367.59	8.48	0496928.83	7.78	
Water	0057366.22	0.89	0057093.79	0.89	0056335.45	0.88	
Settlement	0028278.13	0.44	0028751.08	0.45	0032096.92	0.50	
Total	6388362.09	100	6388362.09	100	6388362.09	100	

Table 6.4: The proportion of predicted land use/ land cover (LULC) categories at Gedaref state

The validation results of the CA-ANN algorithm showed that the model adequately simulated the future LULC pattern in Gedaref state with an overall correctness of 87% and kappa coefficient of 0.86. This indicates that the simulated LULC map of the year 2018 was highly comparable to the actual 2018 map produced in this study, indicating an accurate model performance.

6.3 Discussion

In the present study, combined on-screen digitization of polygons and points from Google Earth Pro platform, RF classification algorithm and dense multi-date Landsat satellite images (1988-2018) were used to map LULC types in Gedaref state, Sudan. For the first time, LULC for 1988-2018 was mapped in the entire Gedaref state, and its future for the years 2028 and 2048 was predicted. In addition, this is the first attempt to evaluate LULC rate of change, intensity and transition in Gedaref for 30 years (1998 - 2018). The methodology for this study has several advantages as acquiring on-screen polygons and points from Google Earth can reliably be utilized to accurately characterize landscape structure, especially for satellite images that have a medium spatial resolution (Asante-Yeboah et al., 2022; Hurskainen et al., 2019; Taylor & Lovell, 2012). Also, for historical satellite image classification, real-time ground truthing observations might not be available. Hence freely accessible platforms like Google Earth can provide accurate reference data for classification experiments. To minimize the expected spatial autocorrelation, point observations with a distance not less than 100 m from the nearest training polygon were used to validate the classified LULC maps. Hence, this mapping approach provided accurate LULC patterns for Gedaref state, which has a dynamic landscape setup. The overall accuracies for the years 1988, 1998, 2008, and 2018 images ranged between 81.75% and 87.70%, which are greater than the recommended acceptable LULC classification accuracy of 70% (Akalu et al., 2019;

Bhunia *et al.*, 2021). Nevertheless, the progressive increment in the LULC classification accuracy from the initial study year (1988) to the last year (2018) could indicate challenges associated with the availability of historical images of relatively better spatial resolution from Google Earth in the past decades.

The results showed that cropland was the most dominant LULC class covering between 78.64% and 89.59% of the total acreage of Gedaref state. This is due to the fact that Gedaref region is the main rainfed agricultural area of Sudan, where about 80% of the population in this state mainly depends on agricultural production for their livelihood (Osman et al., 2021). In addition, clay soil and the high amount of rainfall (400–800mm) that characterize Gedaref state, offer the optimum conditions for the cultivation of many food and cash crops such as sorghum, millet, sunflower, sesame, and cotton. This also explains the domination of this area by agricultural land. The findings of this study also revealed accelerated LULC change between 1988 and 2018, with an expansion in cropland and settlement, and a decrease in forest and grassland areas. These land use dynamics could be due to human activity, such as the horizontal expansion of settlements and cultivated areas, not only by the local communities but also by the investors and other communities from the neighboring states and countries (Sulieman, 2013). Although there was a substantial increase in cropland, the change in water class was unsystematic over the study period, with a dramatic increase between 2008 –2018 (59.52%). This could be explained by the fluctuation in the amount of rainfall that was observed in the study area due to climate variability and change (Osman et al., 2021).

Specifically, cropland has increased from 78.64% to 89.59% during the study period between 1988 and 2018, while the forest area declined from 1.08% to 0.47% of the total area in Gedaref state. These findings agreed with the results of Gadallah et al. (2020), who has reported a decrease of

forest area in Wad Albashir forest in Al Rahad locality, Gedaref state, from 72.2% in 2001 to 58% in 2017, whereas cropland increased from 25.9% to 38.3% over the same period. This has been confirmed by our transition mapping, which showed a huge transformation of forested area to agricultural land, particularly along Al Rahad River where Wad Albashir forest is located, and in other forested areas in the southern Gedaref state. Generally, the decline in the forest area could be linked to agricultural expansion, firewood, charcoal production, timber, construction, flooding, soil erosion and desertification. The decline in grassland (-56.23%) from 1988 to 2018 in Gedaref region confirms the results of other studies that were conducted in some parts of the state (Sulieman & Elagib, 2012; Sulieman, 2010). This decrease can also be explained by the fact that agricultural areas usually expanded at the expense of grassland. Also, the transition analysis showed substantial transformation of agricultural area at expense of grassland mainly in the northern and northerneastern parts of Gedaref state. The regulations and policies of land use in Gedaref are biased toward cropland, compared to the grassland that was used by the pastoralist for grazing. A study by Sulieman & Elagib (2012) reported that this bias was adopted in the solution of disputes during the British colonial period when the Soil Conservation Committee recommended in 1944 that "where nomadic pastoralists were in direct competition for land with settled cultivators, the rights of the cultivator should be considered as paramount because his crops yield a higher return per unit area". Although new laws and regulations were put in place in Gedaref state, the 1944 recommendation was still implemented in some areas. The results also showed that the settlement areas drastically expanded more than three times (319.61%) between 1988 and 2018. This expansion was mainly taking place in central Gedaref where the capital city is located and along Atbara, Al Rahad, Saiteet, and Basalam Rivers. Likewise, the southern and western parts of the state, where the major agricultural schemes are exist, experienced the same transformation from

cropland to settlement. This expansion in settlement is primarily to meet the demand for shelter for the rapid population growth and the increase of industrial areas in the region. This is coupled with the increase in cropland, which can be explained by the rising demand for food to meet the growing population. Biratu et al. (2022) reported an increase in cropland and settlement areas in Ethiopia between 1986 and 2021 because of the growing population's demand to ensure food and nutrition security in the country.

The unsystematic change in water class in this study could be explained by many seasonal waterways and rivers, such as Atbara, Al Rahad, Saiteet, and Basalam, flowing northward from the Ethiopian highlands in the rainy season, which differs from one year to another. However, the high increase in water class between 2008 and 2018 is associated with the construction of the Upper Atbara and Setit Dam complex, which is a twin dam consisting of two dams: Rumela Dam on the Upper Atbarah River and Burdana Dam on the Setit River (Zarroug *et al.*, 2019). Construction of the dam began in 2011 and was completed in 2016, intending to provide irrigation water for agriculture, potable supply water for the eastern states of Sudan, and power generation.

The interval level intensity revealed an intensive change of LULC in the first 10 years (1988 – 1998) compared to the second (1998 – 2008) and third (2008 – 2018) intervals. However, change in 2008–2018 was slower than in 1998–2008. This indicates that the impacts of socio-economic and physical driving factors during the three decades were different. This also implies a rapid change in Gedaref landscape in the first decade of the study period. The rapid LULC changes correspond to the areal extent of mechanized rainfed farming in the area since the late 1970s, which has attracted some migrants from different parts of Sudan, leading to rapid change in LULC over the first interval. A study by Miller (2005), reported that the development of mechanized agricultural and grain trade in Gedaref during the 1970s enhanced immigration to the state from

different parts of Sudan, which increased socio-economic activities in the region.

The categorical intensity analysis main findings showed that forest, grassland and water were active losers and gainers. This explains that gains and losses of these categories happen at intensities greater than the average intensity of all LULC categories. The intensity of active losses and gains in grassland and water could be explained by the associated seasonality of these two categories with the rainy season in Gedaref state. In contrast, the active gaining and losing in forest can be explained by the attempts of controlling the invasive Mesquite trees (*Prosopis juliflora*). This plant rapidly invades and colonizes the uncultivated land and farmers mostly cut down Mesquite trees to clear the land for cultivation, which might result in a fluctuation in forest cover. Secondly, the settlement was active gaining and losing between 1988 and 1998, active gaining and a dormant loser in the 1998–2008 and 2008–2018 periods. This could be linked to lack of visual clarity of settlement pixels when reference data were gathered from Google Earth platform. In addition, the nature of the hut houses (locally called *quttiyya*), which are made of wood, grass and reeds might have been mixed up with the forest class in the classification process. Thirdly, cropland continued as a dormant loser and dormant gainer throughout the studied three intervals. This is because cropland accounts for the biggest percentage of the landscape compared to other LULC categories.

Analysis of intensity at the transition level revealed that cropland gains from forest were higher between 1988 and 1998 compared to the gain in cropland from the similar LULC category over the second study period (1998 – 2008). However, cropland avoided gains and losses of forest in the third interval (2008 - 2018). This suggests that the expansion in cropland resulted in a decline in the forest area. Nevertheless, the loss of agricultural area to the forest is associated with the spreading of Mesquite in cropland as we mentioned earlier. This study also showed the transition of forest to water in the three change periods. This might be linked to misclassification between the two classes as most of the forests in Gedaref state are Nilotic trees such as Acacia nilotica and Acacia seyal, which lie mainly around the seasonal rivers and watercourses. Similarly, the gains in grassland from the settlement in the first period (1988–1998) could be a likely result of the slight misclassification of wood, grass and hut houses as grassland. The second explanation is that the low resolution of the satellite image for 1988–1998 might have contributed to the misclassification of settlements as grassland, as the result of other periods did not show such intensity. The intensity analysis also showed the gain of grassland from water in the second and third periods (1998-2008 and 2008–2018). This is explained by the fact that grassland overlaps with water due to the seasonality of water bodies in Gedaref state. Another interesting key finding was observed in grassland losses to water, which was higher in the period between 2008 and 2018, compared to the losses in grassland to the similar LULC category over the first and second periods. This is perhaps as a result of the construction of the Upper Atbara and Setit Dam complex and the increase of water storage ponds for drinking water, domestic use and irrigation of agricultural land during this period. The settlement loss targets water and grassland in the three study periods, with the addition of the forest in the second period (1998–2008). This could be linked to pastoralism movements through seasonal migration routes and settling where there are grass and water. Whereas gaining in settlement target cropland and water in the second and third periods, respectively, with no gain from any LULC class in the first period. Settlement is usually located in flat areas, where cropland can also be found in these areas. Hence, the expansion in settlement areas is likely to target cropland and water gathering sites.

The predicted results revealed that by 2028 and 2048, cropland is expected to increase by 0.12% and 0.72%, respectively as compared to 2018. This is partly attributed to the expansion of

mechanized farming in the study area. Similarly, the settlement might be increased from 0.44 to 0.50% between 2018 and 2048. This could be attributed to the rapid population growth and refugee influx in the study area (Gadallah *et al.*, 2019). Whereas the forest acreage is anticipated to decrease from 0.47% in 2018 to 0.41% in 2048. This could be linked to illegal cutting, overgrazing and mechanized agriculture since the population in Gedaref state depend mainly on farming activities for livelihoods (Abdalla, 2018; Idreas, 2015). The model prediction showed that the trend of grassland will continuously decrease by 7.78% in 2048 from 8.59% in 2018. Water will be relatively consistent in the region in 2018, 2028 and 2048 by 0.89%, 0.89% and 0.88%, respectively.

The findings presented in this study could guide policymakers and different stakeholders to effectively plan and manage the landscape in Gedaref state, Sudan. For example, to ensure that human settlement does not completely eliminate crop farming in Gedaref State, the government should consider the following strategies: implement land use planning and zoning regulations, promote sustainable agricultural practices, strengthen legal frameworks and enforcement, develop rural infrastructure and services, promote agroforestry and reforestation, community involvement and education, and support research and development. Also, the study provides some insights on the main drivers that could play a vital role in changing the current and future landscape structure in most important rainfed farming areas in Sudan. Knowing the areal extent, change rate, intensity and transition of important LULC categories like cropland and grassland over 30 years could enable informed crop and grassland production monitoring. Specifically, the results of the present study could complement the findings of chapter four, which predicted the relationship between climate factors and crop yield in Gedaref state under climate warming. Hence, total crop production in Gedaref could be predicted and forecasted using both study findings. In general,

different land use and environmental policy and planning initiatives in Gedaref state or even in Sudan could make use of the findings presented in this study.

Despite the fact that this study used a robust and efficient machine learning RF algorithm in LULC classification experiment, the method has some limitations. For instance, when a large number of decision trees are used, the algorithm can be too slow to make the classification predictions as it requires more computational power. Another disadvantage of RF is that the method is a black or grey box approach with very little control over what the algorithm does (Feng *et al.*, 2020). On the other hand, the simulated the future (2028 and 2048) LULC predictions using the current natural and anthropogenic factors that might considerably change in the future. In addition, other factors that could play a significant role in the future LULC shift like fire, flood, conflict, and other geopolitical and socio-economic variables were not considered in our study.

6.4 Conclusions

In conclusion, this study is the first attempt to map and predict future LULC changes and their intensities and underline the processes that cause the change in the landscape of Gedaref state. The results showed that LULC in Gedaref has undergone a distinct change in 30 years period (1988 – 2018), with a considerable decline in forest and grassland areas and an expansion in settlement areas. The classified LULC maps and model validation for future LULC prediction provided high accuracy (overall correctness = 87%). This demonstrates the possibility of mapping and predicting LULC classes using on-screen reference data from Google Earth images, dense multi-date Landsat images, RF classifier and CA-ANN model. The findings of this study provide information on LULC patterns in Gedaref region that could be useful for designing management plans and developing policies for assessing and monitoring crop and grassland production, other natural

resources produce, landscape fragmentation and degradation, and ecosystem functions. This information is, therefore, critical in managing one of the most important rainfed agricultural landscapes in Sudan.

6.5 Summary

Mapping of LULC dynamics has gained significant attention in the past decades. This is due to the role played by LULC change in assessing climate, various ecosystem functions, natural resource activities and livelihoods in general. In Gedaref landscape of Eastern Sudan, there is limited or no knowledge of LULC structure and size, degree of change, transition, intensity and future outlook. Therefore, the aims of the current study were to (1) evaluate LULC changes in the Gedaref state, Sudan, for the past thirty years (1988-2018) using Landsat imageries and the random forest classifier, (2) determine the underlying dynamics that caused the changes in the landscape structure using intensity analysis, and (3) predict future LULC outlook for the years 2028 and 2048 using cellular automata-artificial neural network (CA-ANN). The results exhibited drastic LULC dynamics driven mainly by cropland and settlement expansions, which increased by 13.92% and 319.61%, respectively, between 1988 and 2018. In contrast, forest and grassland declined by 56.47% and 56.23%, respectively. Moreover, the study shows that the gains in cropland coverage in Gedaref state over the studied period was at the expense of grassland and forest acreage, whereas the gains in settlements partially targeted cropland. Future LULC predictions showed a slight increase in cropland area from 89.59% to 90.43% and a considerable decrease in forest area (0.47% to 0.41%) between 2018 and 2048. These findings provide reliable information on LULC patterns in Gedaref region that could be used for designing land use and environmental conservation frameworks for monitoring crop produce and grassland condition. In addition, the result could help in managing other natural resources and mitigating landscape

fragmentation and degradation. The next chapter will present results on the integration of satellite remote sensing data and small-scale farmers' perceptions to determine land use/ land cover changes and their driving factors in Gedaref State, Sudan.

CHAPTER SEVEN

INTEGRATING SATELLITE REMOTE SENSING DATA AND SMALL-SCALE FARMERS' PERCEPTIONS TO DETERMINE LAND USE/ LAND COVER CHANGES AND THEIR DRIVING FACTORS IN GEDAREF STATE, SUDAN

7.1. Introduction

This chapter presents results on assessment of the perception of local land users of LULC change trends and determined the approximate and undelaying drivers that caused LULC dynamics in Gedaref state, Sudan. The LULC trends obtained in chapter six were compared with the perceptions of the local land users and the cultural crop cultivation area in Gedaref state. In addition, the effect of the rapid population growth in Gedaref state and climate variables (temperature and rainfall) on the dynamics of the main three LULC categories (cropland, forest and settlements) was evaluated. Furthermore, this chapter examined whether expansion of cropland in Gedaref state has led to an increase in crop yield.

7.2 Results

7.2.1 Land use/ land (LULC) maps and trends

The LULC maps (Figure 6.1) in chapter six, section 6.2.1 were used for this study to compare the LULC trends generated from the maps with the perception of the local land users. Figure 7.1 shows the proportionate coverage area (Figure 7.1A) and change rate (Figure 7.1B) of each LULC class

in Gedaref state for the same period. In the initial year of the study, i.e., 1988, cropland was the most predominant LULC class, occupying 78.64% of the total area of Gedaref state, followed by grassland (19.64%), while forest, water and settlement covered less than 2% of the study area (Figures 7.1). In the other studied years (1998, 2008 and 2018), the LULC class followed the same trend of coverage. During the entire study period (1988-2018), settlement areas increased nearly fourfold (i.e., 319.61%), whereas cropland increased from 78.64% to 89.59%. Conversely, forest, and grassland, drastically decreased in the same period (Figures 7.1). However, water classes increased throughout the study period, except for 2008. The highest net loss between 1988 and 2018 was detected in the forest area, followed by grassland (Figure 7.2). Although the losses were relatively higher in forest and grassland, the rate of change did not occur equally (Figure 7.2).



Figure 7.1: Land use/ land cover (LULC) change trend (A) and rate (B) of Gedaref state, Sudan



Figure 7.2: Net change in land use/ land cover (LULC) classes of Gedaref state, Sudan between 1988 and 2018.

7.2.2 Socio-economic and demographic characteristics of sampled households

Table 7.1 represents the socio-economic and demographic characteristics of the respondent. The majority (79.8%) of the respondents were male, while 20.3% were female (Table 7.1). Thus, about 58.3% of households were male-headed. The average age of the sampled respondents was 43.3 years old, ranging between 20 and 82 years. Regarding the education status, 27% of the respondents were illiterate, 34.8% and 20.8% had primary and secondary level school qualifications, respectively. Forty-three percent (43%) of the respondents lived in Gedaref state for more than 41 years, while 7.3% lived for less than 15 years (Table 7.1). The size of the household ranged from 1 to 17 individuals, with an average of 6 individuals per household. The proportion of respondents who owned land was 74.5%, while land size varied between 0.42 and 21 ha, with an average of 4.85 ha. Approximately 61% of the respondents were involved in
agricultural activities with 68.0% of the farmers having more than 41 years of experience in farming. Furthermore, 2.3% of the farmers were involved in selling forest products and 36.8% were engaged in on-farm activities, such as construction and government employees and businesses.

Table 7.1:	Summary o	of household	characteristics	for	small	holders	farming	communities	in
Gedaref sta	te, Sudan (N	= 400)							

Household characteristics	Category	Frequency	Percent (%)
Sex	Male	319	79.8
	Female	81	20.3
Head of the household	Male	233	58.3
	Female	167	41.8
	< 15	29	7.3
Number of years living in the village	16 - 30	109	27.3
	31 - 40	90	22.5
	> 41	172	43.0
	< 21	13	3.3
Δσε	22-40	180	45.0
nge	Category Frequency Iteration Male 319 79 Female 81 20 Male 233 59 Female 167 41 Female 167 41 requency 16-30 109 22 16-30 109 22 7 16-30 109 22 7 231-40 90 22 7 22-40 180 42 41-64 163 40 >65 44 11 Farming 244 6 Salaried employment 74 18 Self-employment 73 18 Selling forest products 9 2 <10	40.8	
		11.0	
	Farming	244	61.0
Main sources of income	Salaried employment	74	18.5
	Self-employment	73	18.3
	Selling forest products	9	2.3
Farming experience	< 10	35	8.8
	11 - 20	59	14.8

Table 7.1 (cont.)

Household characteristics	Category	Frequency	Percent (%)
Farming experience	21 - 40	34	8.5
	>41	272	68.0
	Owned	298	74.5
Land tenure	Rent in	82	20.5
	Share in	12	3.0
	Share out	8	2.0
	0.40 - 2.99	169	42.3
	3 - 5.99	93	23.3
	6 - 8.99	91	22.8
	> 9	47	11.8
	< 5	55	13.8
Family size (individuals)	6 - 10	133	33.3
•	11 - 14	138	34.5
	>15	74	18.5
	Illiterate	108	27.0
Education level	Primary	139	34.8
	Secondary	83	20.8
	University graduate	70	17.5

7.2.3 Perception of small-holder farmers of the drivers of land use / land cover (LULC) change in Gedaref state, Sudan

The local land users (respondents) identified 12 drivers of LULC changes in Gedaref state (7 proximate drivers and 5 underlying drivers) during the study period (Tables 7.2 and 7.3). The local

land users ranked firewood collection, agricultural expansion, and charcoal production as the top proximate drivers that influenced LULC changes in the studied landscape, with firewood collection being the main driver, followed by agricultural expansion (Table 7.2). Similar findings were revealed by key informant interviews and FGDs, which ranked firewood collection, agricultural expansion, settlements, and charcoal production as the primary causes of LULC dynamics in Gedaref state.

LULC proximate		Number of respondents per rank							Ranking	
drives	1	2	3	4	5	6	7	Weight	index	Kalik
Firewood collection	221	101	62	13	2	1	0	2523	0.225	1
Agriculture expansion	133	83	140	29	14	1	0	2289	0.204	2
Charcoal production	38	155	95	94	14	4	0	2097	0.187	3
Settlements	9	44	49	157	30	83	28	1484	0.132	4
Construction	0	12	35	72	253	28	0	1350	0.120	5
Timber	2	3	20	26	75	256	18	991	0.088	6
Bush fires	0	1	5	8	8	24	354	489	0.044	7

Table 7.2: Perceived proximate drivers of land use/ land cover (LULC) changes by local land users in Gedaref state, Sudan

Poverty and rapid population growth were ranked as the most vital underlying factors that drive LULC change in Gedaref state, with a ranking index of 0.280 and 0.216, respectively (Table 7.3). This was followed by a lack of law enforcement and financial resources, ranked third and fourth, respectively (Table 7.3). Similarly, the rapid population growth, poverty, rainfall fluctuation, lack

of law enforcement and high cost of agricultural inputs were reported by key informant interviews and FGDs to be the key underlying drivers of LULC changes in Gedaref state.

LULC underlying drives	Nun	iber of r	esponde	nts per 1	Weight	Rankong	Rank	
-	1	2	3	4	5		index	
Poverty	237	60	55	44	4	1682	0.280	1
Population growth	42	156	76	111	15	1299	0.216	2
Lack of law enforcement	73	85	106	131	5	1290	0.215	3
Lack of financial resources	49	89	150	99	13	1262	0.210	4
Demand for timber	0	10	14	16	360	474	0.079	5

 Table 7.3: Perceived underlying drivers of land use/ land cover (LULC) changes in Gedaref state, Sudan

7.2.4 Multinomial logistic regression results of perceived drivers of land use/ land cover (LULC) changes

The sex of the household head had a significant ($p \le 0.05$) effect on the perception of the respondents on the collection of firewood, charcoal production, timber, agricultural expansion, poverty, population growth, lack of financial resources and law enforcement as LULC drivers in Gedaref region (Table 7.4). Poverty was significantly ($p \le 0.05$) affected by the sex of respondents, sex of the household head, age, and education level. Whereas family size had no significant influence on the perception of the small-holder farmers of LULC change.

Perceived	Independent variable	Estimate	Standard error	Wald	n-Vəluo	95% Confidence Interval	
driver of LULC change	independent variable	Listimate	error	vv alu	p ⁻ v and	Lower bound	Upper bound
Firewood	Head of the household	2.148	0.927	5.372	0.020	0.33146	3.963761
Thewood	Land size	-0.243	0.124	4 3.848 0.050		-0.48451	0
Charcoal production	Head of the household	-1.120	0.385	8.461	0.004	-1.87732	-0.36528
	Age	0.031	0.011	7.659	0.006	0.00896	0.052592
	Land tenure	-0.566	0.237	5.698	0.017	-1.03002	-0.10093
	Farming experience	-0.040	0.012	11.183	0.001	-0.06401	-0.01715
Timber	Head of the household	-1.722	0.740	5.408	0.020	-3.17009	-0.2705
Construction	Age	-0.044	0.015	9.251	0.002	-0.07257	-0.01613
Construction	Farming experience	0.031	0.015	4.213	0.040	0.001	0.060154
Agriculture	Head of the household	1.171	0.524	4.993	0.025	0.1441	2.198002
expansion	Age	0.037	0.015	6.091	0.014	0.007968	0.065788
	Sex	1.856	0.442	17.593	0.000	0.988797	2.723202
Poverty	Head of the household	-2.261	0.603	14.047	0.000	-3.44202	-1.07881
	Age	0.043	0.017	6.586	0.010	0.00995	0.076961

Table 7.4: Significant ($p \le 0.05$) socio-economic factors influencing respondents' perceptions on the drivers of land use/ land (LULC) cover changes in Gedaref state, Sudan

Table 7.4 (cont.):

Perceived			Standard			95% Confide	95% Confidence Interval		
driver of LULC change	Independent variable	Estimate	error	Wald	<i>p</i> -Value	Lower bound	Upper bound		
Poverty	Education	0.796	0.197	16.262	0.000	0.408793	1.182953		
Population growth	Sex	0.872	0.290	9.055	0.003	0.303801	1.439361		
	Head of the household	-1.206	0.364	10.955	0.001	-1.91732	-0.49102		
Lack of financial	Sex	-0.967	0.342	8.010	0.005	-1.63476	-0.29706		
resources	Head of the household	0.812	0.389	4.352	0.037	0.04879	1.574018		
	Sex	-1.098	0.339	10.522	0.001	-1.76026	-0.43541		
Lack of law enforcement	Head of the household	1.712	0.376	20.780	0.000	0.976068	2.44807		
	Education	-0.233	0.118	3.895	0.048	-0.46522	-0.002		
Demand for timber	Education	-0.678	0.245	7.639	0.006	-1.15836	-0.19723		

7.2.5 Comparison between small-holder farmers' perception and land use / land cover (LULC) classification

The results of farmers' perceptions of LULC trends are presented in Figure 4. About 55.5% of the respondents indicated that the cropland area had increased in the study area, while 39.5% claimed that there was no change in the cropland coverage (Figure 7.3A). Additionally, the majority of the respondents (90.8%) also claimed that forest area had decreased over the study period (1988–2018) in Gedaref state (Figure 7.3B). Whereas 96.5 % of the interviewed farmers perceived an increment in settlement area during the study period (Figure 7.3C). Although the respondents have indicated an increase in cropland area, the majority of 90.8% claimed that crop production has decreased in Gedaref state in the period between 1988 and 2018 (Figure 7.3D). The results of Mann-Kendall trend analysis of the satellite imagery showed a significant ($p \le 0.05$) increase in cropland area in Gedaref state by about 249,421 ha each 10 years between the years 1984 and 2018, which confirmed the community's perception (Figure 7.4A; Table 7.5). Similarly, farmers' perception agreed with the trend of classified forest area, which showed a significant ($p \le 0.05$) decrease by -9,175 ha each 10 years, with a confidence interval ranging between -28082.75 and -2252.72 ha (Figure 7.4B; Table 7.5). The settlement area had significantly (p = 0.05) increased by 6,414 ha each 10 years, with a confidence interval between 3,179.44 and 12,136.97 ha for the study period (Figure 7.4C; Table 7.5), which also agreed with the farmers' perception.



Figure 7.3: Farmers' perception towards changes in (A) cropland area; (B) forest area; (C) settlement area; and (D) crop production in Gedaref state, Sudan



Figure 7.4: Satellite-based land use/ land cover (LULC) change trends in Gedaref state, Sudan between 1988 and 2018. (A) cropland; (B) forest; and (C) settlement.

Classer	Ra	nge	Son's slong	95% Confidence	n Valua	
Class area	Minimum	linimum Maximum		Interval	p-value	
Cropland	5023958.04	5723662.45	249421	171830.53 - 275872.05	0.006	
Forest	30056.44	69053.13	-9175	-28082.752252.72	0.041	
Settlement	6739.02	28278.13	6414	3179.44 - 12136.97	0.051	

Table 7.5: Estimated Sen's slope values for cropland, forest, and settlement (ha) trends in Gedaref State, Sudan between 1988 and 2018.

7.2.6 The actual trend of cultivated area, crop yield and production

The data obtained from the ministry of agriculture showed that the cultivated areas for the five major crops significantly ($p \le 0.05$) increased between 1988 and 2018 (Figure 7.5; Table 7.6). This further confirmed the increase in cropland area of the satellite-based LULC maps as well as farmers' perceptions. The trend analysis showed that the actual cultivated area for sorghum, millet and sunflower significantly ($p \le 0.05$) increased by 3732, 4200, and 840 ha per year, respectively, between 1988 and 2018 (Figure 7.5; Table 7.6). As an increase in the total cultivated area, the crop production increased by 11249.17, 3539.39, 1543.19, 517.71, and 159.67 tons for sorghum, sesame, millet, sunflower, and cotton, respectively (Figure 7.6; Table 7.7). The farmers' perceptions of crop production over the study period disagreed with the actual trend of crop production, which increased over time (Figure 7.6). The majority of the farmers (81%) indicated that crop production in the study area has decreased, while 14% and 4.8% of the farmers indicated increase and no change in crop production, respectively. Although the actual trend showed an increase in crop production over time, the crop yield per unit (kg ha⁻¹) remained constant, with a weak correlation detected between cultivated area and crop yield (Figure 7.7).



Figure 7.5: Annual actual cultivated area trends for sorghum, sesame, millet, sunflower and cotton in Gedaref state, Sudan between 1988 and 2018.

Table 7.6: Estimated Sen's slope values for culitvated area of sorghum, sesame, millet, sunflowerand cotton (ha) trends in Gedaref State, Sudan between 1988 and 2018

Cultivated area	Range		Sen's slope	95% Confidence	<i>p</i> -Value	
(ha)	Minimum	Maximum		interval		
Sorghum	121800	303576	3732	-12222 - 17200.8	0.000	
Sesame	14280	54390	411.2	-4767.0000 - 5871	0.146	
Millet	4200	165900	4200	-3108.0000 - 13480	0.000	
Sunflower	420	64680	840	-4956 - 6935	0.000	
Cotton	840	63420	142.8	-1219.68 - 4676.86	0.018	



Figure 7.6: Annual crop production trends for sorghum, sesame, millet, sunflower and cotton in Gedaref State, Sudan from 1988–2018

Table 7.7: Estimated slope of linear regression for the annual crop production trends for five crops(sesame, sorghum, cotton, millet, and sunflower) grown in Gedaref state, Sudan from 1988–2018.

	prod	uction			
Сгор	rang	e (ton)	Slope	95% Confidence Interval	<i>p</i> -Value
	Minimum	Maximum			
Sorghum	183000.00	1946000.00	11249.17	-7715.37 - 30213.69	0.234
Sesame	27000.00	289500.00	3539.39	1455.61 - 5623.18	0.001
Millet	1000.00	101000.00	1543.19	930.19 - 2156.19	0.001
Sunflower	66.50	37000.00	517.71	107.24 - 928.18	0.152
Cotton	200.00	19000.00	159.67	0.06 - 0.76	0. 224



Figure 7.7: The relationship between cultivated area (ha) and yield of sorghum, sesame, millet, sunflower and cotton in Gedaref state, Sudan using Pearson's correlation test. The dashed lines above and below the trend lines are the confidence intervals.

7.2.7 Relationship between climatic variables, population and land use/ land cover (LULC) classes

The result of the relationship between each LULC class area (cropland, forest and settlement) and climatic variables (rainfall and temperature) on the one hand, and population on other hand are represented in Figure 7.8. The climatic variables and population were positively correlated with the change in cropland area (Figure 7.8 A-C), with correlation coefficient (r) ranging between 0.80 and 0.98. However, these variables negatively influenced the change in forest area with a correlation coefficient (r) ranging between 0.47 and 0.88 (Figure 7.8 D-F). The correlation between population growth in Gedaref state and the settlement area was significant (p < 0.05), and positive with (r) = 0.97 (Figure 7.8G).

Table 7.8 shows the results of a multiple linear regression that estimated changes in three LULC areas (cropland, forest, and settlement) as a function of rainfall, temperature, and population growth. The combination of the three independent variables in the model revealed that 98%, 83% and 97% variabilities of the change in cropland, forest, and settlement areas, respectively explained by these factors (Table 5). Nevertheless, only temperature and population significantly ($p \le 0.05$) influenced the increase of cropland area. In addition, the coefficient of population growth was significantly ($p \le 0.05$) influencing the expansion of the settlement area (Table 5). Although the relationship between the three variables and the change in forest area was insignificant ($p \ge 0.05$), temperature and population growth natively influenced the change in the forest area.



Figure 7.8: The relationship between: (**A**) cropland area and rainfall; (**B**) cropland area and temperature; (**C**) cropland area and population; (**D**) forest area and rainfall; (**E**) forest area and temperature; (**F**) forest area and population; (**G**) settlement area and population using Pearson's correlation test. The shaded lines above and below the trend lines are the confidence intervals

Table 7.8: The multiple linear regression terms and coefficients of determination (R^2) estimating the change land use/ land cover (LULC) areas as a function of rainfall, temperature, and population in Gedaref state, Sudan

LULC class		Intercept	Tmean (°C)	Rainfall (mm)	Population	R ²	
Cropland	Coefficient	7.000e+05	7.000e+05 1.089e+05 -1.998e+02 6.047		6.047e-01	0.98	
	<i>p</i> -value	0.505	0.027	0.207	0.001	0.90	
Forest	Coefficient	3.199e+05	-6.758e+03	1.631e+01	-2.814e-02	0.83	
	<i>p</i> -value	0.216	0.338	0.594	0.117	0.85	

Settlement	Coefficient	1.652e+04	-9.541e+02	2.614e+00	2.455e-02	0.97
	<i>p</i> -value	0.742	0.525	0.700	0.003	

7.3 Discussion

Between 1988 and 2018, Gedaref state in Sudan experienced significant and increased rates of LULC changes. Cropland is the major LULC class in both 1988 and 2018, accounting for 78.64% and 89.59% of the total Gedaref landscape. This is because Gedaref state is the Sudan's main rainfed agricultural area, with approximately 80% of the population in the state relying on agricultural production as the primary source of income (Osman et al., 2021). Furthermore, Gedaref area is characterized by multi-cultivation of various food and cash crops and this is due to the fact that Gedaref area has rich clay soil and receives, on average, about 600 mm of rainfall (Sulieman & Elagib, 2012). The expansion of cropland area, which was calculated from the satellite-based data and a machine learning classifier, agreed with the actual cultivated area that was obtained from the ministry of agriculture of Gedaref state. Similarly, farmers' perception showed an increase in cropland area, which was in line with the satellite-based cropland area. The increase in crop production trend of the five major crops in Gedaref state can be explained by the increasing trend in the total cultivated area. The farmers' perceptions of crop production over the study period disagreed with the actual trend of crop production. This is mainly attributed to the fact that the present study focused on small-scale farmers who experience limited access to farming technology and production inputs. This can result in low crop yield and production coupled with a small farm size. Also, the crop production trend of the data acquired from the ministry of agriculture was for the whole Gedaref state, which could be affected by large-scale mechanized farming in the region. Although cultivated area has increased during the study period, the crop

yield per unit for the five major crops remained the same with a slight fluctuation. This fluctuation in crop yield might be attributed to the amount of rainfall that varies for one year to other in Gedaref state (Osman *et al.*, 2021). Therefore, the expansion of cropland is probably due to the high demand for food to meet the need of the rapid population growth in Gedaref and Sudan at large.

In the present study, the forest area has dramatically decreased, which is in agreement with the farmers' perception. The increase of cropland area at the cost of forest and grassland areas between 1988 and 2018 could be attributed to the expansion of mechanized farming in Gedaref state (Issa, 2018). Additionally, the interviewed respondent (local land users) correctly perceived an increment in the settlement in recent years, confirming the findings of satellite-based LULC changes in the period of 1988–2018. The results of this study also concur with the findings of Abdelmalik et al. (2021), who recorded that forest area declined by 38.3%, while cropland and settlement areas increased by 61.1% and 225%, between 1990 and 2017, respectively in Gedaref state. The rapid population growth rate, the flow of refugees from the neighboring countries, high rates of natural increase and internal migration from other states to the study area accelerated the rate of LULC dynamics, especially the expansion of the settlement area (Mohammed, 2016).

The findings of the household surveys, FGDs as well as the key informant interviews, revealed that local land users perceived firewood collection, agriculture expansion, charcoal production, and settlements as the main proximate factors that drive LULC dynamic in Gedaref state. These factors were driven by the rapid population growth rate, high poverty level, rainfall fluctuation, high cost of inputs such as fertilizer and pesticides and lack of law enforcement by the government. Many studies emphasized the significance of studying socio-economic group variances in respondents' perceptions of LULC changes (López-Santiago *et al.*, 2014; Zoderer *et al.*, 2016). Accordingly, the results of this study showed a connection between the perception of LULC

change drivers and socio-economic factors. The results of the multinomial logistic regression model showed that the perception of an individual of LULC change drivers differs from one person to another depending on certain socio-economic factors: their sex, sex of the head of the household, age, educational level, land tenure, land size, family size and farming experience. The results showed that sex of the respondents, sex of the head of the household, age, educational level, and farming experience are the most significant factors affecting the farmers' perception of LULC change drivers. The age of the respondents has significantly influenced their perception on charcoal production, construction, agricultural expansion, and poverty as the major drivers of LULC change. This confirmed the results of Ariti et al. (2015), who reported that age of the respondents significantly influenced farmers' perception of LULC changes drivers in Ethiopia. Also, farming experience has significantly influenced farmers' perception of charcoal production and construction as LULC change drivers. Moreover, educational level has significantly influenced farmers' perception of poverty, lack of law enforcement and the demand for timber. According to Kouassi et al. (2021) and Ketema et al. (2021), experienced and educated land users are expected to have more information and knowledge about the drivers of LULC changes.

Overall, the perceived LULC change trends observed by the local land users and quantitatively by remote sensing approach were relatively similar. This revealed the advantage of earth observation tools in detecting and estimating the changes in LULC at a landscape scale and beyond. However, earth observation methods can encounter some sources of errors and cannot provide vital information on why the changes in LULC took place (Ewunetu *et al.*, 2021; Kleemann *et al.*, 2017). In contrast, local land users explained in detail the reasons for LULC changes and the underlining factors that drive such changes. For instance, local land users elaborated on why the area of forest in Gedaref state dropped while the amount of urbanization increased. Local land

users elaborated on why forest coverage decreased and why settlement area increased in Gedaref state. This implies that the quantitative limitation of the earth observations can be improved by some information generated from the knowledge of local land users. Nevertheless, earth observations have advantages over the respondent's perception due to its capability to numerically quantify the rate, extent, and spatial-temporal dynamics of LULC patterns. However, the local viewpoint approach allowed for a more thorough explanation of the reasons these processes occurred during the study period. Consequently, the knowledge of local communities on LULC dynamics enables the identification of the important LULC drivers in a particular area. The LULC drivers cannot be easily determined through earth observation tools and this limitation could be complemented by the knowledge of local land users, a better understanding of LULC dynamics and consequently, better resource management.

The expansion in cropland area is positively correlated with climatic variables (temperature and rainfall) and population. Being a rainfed agricultural area, the fluctuation in the amount of rainfall in the past thirty years in Gedaref (Osman *et al.*, 2021) might have attracted more investors and small-scale farmers, leading to an expansion in cropland. On the other hand, the high demand for food to meet the high population growth rate in the state could have led to an increase in cropland, as earlier mentioned. This could also explain the strong positive correlation between population growth and the change in cropland area. A study by van Vliet et al. (2013) reported that the expansion in cropland prompted in the Sahel region is attributed to the population growth and the amount of rainfall, which is in accordance with the findings of this study. The results of this study also showed that climatic variables (temperature and rainfall) and population were negatively correlated with changes in the forest area. It has been demonstrated that a rapid increase in rural population growth resulted in a decreased forest area (Misra *et al.*, 2014). In the case of Gedaref

state, the negative relationship between the population growth and the change in forest area might be due to the need for land and timber for house construction to meet the increased population. This could also explain the strong positive correlation between the population and the change in the settlement area. Indeed, the clearance of forests in rural areas is highly affected by agricultural expansion, woodcutting for timber and charcoal production (Misra *et al.*, 2014).

Although few significant relationships between climate, population, and LULC class coverage were detected in this study using the linear regression models, the coefficients of the models can be utilized to quantify the impact of these variables on LULC dynamics (Osman *et al.*, 2021; Poudel & Shaw, 2016). For instance, a unit increase in the population could lead to an increase in cropland and settlements areas by 0.6 and 0.02 ha, respectively, and a reduction in forest area by - 0.03 ha. In addition, the positive or negative sign of the linear regression coefficients indicates a positive and negative change in the LULC class versus the change in climate variables and population growth (Nicholls, 1997). Furthermore, this study showed that the linear regression model captured 98%, 83% and 97% of the variability in the change of cropland, forest and settlements areas, respectively as a function of temperature, rainfall and population growth. This indicates that the change in cropland, forest and settlement areas were well explained by these three factors and the rest of the variations could be explained by the other confounding variables that were not included in the model.

7.4 Conclusions

In conclusion, this study examined LULC changes in Gedaref state, Sudan for the period between 1988 and 2018 using satellite-based data and a machine learning classification algorithm. The satellite-based LULC maps were compared with the farmers' perception and with actual ground

truth data obtained from the Gedaref ministry of agriculture. The results showed an increase in cropland and settlement areas and a substantial decrease in forested areas during the study period (1988–2018). Also, the expansion in cropland was in line with the actual cultivated area during the study period. Firewood collection, agriculture expansion, and charcoal production were ranked by the local communities as the most important proximate drivers for LULC changes in Gedaref state. On the other hand, farmers identified poverty, population growth and the lack of law enforcement as the most important underlying drivers of LULC changes in the study area. Therefore, sustainable resource management in Gedaref state that allows for successful coexistence between local users and conditions for environmental, social, and economic well-being remains a significant challenge. Even though the current study was conducted at a landscape scale to assess how rapid human modifications and climate become a critical driving force for LULC changes, it implies the implication of these drivers on the global LULC change trends. Moreover, the findings of this study could be used in formulating environmental planning strategies, natural resources management, guidelines for the maintenance of ecosystem services, and conservation and utilization of natural resources in Gedaref region or other regions with similar settings.

7.5 Summary

Understanding land use/ land cover (LULC) dynamics and the factors that drive these changes is critical for future prediction of landscape structure and development of sustainable and robust land-management strategies and policies. However, little is known about the proximate and underlying factors driving LULC dynamics and perceptions of land users of these dynamics. This study aimed to i) assess the local farmers' perception of LULC changes in Gedaref state; and (ii) determine the drivers of LULC changes in Gedaref state. Initially, satellite-based LULC maps for the period 1988 – 2018 that were generated using a robust approach were utilized in this study. Subsequently,

Mann–Kendall trend analysis was used for the satellite-based LULC changes (area in ha) and the actual crop cultivated area (ha), which were compared with local farmers' perceptions. A multinomial logistic regression was also employed to determine the major drivers that influence LULC changes in Gedaref state. This was based on a semi-structured questionnaire with 400 respondents. Furthermore, Pearson correlation test and multiple linear regression were employed to assess the effect of change drivers on the area (ha) of the main landscape classes in Gedaref state (i.e., cropland, forest and settlement). Cropland and settlement areas increased (from 78.64% to 89.59%) and (from 0.11% to 0.44%), respectively, during the study period. In contrast, forest area decreased (from 1.08% to 0.47%). The results of the semi-structured questionnaire revealed that the satellite-based LULC change trends (area) agreed with respondents' perceptions. Specifically, 55.5% of the respondents observed an expansion in the cropland area and 96.5% reported an increment in the settlement. Whereas 96.5% of the respondents observed a decline in the forest area. Furthermore, the study shows that the main drivers for LULC changes in Gedaref state were firewood collection, agricultural expansion, charcoal production, settlements, poverty and population growth. The sex of the household head, age, sex of the respondent, education level and farming experience significantly ($p \le 0.05$) influenced the respondents' perceptions of LULC changes in Gedaref state. Moreover, temperature, amount of rainfall and population growth were positively correlated with the change in cropland and settlement areas (a correlation coefficient (r)ranged between 0.80 and 0.98). While the same variables negatively influenced change in forest areas, with r ranging between 0.47 and 0.88. The findings of this study could be useful for planners and decision-makers for developing coherent land use policies and management strategies. The next chapter will present an overall conclusions and recommendations on mitigating the impacts

of climate change, and land use-land cover dynamics on small-scale farmers in Gedaref State of Sudan.

CHAPTER EIGHT

SYNTHESIS, GENERAL CONCLUSIONS AND RECOMMENDATIONS

8.1 Introduction

This chapter provides general conclusions and recommendations and highlights potential future research areas on climate change and land use/ land cover (LULC) in Gedaref State and other similar settings.

8.2 Synthesis

The preceding chapters of this thesis investigated the impacts of climate trends and LULC changes on crop production of small-holder farmers in Gedaref state, Sudan. Despite the importance of climate change in the agricultural sector, little is known about climate variability and change in Sudan and there is no study that had assessed the relationship between climate trends and crop yield in Gedaref state, which is the Sudan's hub of rainfed agriculture area. This might be due to the fact that assessing climate impacts on crop yield is a challenging endeavor, because crop yield is associated with many factors, such as soil fertility and agricultural inputs, rather than climatic variables (Nicholls, 1997). In addition, mapping LULC trends, and their intensities and the driving factors triggered changes in landscape structure for the entire Gedaref state. This has not been investigated before. Furthermore, small-scale farmers' perception of climate trends and LULC changes has not been assessed. Therefore, further work is required to comprehend LULC dynamics and climatic changes in Gedaref state. This thesis has contributed to knowledge generation that helps to bridge these information gaps. Chapter four looked at climate and crop yield trends in Gadaref state between 1984-2018 using Man-Kendall trend analysis (Kendall, 1948), daily climatic observations (rainfall and temperature) and annual crop yield data for five major crops that are commonly cultivated in Gedaref state under rainfed conditions; namely, sorghum, sesame, millet, cotton and sunflower. An in-depth analysis of temperature and rainfall variability indices between the years, rainfall patterns such as the amount of rainfall, onset and cessation dates, length of the rainy season, and the number of rainy days in each year were carried out. In addition, the relationship between the length of the rainy season and crop yields was established for a better understanding of the effect of rainfall patterns on crop yields. Furthermore, the relationship between climatic variables and crop yields was determined. Although assessing the impact of climate change on crop yield is difficult as mentioned earlier, the present study overcame the challenge of separating non-climatic variables using the so-called 'first difference approach'. This method was initially introduced by Nicholls (1997) and thereafter adopted by studies that evaluated the effect of climate change on crop yield. (El-Maayar & Lange, 2013; Peltonen-Sainio *et al.*, 2010; Zhang *et al.*, 2014).

Since chapter four looked at climate and crop yield relationships, it is important to understand the perceptions of the affected community, especially small-scale farmers who are more vulnerable to climate change. Thus, chapter five analysed the perceptions of small-holder farmers of climate variability and change and identified the adaptation measures that they use to cope with its negative impacts on their farming activities. This was done using a semi-structured questionnaire, key informant interview and FGD. The farmers' perception of climate variability and change was compared with the actual trend of meteorological records. This enabled the understanding of whether the small-holder farmers are aware of climate change in Gedaref state. Understanding climate trends, climate and crop relationship, and farmers' perception are very important in

awareness creation amongst different stakeholders and policymakers. In addition, it highlights the need for resource allocation to support the uptake of adaptation practices that ensure resilience amongst agricultural communities within the state. This can be useful in implementing and monitoring adaptation measures through the integration of farmers' perceptions with scientific knowledge to improve crop yield in Gedaref state and other rainfed agricultural areas in Sudan.

Land use land cove change (LULC) and climate change are related to each other (Dale, 1997). For example, in Sudan, the reduction in rainfall has turned millions of hectares of marginal semi-desert grazing land into a desert (UNEP, 2007). On the other hand, large-scale deforestation might lead to warmer climatic conditions and change the rainfall pattern. Therefore, the emerging challenge of climate and LULC changes should not be considered independently but to evaluate the combined effects of these changes. Hence, chapter six mapped the spatio-temporal changes in LULC over time in Gedaref state for the years 1988 to 2018 using remote sensing techniques, Google Earth Engine and the machine learning random forest classifier. For the first time, this study determined the underlying dynamics that cause the changes in the landscape structure using intensity analysis. In addition, the study predicted the future LULC trends for the years 2028 and 2048 using cellular automata-artificial neural network (CA-ANN). This helped in providing information on LULC patterns in Gedaref region that could be useful for designing management plans and developing policies for assessing and monitoring crop and grassland production, other natural resources produce, landscape fragmentation and degradation, and ecosystem functions.

Overall, the earth observation tools (remote sensing) have an advantage of detecting and estimating the changes in LULC at a landscape scale and beyond. However, earth observation methods can encounter some sources of errors and cannot provide vital information on why the changes in LULC took place (Ewunetu *et al.*, 2021; Kleemann *et al.*, 2017). Therefore, it is important to

integrate the results generated from remotely sensed data with the perception of local land users for better understanding the approximate and underlying drivers that cause the change in landscape structure. In this regard, chapter seven assessed the perception of local land users on LULC change trends and determined the approximate and underlining drivers that caused LULC dynamics in Gedaref state. The local land users explained in detail the reasons for LULC changes and the underlining factors that drive such changes. For example, the local farmers provided a detailed explanation on why forest area in Gedaref state declined while the cropland and settlements areas increased in the last decades. Such information cannot be generated from remote sensing imagery and that is the advantage of engaging the local land users in such studies. In fact, the perceived LULC change trends observed by the local land users and that quantified by the remote sensing approach were relatively similar, which indicated the reliability of the classified LULC maps in chapter six. In addition, the expansion in cropland area generated from LULC maps was also similar to the trends of the cultivated areas that were obtained from the ministry of agriculture in Gedaref state. Furthermore, the relationship between the expansion in cropland area and crop yields was also determined in chapter seven.

8.3 Conclusions

The average annual temperature in Gedaref state has increased by 0.04°C per year for the period between 1980 and 2018, with fluctuation in the amount of rainfall and frequent occurrence of droughts, confirming the earlier report of climate variability and change in Sudan. Between 1984 and 2000, the anomalies of annual temperature were below the long-term average, indicating a cold period in the study area. However, after the year 2000, temperature anomalies were above the long-term average, indicating a warm period. The increase in maximum and minimum temperatures by 0.03 and 0.05 per year, respectively, has negatively affected crop yields of five

major crops that are cultivated in Gedaref state under rainfed conditions, namely; sorghum, sesame, millet, cotton and sunflower. On the other hand, the trend and standardized anomalies of rainfall in Gedaref states showed that the amount of rainfall fluctuated over time with frequent occurrence of droughts in the years 1984, 1987, 1990, 1991, 2011, and 2013. The amount of rain and the length of the rainy season were the most important climate factors that positively influenced the yield of sesame, sorghum, sunflower, and cotton in Gedarfe state.

Climate change has taken place in Gedaref state, and it has a negative impact on crop yields. Fortunately, small-scale farmers in this state are aware of climate variability and change and they have developed some adaptation measures to cope with this negative climate change impact. These adaptation measures include crop rotation, early cultivation, and cultivation of short-maturing crop varieties, mixing farming, use of appropriate agricultural inputs such as fertilizers and introduction of drought tolerant varieties. However, the uptake of these adaptation measures depends on the socio-economic characteristics of the farmer. For example, the uptake of mixed farming considerably depends on sex of the head of the household, land tenure, and family size. Therefore, government policies that promote climate adaptation strategies in Gedaref state should consider the socio-economic indicators of the farmers to enhance their likelihood and adoption rate of such climate-smart technologies.

There was a drastic change in LULC in the period between 1988 and 2018. Cropland was the most dominant LULC class in Gedaref state. Forest and grassland areas were considerably decreased by 56.47% and 56.23%, respectively, between 1988 and 2018. However, cropland and settlement areas dramatically increased by 13.92% and 319.61%, respectively, during the study period. This clearly demonstrates that the expansion of cropland and settlements areas was at the expense of forest and grassland areas. The future prediction showed a slight increase in cropland area from

89.59% to 90.43% and a considerable decrease in forest area (0.47% to 0.41%) between 2018 and 2048. Therefore, the study concludes that the total cultivated area and settlements are increasing in Gedaref state at the expense of forest and grassland due to cumulative anthropogenic activities.

The increase in cropland and settlements areas and decrease in the forest area in Gedaref state were validated using the perceptions of the local land users. The small-holder farmers identified 12 approximate and underlining drivers of LULC changes in Gedaref state. The approximate drivers that caused LULC dynamics were firewood collection, agricultural expansion, charcoal production, settlements, construction, timber, bush fires. While the underlining drivers were poverty, population growth, lack of law enforcement, lack of financial resources and demand for timber.

8.4 Recommendations

Based on the aforementioned concluding remarks, this study recommends the following for new insights, policy strategies and future research:

- There is a need for awareness creation amongst different stakeholders and policymakers in Gedaref state on the impacts of climate variability and change on crop production and the need for resource allocation to support the uptake of adaptation practices that ensure resilience amongst agricultural communities within the state,
- Government should emphasize and put more effort into enhancing the adaptive capacity of small-scale farmers to climate change in Gedaref state through extension services, agricultural inputs, education of the farmers and diversification of the source of income to improve food and nutrition security,

- 3. Government policies that promote climate adaptation strategies in Gedaref state should consider the socio-economic factors that influence the uptake of adaptation measures to improve the likelihood of small-holder farmers. This enables an enhanced adoption of climate-smart strategies to cope with negative impact of climate change,
- 4. Farmers should be well trained in adopting climate-smart adaptation measures to improve crop yields and income under a changing climate,
- 5. The protection of the natural resources in Gedaref state, such as forest, requires strong policy enforcement framework to minimize the risk of deforestation and land degradation,
- 6. This study used only climatic variables (temperature and rainfall) to assess crop yields under climate change and it did not incorporate other factors that influence yields such as soil properties and other farming practices. Future studies should include all factors that influencing crop yield in a holistic modelling framework such as DSSAT, and
- 7. The study recommends that the models developed for estimating crop yields should be tested and assessed using an independent test dataset collected at different points in time.

References

- Abbas, Z., Yang, G., Zhong, Y., & Zhao, Y. (2021). Spatiotemporal change analysis and future scenario of lulc using the CA-ANN approach: A case study of the greater bay area, China. *Land*, *10*(6), 584. https://doi.org/10.3390/land10060584
- Abdalla, F. I. M. (2015). Refugees consumption of Forest Products prior to Repatriation Kuna Zeberma Camp, Kassab Locality, Gedaref State. *International Journal of Current Microbiology and Applied Sciences*, 4(2), 843–846.
- Abdalla, N. A. (2018). Participation of rural woman in forestry extension activities (Eastern Galabat Locality Gedaref State Sudan). MSc. thesis. Sudan University of Science & Technology.
- Abdelmalik, A. M., Musa, A. I., F-E Mohamed, I., Kurosaki, Y., Tsubo, M., Khatir, A. A., Eldin Ali Babiker, I. A., Hassan, R. A., Elhassan, N. G., Zakieldeen, S. A., Aljloon, F. I., Forabosco, F., Ijaimi, A., Osman, A. K., & El Hag, F. M. (2021). Climate change impacts on land use in Gadaref and North Kordofan States and future desert sheep distribution in Sudan. *International Journal of Agriculture and Agricultural Sciences*, 6(2), 176–184. www.advancedscholarsjournals.org
- Adam, M. I. A. (2019). Assessment of El-Rawashda reserved forest resources using selected climate change indicators - Gedaref State - Sudan. MSc Thesis. Sudan University of Science and Technology, Suda.
- Ahmed, I., ur Rahman, M. H., Ahmed, S., Hussain, J., Ullah, A., & Judge, J. (2018). Assessing the impact of climate variability on maize using simulation modeling under semi-arid environment of Punjab, Pakistan. *Environmental Science and Pollution Research*, 25(28), 28413–28430. https://doi.org/10.1007/s11356-018-2884-3
- Ahmed, N. M. E., & Elsaied, M. (2017). Status of agricultural statistics in Sudan. *Universal Journal of Plant Science*, 5(2), 29–35. https://doi.org/10.13189/ujps.2017.050203
- Akalu, F., Raude, J. M., Sintayehu, E. G., Jeremiah Kiptala, & ... (2019). Evaluation of land use and land cover change (1986–2019) using Remote Sensing and GIS in Dabus sub-catchment, South-Western Ethiopia. *Journal of Sustainable Research in Engineering*, 5(2), 91–100. http://jsre.jkuat.ac.ke/index.php?journal=sri&page=article&op=view&path%5B%5D=768
- Akinyemi, F. O., & Speranza, C. I. (2022). Agricultural landscape change impact on the quality of land: An African continent-wide assessment in gained and displaced agricultural lands. *International Journal of Applied Earth Observation and Geoinformation*, 106, 102644. https://doi.org/10.1016/j.jag.2021.102644
- Akudugu, M. A., Dittoh, S., & Mahama, E. S. (2012). The implications of climate change on food security and rural livelihoods: Experiences from Northern Ghana. *Ournal of*

Environment and Earth Science, 2, 21–29. https://www.ptonline.com/articles/how-to-get-better-mfi-results

- Aldwaik, S. Z., & Pontius, R. G. (2012). Intensity analysis to unify measurements of size and stationarity of land changes by interval, category, and transition. *Landscape and Urban Planning*, *106*(1), 103–114. https://doi.org/10.1016/j.landurbplan.2012.02.010
- Aldwaik, S. Z., & Pontius, R. G. (2013). Map errors that could account for deviations from a uniform intensity of land change. *International Journal of Geographical Information Science*, *27*(9), 1717–1739. https://doi.org/10.1080/13658816.2013.787618
- Ali, A. Y. A., Guisheng, Z., Hassan, A., Yagoub, S. O., Farah, G. A., Ahamed, N. E., Ibrahim, A. M., Ibrahim, M. E. H., Suliman, M., Elradi, S. B., Ibrahim, E. G., & Omer, S. M. (2020). Sesame seed yield and growth traits response to different row spacing in semi-arid regions. *Universal Journal of Agricultural Research*, 8(4), 88–96. https://doi.org/10.13189/ujar.2020.080402
- Almazroui, M., Saeed, F., Saeed, S., Nazrul Islam, M., Ismail, M., Klutse, N. A. B., & Siddiqui, M. H. (2020). Projected change in temperature and precipitation over Africa from CMIP6. *Earth Systems and Environment*, 4(3), 455–475. https://doi.org/10.1007/s41748-020-00161-x
- Alonso-Pérez, F., Ruiz-Luna, A., Turner, J., Berlanga-Robles, C. A., & Mitchelson-Jacob, G. (2003). Land cover changes and impact of shrimp aquaculture on the landscape in the Ceuta coastal lagoon system, Sinaloa, Mexico. *Ocean and Coastal Management*, 46(6), 583–600. https://doi.org/10.1016/S0964-5691(03)00036-X
- Anand, V., & Oinam, B. (2020). Future land use land cover prediction with special emphasis on urbanization and wetlands. *Remote Sensing Letters*, 11(3), 225–234. https://doi.org/10.1080/2150704X.2019.1704304
- Arfat, Y. (2010). Land use/land cover change detection and quantification A Case study in *Eastern Sudan*. MSc. thesis. Lund University, Sweden.
- Ariti, A. T., van Vliet, J., & Verburg, P. H. (2015). Land-use and land-cover changes in the Central Rift Valley of Ethiopia: Assessment of perception and adaptation of stakeholders. *Applied Geography*, 65, 28–37. https://doi.org/10.1016/j.apgeog.2015.10.002
- Asante-Yeboah, E., Ashiagbor, G., Asubonteng, K., Sieber, S., Mensah, J. C., & Fürst, C. (2022). Analyzing variations in size and intensities in land use dynamics for sustainable land use management: A case of the Coastal landscapes of South-Western Ghana. *Land*, *11*(6), 815. https://doi.org/10.3390/land11060815
- Asrat, P., & Simane, B. (2018). Farmers' perception of climate change and adaptation strategies in the Dabus watershed, North-West Ethiopia. *Ecological Processes*, 7(1), 1–13. https://doi.org/10.1186/s13717-018-0118-8

Ayal, D. Y., & Filho, W. L. (2017). Farmers' perceptions of climate variability and its adverse

impacts on crop and livestock production in Ethiopia. *Journal of Arid Environments*, 140, 20–28. https://doi.org/10.1016/j.jaridenv.2017.01.007

- Ayoub, A. T. (1999). Land degradation, rainfall variability and food production in the Sahelian zone of the Sudan. *Land Degradation and Development*, 10(5), 489–500. https://doi.org/10.1002/(SICI)1099-145X(199909/10)10:5<489::AID-LDR336>3.0.CO;2-U
- Baig, M. F., Mustafa, M. R. U., Baig, I., Takaijudin, H. B., & Zeshan, M. T. (2022). Assessment of land use land cover changes and future predictions using CA-ANN simulation for selangor, Malaysia. *Water*, 14(3), 402. https://doi.org/10.3390/w14030402
- Bannayan, M., Asadi, S., Nouri, M., & Yaghoubi, F. (2020). Time trend analysis of some agroclimatic variables during the last half century over Iran. *Theoretical and Applied Climatology*, *140*(3–4), 839–857. https://doi.org/10.1007/s00704-020-03105-7
- Barnieh, B. A., Jia, L., Menenti, M., Zhou, J., & Zeng, Y. (2020). Mapping land use land cover transitions at different spatiotemporal scales in West Africa. *Sustainability*, 12(20), 8565. https://doi.org/10.3390/su12208565
- Basse, R. M., Omrani, H., Charif, O., Gerber, P., & Bódis, K. (2014). Land use changes modelling using advanced methods: Cellular automata and artificial neural networks. The spatial and explicit representation of land cover dynamics at the cross-border region scale. *Applied Geography*, 53, 160–171. https://doi.org/10.1016/j.apgeog.2014.06.016
- Bhunia, G., Chatterjee, U., Kashyap, A., & Shit, P. (2021). Land reclamation and restoration strategies for sustainable development: Geospatial technology based approach. Elsevier Ltd.
- Biratu, A. A., Bedadi, B., Gebrehiwot, S. G., Melesse, A. M., Nebi, T. H., Abera, W., Tamene, L., & Egeru, A. (2022). Ecosystem service valuation along landscape transformation in Central Ethiopia. *Land*, 11(4), 500. https://doi.org/10.3390/land11040500
- Biro, K., Pradhan, B., Buchroithner, M., & Makeschin, F. (2013). Land use/land cover change analysis and its impact on soil properties in the Northern part of Gadarif region, Sudan. *Land Degradation and Development*, 24(1), 90–102. https://doi.org/10.1002/ldr.1116
- Bonilla-Moheno, M., & Aide, T. M. (2020). Beyond deforestation: Land cover transitions in Mexico. *Agricultural Systems*, *178*, 102734. https://doi.org/10.1016/j.agsy.2019.102734
- Breiman, L. (2001). Random Forests. *Machine Learning*, *45*, 5–32. https://doi.org/10.1109/ICCECE51280.2021.9342376
- Brovkin, V., Boysen, L., Arora, V. K., Boisier, J. P., Cadule, P., Chini, L., Claussen, M.,
 Friedlingstein, P., Gayler, V., Van den hurk, B. J. J. M., Hurtt, G. C., Jones, C. D., Kato, E.,
 De noblet-ducoudre, N., Pacifico, F., Pongratz, J., & Weiss, M. (2013). Effect of
 anthropogenic land-use and land-cover changes on climate and land carbon storage in
 CMIP5 projections for the twenty-first century. *Journal of Climate*, *26*(18), 6859–6881.
 https://doi.org/10.1175/JCLI-D-12-00623.1

- Bryan, E., Ringler, C., Okoba, B., Roncoli, C., Silvestri, S., & Herrero, M. (2013). Adapting agriculture to climate change in Kenya: Household strategies and determinants. *Journal of Environmental Management*, *114*, 26–35. https://doi.org/10.1016/j.jenvman.2012.10.036
- Buğday, E., & Erkan Buğday, S. (2019). Modeling and simulating land use/cover change using artificial neural network from remotely sensing data. *Cerne*, 25(2), 246–254. https://doi.org/10.1590/01047760201925022634
- Calzadilla, A., Zhu, T., Rehdanz, K., Tol, R. S. J., & Ringler, C. (2013). Economywide impacts of climate change on agriculture in Sub-Saharan Africa. *Ecological Economics*, 93, 150–165. https://doi.org/10.1016/j.ecolecon.2013.05.006
- Cánovas-García, F., Alonso-Sarría, F., Gomariz-Castillo, F., & Oñate-Valdivieso, F. (2017). Modification of the random forest algorithm to avoid statistical dependence problems when classifying remote sensing imagery. *Computers and Geosciences*, 103, 1–11. https://doi.org/10.1016/j.cageo.2017.02.012
- Cetin, O., & Basbag, S. (2010). Effects of climatic factors on cotton production in semi-arid regions A review. *Research on Crops*, *11*(3), 785–791.
- Chang, Y., Hou, K., Li, X., Zhang, Y., & Chen, P. (2018). Review of land use and land cover change research progress. *IOP Conference Series: Earth and Environmental Science*, *113*, 012087. https://doi.org/10.1088/1755-1315/113/1/012087
- Chen, H., Guo, J., Zhang, Z., & Xu, C. Y. (2013). Prediction of temperature and precipitation in Sudan and South Sudan by using LARS-WG in future. *Theoretical and Applied Climatology*, *113*(3–4), 363–375. https://doi.org/10.1007/s00704-012-0793-9
- Creutzig, F., Bren D'Amour, C., Weddige, U., Fuss, S., Beringer, T., Gläser, A., Kalkuhl, M., Steckel, J. C., Radebach, A., & Edenhofer, O. (2019). Assessing human and environmental pressures of global land-use change 2000-2010. *Global Sustainability*, 2, 1–17. https://doi.org/10.1017/sus.2018.15
- Daba, M. H., & You, S. (2022). Quantitatively Assessing the Future Land-Use/Land-Cover Changes and Their Driving Factors in the Upper Stream of the Awash River Based on the CA–Markov Model and Their Implications for Water Resources Management. Sustainability, 14(3). https://doi.org/10.3390/su14031538
- Dafalla, M. S., Abdel-Rahman, E. M., Siddig, K. H. A., Ibrahim, I. S., & Csaplovics, E. (2014). Land use and land cover changes in Northern Kordofan state of Sudan: A remotely sensed data analysis. In A. M. Melesse & W. A. S. G. Setegn (Eds.), *Nile River Basin* (pp. 269– 283). Springer, Cham.
- Dale, V. H. (1997). The relationship between land-use change and climate change. *Ecological Applications*, 7(3), 753–769.
- de Chazal, J., & Rounsevell, M. D. A. (2009). Land-use and climate change within assessments of biodiversity change: A review. *Global Environmental Change*, *19*(2), 306–315.

https://doi.org/10.1016/j.gloenvcha.2008.09.007

- De Sousa, C., Fatoyinbo, L., Neigh, C., Boucka, F., Angoue, V., & Larsen, T. (2020). Cloudcomputing and machine learning in support of country-level land cover and ecosystem extent mapping in Liberia and Gabon. *PLoS ONE*, 15, e0227438. https://doi.org/10.1371/journal.pone.0227438
- Deng, X., Zhao, C., & Yan, H. (2013). Systematic modeling of impacts of land use and land cover changes on regional climate: A review. *Advances in Meteorology*, 317678. https://doi.org/10.1155/2013/317678
- Déqué, M., Calmanti, S., Christensen, O. B., Dell Aquila, A., Maule, C. F., Haensler, A., Nikulin, G., & Teichmann, C. (2017). A multi-model climate response over tropical Africa at +2 °C. *Climate Services*, 7, 87–95. https://doi.org/10.1016/j.cliser.2016.06.002
- Deressa, T., Hassan, R. M., Alemu, T., Yesuf, M., & Ringler, C. (2008). Analyzing the determinants of farmers' choice of adaptation methods and perceptions of climate change in the Nile Basin of Ethiopia. *IFPRI Discussion Paper*, 798, 26. http://www.ifpri.org/publication/determinants-farmers-choice-adaptation-methods-andperceptions-climate-change-nile-basi-0%5Cnhttp://www.ifpri.org/sites/default/files/publications/ifpridp00798.pdf
- Ding, Z., Ali, E. F., Elmahdy, A. M., Ragab, K. E., Seleiman, M. F., & Kheir, A. M. S. (2021). Modeling the combined impacts of deficit irrigation, rising temperature and compost application on wheat yield and water productivity. *Agricultural Water Management*, 244(May 2020), 106626. https://doi.org/10.1016/j.agwat.2020.106626
- Dubovik, O., Schuster, G. L., Xu, F., Hu, Y., Bösch, H., Landgraf, J., & Li, Z. (2021). Grand challenges in satellite remote sensing. *Frontiers in Remote Sensing*, *2*, 619818. https://doi.org/10.3389/frsen.2021.619818
- El-Maayar, M., & Lange, M. A. (2013). A methodology to infer crop yield response to climate variability and change using long-term observations. *Atmosphere*, *4*(4), 365–382. https://doi.org/10.3390/atmos4040365
- El-Tantawi, A. M., Bao, A., Chang, C., & Liu, Y. (2019). Monitoring and predicting land use/cover changes in the Aksu-Tarim River Basin, Xinjiang-China (1990–2030). *Environmental Monitoring and Assessment*, 191(8), 480. https://doi.org/10.1007/s10661-019-7478-0
- Elagib, N. A. (2010). Trends in intra-and inter-annual temperature variabilities across Sudan. *Ambio*, *39*(5), 413–429. https://doi.org/10.1007/s13280-010-0042-3
- Elagib, N. A., & Elhag, M. M. (2011). Major climate indicators of ongoing drought in Sudan. *Journal of Hydrology*, 409(3–4), 612–625. https://doi.org/10.1016/j.jhydrol.2011.08.047
- Elagib, N. A., Khalifa, M., Rahma, A. E., Babker, Z., & Gamaledin, S. I. (2019). Performance of major mechanized rainfed agricultural production in Sudan: Sorghum vulnerability and
resilience to climate since 1970. *Agricultural and Forest Meteorology*, 276–277, 107640. https://doi.org/10.1016/j.agrformet.2019.107640

- Elagib, N. A., & Mansell, M. G. (2000). Recent trends and anomalies in mean seasonal and annual temperatures over Sudan. *Journal of Arid Environments*, 45(3), 263–288. https://doi.org/10.1006/jare.2000.0639
- Eldredge, E., Khalil, S. E. S., Nicholds, N., Abdalla, A. A., & Rydjeski, D. (1988). Changing rainfall patterns in Western Sudan. *Journal of Climatology*, 8(1), 45–53. https://doi.org/10.1002/joc.3370080105
- Elhag, K. M., & Zhang, W. (2018). Monitoring and assessment of drought focused on its impact on sorghum yield over sudan by using meteorological drought indices for the period 2001-2011. *Remote Sensing*, *10*(8), 1–21. https://doi.org/10.3390/rs10081231
- Ellis, E. C. (2011). Anthropogenic transformation of the terrestrial biosphere. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences,* 369(1938), 1010–1035. https://doi.org/10.1098/rsta.2010.0331
- Elramlawi, H. R., Mohammed, H. I., Elamin, A. W., Abdallah, O. A., & Taha, A. A. M. (2020). Adaptation of sorghum (Sorghum bicolor L. Moench) crop yield to climate change in Eastern dryland of Sudan. In W. L. Filho (Ed.), *Handbook of Climate Change Resilience*. Springer: Cham, Switzerland. https://doi.org/10.1007/978-3-319-71025-9
- Eltohami, A. B. E. A. (2016). Anthropgenic and climatic Factors : As causes of drought disaster in Sudan. 2nd World Irrigation Forum (WIF2), 1–8.
- Ewert, F., Rötter, R. P., Bindi, M., Webber, H., Trnka, M., Kersebaum, K. C., Olesen, J. E., van Ittersum, M. K., Janssen, S., Rivington, M., Semenov, M. A., Wallach, D., Porter, J. R., Stewart, D., Verhagen, J., Gaiser, T., Palosuo, T., Tao, F., Nendel, C., ... Asseng, S. (2015). Crop modelling for integrated assessment of risk to food production from climate change. *Environmental Modelling and Software*, 72, 287–303. https://doi.org/10.1016/j.envsoft.2014.12.003
- Ewunetu, A., Simane, B., Teferi, E., & Zaitchik, B. F. (2021). Land cover change in the blue nile river headwaters: Farmers' perceptions, pressures, and satellite-based mapping. *Land*, 10(1), 68. https://doi.org/10.3390/land10010068
- FAO. (2015a). FAO. Plan of Action (2015 2019): Resilient Livelihoods for Sustainable Agriculture, Food Security and Nutrition; FAO: Rome, Italy, 2015.
- FAO. (2015b). *National investment profile: Water for agriculture and energy in Sudan*. Khartoum/Rome.
- Feng, W., Ma, C., Zhao, G., & Zhang, R. (2020). FSRF: An Improved Random Forest for Classification. Proceedings of IEEE International Conference on Advances in Electrical Engineering and Computer Applications, AEECA, 173–178. https://doi.org/10.1109/AEECA49918.2020.9213456

- Fereres, E., Orgaz, F., & Gonzalez-Dugo, V. (2011). Reflections on food security under water scarcity. *Journal of Experimental Botany*, 62(12), 4079–4086. https://doi.org/10.1093/jxb/err165
- Floreano, I. X., & de Moraes, L. A. F. (2021). Land use/land cover (LULC) analysis (2009–2019) with Google Earth Engine and 2030 prediction using Markov-CA in the Rondônia State, Brazil. *Environmental Monitoring and Assessment*, 193(4), 239. https://doi.org/10.1007/s10661-021-09016-y
- Foguesatto, C. R., & Machado, J. A. D. (2021). What shapes farmers' perception of climate change? A case study of southern Brazil. *Environment, Development and Sustainability*, 23(2), 1525–1538. https://doi.org/10.1007/s10668-020-00634-z
- Fonseka, H. P. U., Zhang, H., Sun, Y., Su, H., Lin, H., & Lin, Y. (2019). Urbanization and its impacts on land surface temperature in Colombo Metropolitan Area, Sri Lanka, from 1988 to 2016. *Remote Sensing*, 11(8), 957. https://doi.org/10.3390/rs11080926
- Foody, G. M. (2020). Explaining the unsuitability of the kappa coefficient in the assessment and comparison of the accuracy of thematic maps obtained by image classification. *Remote Sensing of Environment*, 239, 111630. https://doi.org/10.1016/j.rse.2019.111630
- Gadallah, N. A. H., Adewole, N. A., & Ajayi, D. D. (2019). Potential of agroforestry as forest landscape restoration tool to solve forest cover loss cum food security in Sennar and Gedaref States, Sudan Nasradeen. *International Journal of Development and Sustainability*, 8(2), 199–210. www.isdsnet.com/ijds
- Gadallah, N. A. H., Hano, A., & Yagoub, Y. (2020). Characterizing forest cover changes based on satellite images cum forest dependents' data. *Agriculture and Forestry Journal*, 4(2), 63–70. https://doi.org/10.5281/zenodo.4310945
- Garrity, D., Dixon, J., & Boffa Jean-Marc. (2012). Understanding African Farming Systems, Science and Policy Implications. *Food Security in Africa: Bridging Research and Practice*, 1–55.
- Ge, Y., Hu, S., Ren, Z., Jia, Y., Wang, J., Liu, M., Zhang, D., Zhao, W., Luo, Y., Fu, Y., Bai, H., & Chen, Y. (2019). Mapping annual land use changes in China's poverty-stricken areas from 2013 to 2018. *Remote Sensing of Environment*, 232, 111285. https://doi.org/10.1016/j.rse.2019.111285
- Ghimire, B., Rogan, J., Galiano, V., Panday, P., & Neeti, N. (2012). An evaluation of bagging, boosting, and random forests for land-cover classification in Cape Cod, Massachusetts, USA. *GIScience and Remote Sensing*, 49(5), 623–643. https://doi.org/10.2747/1548-1603.49.5.623
- Gismondi, M. (2013). MOLUSCE-an open source land use change analyst. https://2013.foss4g.org/conf/programme/presentations/107/ (accessed on 30 May 2022).

Glover, E. K., & Elsiddig, E. A. (2012). The causes and consequences of environmental changes

in Gedaref, Sudan. *Land Degradation and Development*, 23, 339–349. https://doi.org/10.1002/ldr.2167

- Glover, M. K. (2017). Constraints associated with the marketing channel of lettuce and cabbage trade in Ghana. *Journal of Agriculture and Sustainability*, *10*(2), 116–143.
- Glover, M. K., & Elsiddig, E. A. (2012). The causes and consequences of environmental changes in Gedaref, Sudan. *Land Degradation and Development*, 23(4), 339–349. https://doi.org/10.1002/ldr.2167
- Godfray, C. J., Beddington, J. R., Crute, I. R., Haddad, L., Lawrence, D., Muir, J. F., Pretty, J., Robinson, S., Thomas, S. M., & Toulmin, C. (2010). Food security: The challenge of the present. *Science*, 327, 812–818. https://doi.org/10.1016/j.geoforum.2018.02.030
- Gounaridis, D., Chorianopoulos, I., Symeonakis, E., & Koukoulas, S. (2019). A Random Forest-Cellular Automata modelling approach to explore future land use/cover change in Attica (Greece), under different socio-economic realities and scales. *Science of the Total Environment*, 646, 320–335. https://doi.org/10.1016/j.scitotenv.2018.07.302
- Griffiths, P., van der Linden, S., Kuemmerle, T., & Hostert, P. (2013). A pixel-based landsat compositing algorithm for large area land cover mapping. *IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing*, 6(5), 1939–1404. https://doi.org/10.1109/jstars.2012.2228167
- Gyampoh, B. A., Amisah, S., Idinoba, M., & Nkem, J. N. (2009). Using traditional knowledge to cope with climate change in rural Ghana. *Unasylva*, 60, 70–74.
- Hamid, A. A., & Eltayeb, Y. H. (2017). *Space borne technology for Drought Monitoring in Sudan* (p. 21). https://www.un-spider.org/sites/default/files/Hamid.pdf
- Hammer, G., Chapman, S., Singh, V., Nguyen, C., Oosterom, E. van, McLean, G., Zheng, B., & Jordan, D. (2015). Grain sorghum varietal reactions to heat stress and environment. *The Grains Research and Development Corporation*, 212, 2–7.
- Hansen, M. C., Roy, D. P., Lindquist, E., Adusei, B., Justice, C. O., & Altstatt, A. (2008). A method for integrating MODIS and Landsat data for systematic monitoring of forest cover and change in the Congo Basin. *Remote Sensing of Environment*, 112(5), 2495–2513. https://doi.org/10.1016/j.rse.2007.11.012
- Hassan, R., Hertzler, G., & Benhin, J. K. A. (2009). Depletion of forest resources in Sudan: Intervention options for optimal control. *Energy Policy*, *37*(4), 1195–1203. https://doi.org/10.1016/j.enpol.2008.10.049
- Hennink, M. M. (2007). *International focus group research: A Handbook for the health and social sciences*. Cambridge University Press.
- Hermosilla, T., Wulder, M. A., White, J. C., Coops, N. C., & Hobart, G. W. (2015). An integrated landsat time series protocol for change detection and generation of annual gap-

free surface reflectance composites. *Remote Sensing of Environment*, *158*, 220–234. https://doi.org/10.1016/j.rse.2014.11.005

- Houghton, R. A. (1994). The worldwide extent of land-use change. *BioScience*, 44(5), 305–313. https://doi.org/10.2307/1312380
- Hurskainen, P., Adhikari, H., Siljander, M., Pellikka, P. K. E., & Hemp, A. (2019). Auxiliary datasets improve accuracy of object-based land use/land cover classification in heterogeneous savanna landscapes. *Remote Sensing of Environment*, 233, 111354. https://doi.org/10.1016/j.rse.2019.111354
- Hussein, M. A., Elgali, M. B., & Mustafa, R. H. (2022). Sudan agricultural markets performance under climate change. *Journal of Positive School Psychology*, *4*, 9955 – 9968. http://www.fao.org/3/a-bu259e.pdf
- Ibrahim, A. M. (2015). The impact of rainfall on the yields of staple crops sorghum and sesame in Sudan. *Journal of Plant Science and Research Open Science*, 2(2), 1–4.
- Idreas, A. E. L. A. (2015). *Effect of mechanized rain fed farming on vegetation cover and effect of shelter belts on environment at Ghadambaliya Area Gedaref State (SUDAN)*. PhD thesis. Sudan University of Science and Technology.
- IPCC. (2014). Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Cambridge University Press: Cambridge, UK. https://www.ipcc.ch/pdf/assessment-report/ar5/wg2/WGIIAR5-FrontMatterA_FINAL.pdf
- Issa, A. F. E. (2018). Assessment of the vegetation cover change at Qala El-Nahal locality -Gedaref state - Sudan. PhD thesis, Sudan University of Science and Technology, https://repository.sustech.edu/handle/123456789/23146.
- Jaramillo, S., Graterol, E., & Pulver, E. (2020). Sustainable Transformation of Rainfed to Irrigated Agriculture Through Water Harvesting and Smart Crop Management Practices. *Frontiers in Sustainable Food Systems*, 4(November). https://doi.org/10.3389/fsufs.2020.437086
- Jones, J. W., Hoogenboom, G., Porter, C. H., Boote, K. J., Batchelor, W. D., Hunt, L. A., Wilkens, P. W., Singh, U., Gijsman, A. J., & Ritchie, J. T. (2003). The DSSAT cropping system model. In *European Journal of Agronomy* (Vol. 18, Issues 3–4). https://doi.org/10.1016/S1161-0301(02)00107-7
- Kafy, A. Al, Naim, M. N. H., Subramanyam, G., Faisal, A. Al, Ahmed, N. U., Rakib, A. Al, Kona, M. A., & Sattar, G. S. (2021). Cellular Automata approach in dynamic modelling of land cover changes using RapidEye images in Dhaka, Bangladesh. *Environmental Challenges*, 4, 100084. https://doi.org/10.1016/j.envc.2021.100084
- Kamwi, J. M., Chirwa, P. W. C., Manda, S. O. M., Graz, P. F., & Kätsch, C. (2015). Livelihoods, land use and land cover change in the Zambezi Region, Namibia. *Population and Environment*, 37(2), 207–230. https://doi.org/10.1007/s11111-015-0239-2

- Kangalawe, R. Y. M., & Lyimo, J. G. (2013). Climate change, adaptive strategies and rural livelihoods in semiarid Tanzania. *Natural Resources*, 04(03), 266–278. https://doi.org/10.4236/nr.2013.43034
- Kanianska, R. (2016). Agriculture and its impact on land Use, environment, and ecosystem services. In A. Almusaed (Ed.), *Landscape ecology : The influences of land use and anthropogenic impacts of landscape creation* (pp. 3–26).
- Kendall, M. G. (1948). Rank correlation methods. Charles Griffin:
- Ketema, H., Wei, W., Legesse, A., Zinabu, W., Temesgen, H., & Yirsaw, E. (2021). Ecosystem service variation and its importance to the wellbeing of smallholder farmers in contrasting agro-ecological zones of East African Rift. *Food and Energy Security*, 10(4), 1–18. https://doi.org/10.1002/fes3.310
- Khalifa, E. A. (2016). Volatility of sorghum production under rainfall system in Gadarif State, Sudan. *Journal of Agricultural and Socio-Economic Sciences*, 8, 137–144.
- Kindu, M., Schneider, T., Teketay, D., & Knoke, T. (2015). Drivers of land use/land cover changes in Munessa-Shashemene landscape of the south-central highlands of Ethiopia. *Environmental Monitoring and Assessment*, 187(7). https://doi.org/10.1007/s10661-015-4671-7
- Klayman, D. (2022). Statistical and graphical procedures for evaluating data quality : The case of large-scale data collection and monitoring system. *Journal of Applied Sociology*, *19*(2), 22–34.
- Kleemann, J., Baysal, G., Bulley, H. N. N., & Fürst, C. (2017). Assessing driving forces of land use and land cover change by a mixed-method approach in north-eastern Ghana, West Africa. *Journal of Environmental Management*, 196, 411–442. https://doi.org/10.1016/j.jenvman.2017.01.053
- Klenk, N., & Meehan, K. (2015). Climate change and transdisciplinary science: Problematizing the integration imperative. *Environmental Science and Policy*, 54, 160–167. https://doi.org/10.1016/j.envsci.2015.05.017
- Kouassi, J. L., Gyau, A., Diby, L., Bene, Y., & Kouamé, C. (2021). Assessing land use and land cover change and farmers' perceptions of deforestation and land degradation in south-west Côte d'Ivoire, West Africa. *Land*, 10(4), 429. https://doi.org/10.3390/land10040429
- Koudahe, K., Kayode, A. J., Samson, A. O., Adebola, A. A., & Djaman, K. (2017). Trend analysis in standardized precipitation index and standardized anomaly index in the context of climate change in Southern Togo. *Atmospheric and Climate Sciences*, 07(04), 401–423. https://doi.org/10.4236/acs.2017.74030
- Kumazaki, T., Yamada, Y., Karaya, S., Tokumitsu, T., Hirano, T., Yasumoto, S., Katsuta, M., & Michiyama, H. (2008). Effects of day length and air temperature on stem growth and flowering in sesame. *Plant Production Science*, 11(2), 178–183.

https://doi.org/10.1626/pps.11.178

- Lambin, E. F., & Ehrlich, D. (1997). Land-cover changes in sub-Saharan Africa (1982-1991): Application of a change index based on remotely sensed surface temperature and vegetation indices at a continental scale. *Remote Sensing of Environment*, 61(2), 181–200. https://doi.org/10.1016/S0034-4257(97)00001-1
- Lambin, E. F., & Meyfroidt, P. (2011). Global land use change, economic globalization, and the looming land scarcity. *Proceedings of the National Academy of Sciences of the United States of America*, 108(9), 3465–3472. https://doi.org/10.1073/pnas.1100480108
- Laux, P., Jäckel, G., Tingem, M., & Kunstmann, H. (2009). Onset of the rainy season and crop yield in West Africa. *Geophysical Research Abstracts*, 11, 11276.
- Licker, R., Johnston, M., Foley, J. A., Barford, C., Kucharik, C. J., Monfreda, C., & Ramankutty, N. (2010). Mind the gap: How do climate and agricultural management explain the "yield gap" of croplands around the world? *Global Ecology and Biogeography*, 19(6), 769–782. https://doi.org/10.1111/j.1466-8238.2010.00563.x
- Lim, B., Rector, I., Angell, P. S., Nuttall, N., Leach, A., Barton-dock, M., & Evans, W. (2011). Decision making in a chalenging climate: adaptation challenges and choices. In *World resources report*.
- Lin, L., Hao, Z., Post, C. J., Mikhailova, E. A., Yu, K., Yang, L., & Liu, J. (2020). Monitoring land cover change on a rapidly urbanizing island using google earth engine. *Applied Sciences*, 10, 7336. https://doi.org/10.3390/app10207336
- Loh, P. S., Alnoor, H. I. M., & He, S. (2020). Impact of climate change on vegetation cover at South Port Sudan area. *Climate*, 8(10), 114. https://doi.org/10.1007/978-981-32-9174-4_40
- López-Santiago, C. A., Oteros-Rozas, E., Martín-López, B., Plieninger, T., Martín, E. G., & González, J. A. (2014). Using visual stimuli to explore the social perceptions of ecosystem services in cultural landscapes: The case of transhumance in Mediterranean Spain. *Ecology* and Society, 19(2). https://doi.org/10.5751/ES-06401-190227
- Maddison, D. (2007). *The perception of and adaptation to climate change in Africa*. The World Bank Development Research Group. http://www.ceepa.co.za/docs/CDPNo10.pdf
- Maharjan, S. K., Sigdel, E. R., Sthapit, B. R., & Regmi, B. R. (2011). Tharu community's perception on climate changes and their adaptive initiations to withstand its impacts in Western Terai of Nepal. *International NGO Journal*, 6(2), 035–042. https://doi.org/10.5897/NGO10.003
- Mahgoub, F. (2014). Current status of agriculture and future challenges in Sudan. In *Current African Issues* (Issue 57). http://sites.duke.edu/minerva/files/2014/04/2013-08-28_CGGC_Report_Wheat_GVC_and_food_security_in_MENA.pdf%255Cnhttp://elibrary.imf.org/view/IMF001/20530-9781484305164/20530-9781484305164/20530-9781484305164/20530-9781484305164.xml%255Cnhttp://www.mafhoum.com/press7/206E13.p

- Mahmood, R., Pielke, R. A., Hubbard, K. G., Niyogi, D., Bonan, G., Lawrence, P., McNider, R., McAlpine, C., Etter, A., Gameda, S., Qian, B., Carleton, A., Beltran-Przekurat, A., Chase, T., Quintanar, A. I., Adegoke, J. O., Vezhapparambu, S., Conner, G., Asefi, S., ... Syktus, J. (2010). Impacts of land use/land cover change on climate and future research priorities. *Bulletin of the American Meteorological Society*, *91*(1), 37–46. https://doi.org/10.1175/2009BAMS2769.1
- Mann, H. B. (1945). Nonparametric tests against trend. *Econometrica*, 13(3), 245–259.
- Marizin, J., Bonnet, P., Bessaoud, O., & Ton-Nu Christine. (2017). *Small-scale family farming in the Near-East and North-Africa region: Synthesis*. Food and Agriculture Organization of the United Nations. www.fao.org/publications
- Martin, G., Martin-Clouaire, R., & Duru, M. (2013). Farming system design to feed the changing world. A review. Agronomy for Sustainable Development, 33(1), 131–149. https://doi.org/10.1007/s13593-011-0075-4
- Matlhodi, B., Kenabatho, P. K., Parida, B. P., & Maphanyane, J. G. (2019). Evaluating land use and land cover change in the Gaborone dam catchment, Botswana, from 1984-2015 using GIS and remote sensing. *Sustainability*, *11*(19). https://doi.org/10.3390/su11195174
- Mbaabu, P. R., Ng, W. T., Schaffner, U., Gichaba, M., Olago, D., Choge, S., Oriaso, S., & Eckert, S. (2019). Spatial evolution of prosopis invasion and its effects on LULC and livelihoods in Baringo, Kenya. *Remote Sensing*, 11(10), 1217. https://doi.org/10.3390/rs11101217
- McCarthy, J. J., Canziani, O. F., Leary, N. A., Dokken, D. J., & White, K. S. (2001). Climate change 2001: impacts, adaptation, and vulnerability: contribution of Working Group II to the third assessment report of the Intergovernmental Panel on Climate Change. In *IPCC*. Cambridge University Press. http://www.ipcc.ch/ipccreports/tar/wg2/index.htm
- Mekuyie, M., Jordaan, A., & Melka, Y. (2018). Land-use and land-cover changes and their drivers in rangeland-dependent pastoral communities in the southern Afar Region of Ethiopia. *African Journal of Range and Forage Science*, 35(1), 33–43. https://doi.org/10.2989/10220119.2018.1442366
- Mertz, O., Halsnæs, K., Olesen, J. E., & Rasmussen, K. (2009). Adaptation to climate change in developing countries. *Environmental Management*, 43(5), 743–752. https://doi.org/10.1007/s00267-008-9259-3
- Mertz, O., Mbow, C., Reenberg, A., & Diouf, A. (2009). Farmers' perceptions of climate change and agricultural adaptation strategies in rural sahel. *Environmental Management*, 43(5), 804–816. https://doi.org/10.1007/s00267-008-9197-0
- Midekisa, A., Holl, F., Savory, D. J., Andrade-Pacheco, R., Gething, P. W., Bennett, A., & Sturrock, H. J. W. (2017). Mapping land cover change over continental Africa using Landsat and Google Earth Engine cloud computing. *PLoS ONE*, *12*(9), e0184926. https://doi.org/10.1371/journal.pone.0184926

- Mijić, A., Liović, I., Kovačević, V., & Pepó, P. (2012). Impact of weather conditions on variability in sunflower yield over years in eastern parts of Croatia and Hungary. Acta Agronomica Hungarica, 60(4), 397–405. https://doi.org/10.1556/AAgr.60.2012.4.10
- Miller, C. (2005). Power, land and ethnicity in the Kassala-Gedaref States. Land, Ethnicity and Political Legitimacy in Eastern Sudan.Le Caire,, CEDEJ, 3–58.
- Misra, A. K., Lata, K., & Shukla, J. B. (2014). Effects of population and population pressure on forest resources and their conservation: A modeling study. *Environment, Development and Sustainability*, 16(2), 361–374. https://doi.org/10.1007/s10668-013-9481-x
- Mohamed, M. S. D. (2006). *Mapping and assessment of land use/land cover using remote sensing and GIS in North Kordofan State, Sudan* [PhD. thesis. Dresden University of Technology]. http://nbn-resolving.de/urn:nbn:de:swb:14-1171981536181-44423 LA - eng
- Mohammed, H. F. M. (2016). Use of MODIS satellite imagery to generate a historical background of wildland fire regime in the Southeastern part of Gedaref state [PhD. thesis,Sudan University of Science and Technology]. http://repository.sustech.edu//handle/123456789/14202
- Mohmmed, A., Li, J., Elaru, J., Elbashier, M. M. A., Keesstra, S., Artemi, C., Martin, K., Reuben, M., & Teffera, Z. (2018). Assessing drought vulnerability and adaptation among farmers in Gadaref region, Eastern Sudan. *Land Use Policy*, 70, 402–413. https://doi.org/10.1016/j.landusepol.2017.11.027
- Morgan, H. (2019). Sudan: One of 10 countries most vulnerable to climate change.https://www.aljazeera.com/news/2019/11/sudan-10-countries-vulnerable-climatechange-191121123822186.html. https://www.aljazeera.com/news/2019/11/sudan-10countries-vulnerable-climate-change-191121123822186.html
- Moyo, M., Dorward, P., & Craufurd, P. (2017). Characterizing long term rainfall data for estimating climate risk in semi-arid Zimbabwe. In W. L. Filho, S. Belay, J. Kalangu, W. Menas, P. Munishi, & K. Musiyiwa (Eds.), *Climate Change Adaptation in Africa* (pp. 661–675). Springer International Publishing. https://doi.org/10.1007/978-3-319-49520-0_41
- Msongaleli, B. M., Tumbo, S. D., Kihupi, N. I., & Rwehumbiza, F. B. (2017). Performance of sorghum varieties under variable rainfall in Central Tanzania. *International Scholarly Research Notices*, 2017, 1–10. https://doi.org/10.1155/2017/2506946
- Mudereri, B. T., Abdel-Rahman, E. M., Ndlela, S., Makumbe, L. D. M., Nyanga, C. C., Tonnang, H. E. Z., & Mohamed, S. A. (2022). Integrating the strength of multi-aate Sentinel-1 and-2 datasets for detecting mango (Mangifera indica L.) orchards in a semi-arid environment in Zimbabwe. *Sustainability*, 14, 5741. https://doi.org/10.3390/su14105741
- Muhammad, R., Zhang, W., Abbas, Z., Guo, F., & Gwiazdzinski, L. (2022). Spatiotemporal change analysis and prediction of future land use and land cover changes using QGIS MOLUSCE Plugin and remote sensing big data: A cases study of Linyi, China. *Land*, 11(3), 419. https://doi.org/10.3390/land11030419

- Mumtaz, M., Oliveira, J. A. P. de, & Ali, S. H. (2019). Climate change impacts and adaptation in agricultural sector: The case of local responses in Punjab, Pakistan. In S. Hussain (Ed.), *Climate Change and Agriculture* (pp. 1–14). IntechOpen. https://doi.org/10.5772/intechopen.83553
- Mundia, C. W., Secchi, S., Akamani, K., & Wang, G. (2019). A regional comparison of factors affecting global sorghum production: The case of North America, Asia and Africa's Sahel. *Sustainability*, *11*(7), 2135. https://doi.org/10.3390/su11072135
- Munthali, M. G., Davis, N., Adeola, A. M., Botai, J. O., Kamwi, J. M., Chisale, H. L. W., & Orimoogunje, O. O. I. (2019). Local perception of drivers of Land-Use and Land- Cover change dynamics across Dedza district, Central Malawi region. *Sustainability*, 11(3), 832. https://doi.org/10.3390/su11030832
- Murenzi, H. (2019). *Investigating the effect of climate variability and change and change on maize yield in Rwanda*. Master thesis, University of Nairobi, Kenya.
- Musa, M. A., Peters, K. J., & Ahmed, M. K. A. (2006). On farm characterization of Butana and Kenana cattle breed production systems in Sudan. *Livestock Research for Rural Development*, 18(12).
- Näschen, K., Diekkrüger, B., Evers, M., Höllermann, B., Steinbach, S., & Thonfeld, F. (2019). The impact of land use/land cover change (LULCC) on water resources in a tropical catchment in Tanzania under different climate change scenarios. *Sustainability*, *11*, 7083. https://doi.org/10.3390/su11247083
- Nath, B., Niu, Z., & Singh, R. P. (2018). Land Use and Land Cover changes, and environment and risk evaluation of Dujiangyan city (SW China) using remote sensing and GIS techniques. *Sustainability (Switzerland)*, 10(12), 4631. https://doi.org/10.3390/su10124631
- Nath, R., Chakraborty, P. K., & Chakraborty, A. (2001). Effect of climatic variation on yield of sesame (Sesamum indicum L.) at different dates of sowing. *Journal of Agronomy and Crop Science*, 186(2), 97–102. https://doi.org/10.1046/j.1439-037X.2001.00456.x
- Nelson, G. C., Rosegrant, M. W., Palazzo, A., Gray, I., Ingersoll, C., Robertson, R., Tokgoz, S., Zhu, T., Sulser, T. B., Ringler, C., Msangi, S., & You, L. (2010). *Food security challenges* to 2050 and Beyond. https://cgspace.cgiar.org/handle/10568/33400
- Ngetich, K. F., Mucheru-Muna, M., Mugwe, J. N., Shisanya, C. A., Diels, J., & Mugendi, D. N. (2014). Length of growing season, rainfall temporal distribution, onset and cessation dates in the Kenyan highlands. *Agricultural and Forest Meteorology*, 188, 24–32. https://doi.org/10.1016/j.agrformet.2013.12.011
- Nicholls, N. (1997). Increased Australian wheat yield due to recent climate trends. *Nature*, *3*87, 484–485.
- Ntirenganya, F. (2018). Analysis of rainfall variability in rwanda for small-scale farmars coping strategies to climate variability. *East African Journal of Science and Technology*, 8(1), 75–

96.

- Ojara, M. A., Lou, Y., Aribo, L., Namumbya, S., & Uddin, M. J. (2020). Dry spells and probability of rainfall occurrence for Lake Kyoga Basin in Uganda, East Africa. *Natural Hazards*, *100*(2), 493–514. https://doi.org/10.1007/s11069-019-03822-x
- Osman, M. A. A., Onono, J. O., Olaka, L. A., Elhag, M. M., & Abdel-Rahman, E. M. (2021). Climate variability and change affect crops yield under rainfed conditions: A case study in gedaref state, sudan. *Agronomy*, 11(9), 1680. https://doi.org/10.3390/agronomy11091680
- Öztürk, Ö., & Şaman, O. (2012). Effects of different plant densities on the yield and quality of second crop sesame. *International Journal of Agricultural and Biosystems Engineering*, 6(9), 644–649.
- Pathak, P., Sahrawat, K. L., Wani, S. P., Sachan, R. C., & Sudi, R. (2009). Opportunities for water harvesting and supplemental irrigation for improving rainfed agriculture in semi-arid areas. *Rainfed Agriculture: Unlocking the Potential*, 197–221. https://doi.org/10.1079/9781845933890.0197
- Pelletier, C., Valero, S., Inglada, J., Champion, N., & Dedieu, G. (2016). Assessing the robustness of random forests to map land cover with high resolution satellite image time series over large areas. *Remote Sensing of Environment*, 187, 156–168. https://doi.org/10.1016/j.rse.2016.10.010
- Peltonen-Sainio, P., Jauhiainen, L., Trnka, M., Olesen, J. E., Calanca, P., Eckersten, H., Eitzinger, J., Gobin, A., Kersebaum, K. C., Kozyra, J., Kumar, S., Dalla Marta, A., Micale, F., Schaap, B., Seguin, B., Skjelvåg, A. O., & Orlandini, S. (2010). Coincidence of variation in yield and climate in Europe. *Agriculture, Ecosystems and Environment*, 139(4), 483–489. https://doi.org/10.1016/j.agee.2010.09.006
- Penman, J., Gytarsky, M., Hiraishi, T., Krug, T., Kruger, D., Pipatti, R., Buendia, L., Miwa, K., Ngara, T., Tanabe, K., & Wagner, F. (2003). Good practice guidance for land use, land-use change and forestry. In *Comptes Rendus - Biologies*. Institute for Global Environmental Strategies (IGES).
- Petersen, M., Bergmann, C., Roden, P., & Nüsser, M. (2021). Contextualizing land-use and landcover change with local knowledge: A case study from Pokot Central, Kenya. *Land Degradation & Development*, 32(10), 2992-3007.
- Phan, N. T., Kuch, V., & Lehnert, L. W. (2020). Land cover classification using google earth engine and random forest classifier-the role of image composition. *Remote Sensing*, *12*, 2411. https://doi.org/10.3390/RS12152411
- Platts1, P. J., Omeny, P. A., & Marchant, R. (2015). AFRICLIM: high-resolution climate projections for ecological applications in Africa. *African Journal of Ecology*, 53(1), 103– 108.
- Pontius, R. G., & Millones, M. (2011). Death to Kappa: Birth of quantity disagreement and

allocation disagreement for accuracy assessment. *International Journal of Remote Sensing*, 32(15), 4407–4429. https://doi.org/10.1080/01431161.2011.552923

- Poudel, S., & Shaw, R. (2016). The relationships between climate variability and crop yield in a mountainous environment: A case study in Lamjung District, Nepal. *Climate*, 4(1), 13. https://doi.org/10.3390/cli4010013
- Pretty, J., Toulmin, C., & Williams, S. (2011). Sustainable intensification in African agriculture. *International Journal of Agricultural Sustainability*, 9(1), 5–24. https://doi.org/10.3763/ijas.2010.0583
- Qiang, Y., & Lam, N. S. N. (2015). Modeling land use and land cover changes in a vulnerable coastal region using artificial neural networks and cellular automata. *Environmental Monitoring and Assessment*, 187, 1–16. https://doi.org/10.1007/s10661-015-4298-8
- Qiong, O. U. (2017). A Brief introduction to Perception. *Studies in Literature and Language*, *15*(4), 18–28. https://doi.org/10.3968/10055
- Qu, L., Chen, Z., Li, M., Zhi, J., & Wang, H. (2021). Accuracy improvements to pixel-based and object-based LULC classification with auxiliary datasets from google earth engine. *Remote Sensing*, 13(3), 453. https://doi.org/10.3390/rs13030453
- Rahman, M. T. U., Tabassum, F., Rasheduzzaman, M., Saba, H., Sarkar, L., Ferdous, J., Uddin, S. Z., & Zahedul Islam, A. Z. M. (2017). Temporal dynamics of land use/land cover change and its prediction using CA-ANN model for southwestern coastal Bangladesh. *Environmental Monitoring and Assessment*, 189(11), 565. https://doi.org/10.1007/s10661-017-6272-0
- Ray, D. K., Mueller, N. D., West, P. C., & Foley, J. A. (2013). Yield trends are insufficient to double global crop production by 2050. *PLoS ONE*, 8(6), e66428. https://doi.org/10.1371/journal.pone.0066428
- Raza, A., Razzaq, A., Mehmood, S. S., Zou, X., Zhang, X., Lv, Y., & Xu, J. (2019). Impact of climate change on crops adaptation and strategies to tackle its outcome: A review. *Plants*, 8(2), 34. https://doi.org/10.3390/plants8020034
- Rezaei, E. E., Webber, H., Gaiser, T., Naab, J., & Ewert, F. (2015). Heat stress in cereals: Mechanisms and modelling. *European Journal of Agronomy*, 64, 98–113. https://doi.org/10.1016/j.eja.2014.10.003
- Roco, L., Engler, A., Bravo-Ureta, B. E., & Jara-Rojas, R. (2015). Farmers' perception of climate change in mediterranean Chile. *Regional Environmental Change*, 15(5), 867–879. https://doi.org/10.1007/s10113-014-0669-x
- Rowhani, P., Lobell, D. B., Linderman, M., & Ramankutty, N. (2011). Climate variability and crop production in Tanzania. *Agricultural and Forest Meteorology*, 151(4), 449–460. https://doi.org/10.1016/j.agrformet.2010.12.002

- Saputra, M. H., & Lee, H. S. (2019). Prediction of land use and land cover changes for North Sumatra, Indonesia, using an artificial-neural-network-based cellular automaton. *Sustainability*, 11, 3024. https://doi.org/10.3390/su11113024
- Sawa, B. A., & Adebayo, A. A. (2018). Effects of pentad dry spells on the yield of some crops in the semi-arid eco-climatic region of northern Nigeria. *The Zaria Geographer*, *19*, 49–60.
- Schmidhuber, J., & Tubiello, F. N. (2007). Global food security under climate change. Proceedings of the National Academy of Sciences of the United States of America, 104(50), 19703–19708. https://doi.org/10.1073/pnas.0701976104
- Sen, P. K. (1968). Estimates of the regression coefficient based on Kendall's Tau. Journal of the American Statistical Association, 63(324), 1379–1389. https://doi.org/10.1080/01621459.1968.10480934
- Setiawan, B. I. (2020). A simple method to determine patterns of wet and dry seasons. *IOP Conference Series: Earth and Environmental Science*, 542, 012055. https://doi.org/10.1088/1755-1315/542/1/012055
- Seto, K. C., Fragkias, M., Guneralp, B., & Reilly, M. K. (2011). A meta-analysis of global urban land expansion. *PloS One*, 6(8), e23777. https://doi.org/10.1371/Citation
- Shongwe, P., & Manyatsi, A. M. (2014). Factors influencing the choice of climate change adaptation strategies by households: A case of Mpolonjeni area development programme (ADP) in Swaziland Swaziland. *Journal of Agricultural Studies*, 2, 86–98.
- Sibanda, S., Grab, S. W., & Ahmed, F. (2020). Long-term rainfall characteristics in the Mzingwane catchment of south-western Zimbabwe. *Theoretical and Applied Climatology*, *139*(3–4), 935–948. https://doi.org/10.1007/s00704-019-03020-6
- Siddig, K., Stepanyan, D., Wiebelt, M., Grethe, H., & Zhu, T. (2020). Climate change and agriculture in the Sudan: Impact pathways beyond changes in mean rainfall and temperature. *Ecological Economics*, 169, 106566. https://doi.org/10.1016/j.ecolecon.2019.106566
- Simane, B., Zaitchik, B. F., & Foltz, J. D. (2016). Agroecosystem specific climate vulnerability analysis: application of the livelihood vulnerability index to a tropical highland region. *Mitigation and Adaptation Strategies for Global Change*, 21(1), 39–65. https://doi.org/10.1007/s11027-014-9568-1
- Singh, P. K., & Chudasama, H. (2021). Pathways for climate change adaptations in arid and semi-arid regions. *Journal of Cleaner Production*, 284, 124744. https://doi.org/10.1016/j.jclepro.2020.124744
- Solomon, N., Birhane, E., Tadesse, T., Treydte, A. C., & Meles, K. (2017). Carbon stocks and sequestration potential of dry forests under community management in Tigray, Ethiopia. *Ecological Processes*, 6(1). https://doi.org/10.1186/s13717-017-0088-2

Stern, B. R., Parsons, D., Stern, D., & Torgbor, F. (2021). R-Instat climatic guide (pp. 1–283).

- Sulieman, H. M. (2008). *Mapping and modelling of vegetation changes in the Southern Gadarif region*, *Sudan, using remote sensing*. PhD. thesis. Dres- den University of Technology, Germany.
- Sulieman, H. M., & Elagib, N. A. (2012). Implications of climate, land-use and land-cover changes for pastoralism in eastern Sudan. *Journal of Arid Environments*, 85, 132–141. https://doi.org/https://doi.org/10.1016/j.jaridenv.2012.05.001
- Sulieman, Hussein M. (2010). Expansion of mechanised rain-fed agriculture and land-use/landcover change in Southern Gadarif, Sudan. *African Journal of Agricultural Research*, 5(13), 1609–1615. https://doi.org/10.5897/AJAR09.078
- Sulieman, Hussein M. (2013). *LDPI Working Paper 19. Land grabbing along livestock migration routes in Gadarif State, Sudan: Impacts on pastoralism and the environment.* The Land Deal Politics Initiative.
- Sultan, B., Bella-Medjo, M., Berg, A., Quirionb, P., & Janicot, S. (2008). Multi-scales and multisites analyses of the role of rainfall in cotton yields in West Africa. *International Journal of Climatology*, *30*, 58–71. https://doi.org/10.1002/joc
- Tadele, Z. (2016). Drought Adaptation in Millets. In A. Shanker & C. Shanker (Eds.), Abiotic and Biotic Stress in Plants: Recent Advances and Future Perspectives (pp. 639–662). IntechOpen. https://doi.org/10.5772/61929
- Tadross, M. A., Hewitson, B. C., & Usman, M. T. (2005). The interannual variability on the onset of the maize growing season over South Africa and Zimbabwe. *Journal of Climate*, *18*(16), 3356–3372. https://doi.org/10.1175/JCLI3423.1
- Tamiminia, H., Salehi, B., Mahdianpari, M., Quackenbush, L., Adeli, S., & Brisco, B. (2020). Google Earth Engine for geo-big data applications: A meta-analysis and systematic review. *ISPRS Journal of Photogrammetry and Remote Sensing*, 164, 152–170. https://doi.org/10.1016/j.isprsjprs.2020.04.001
- Taylor, J. R., & Lovell, S. T. (2012). Mapping public and private spaces of urban agriculture in Chicago through the analysis of high-resolution aerial images in Google Earth. *Landscape* and Urban Planning, 108(1), 57–70. https://doi.org/10.1016/j.landurbplan.2012.08.001
- Tesfahunegn, G. B., Mekonen, K., & Tekle, A. (2016). Farmers' perception on causes, indicators and determinants of climate change in northern Ethiopia: Implication for developing adaptation strategies. *Applied Geography*, 73, 1–12. https://doi.org/10.1016/j.apgeog.2016.05.009
- The World Bank Group. (2016). Accelerating climate-resilient and low-carbon development. *Pprogress report on the implementation of the Africa climate business plan.* International Bank for Reconstruction and Development / The World Bank 1818 H Street NW, Washington, DC 20433.

- Thornton, P. K., Ericksen, P. J., Herrero, M., & Challinor, A. J. (2014). Climate variability and vulnerability to climate change: A review. *Global Change Biology*, *20*(11), 3313–3328. https://doi.org/10.1111/gcb.12581
- Tian, P., Li, J., Gong, H., Pu, R., Cao, L., Shao, S., Shi, Z., Feng, X., Wang, L., & Liu, R. (2019). Research on land use changes and ecological risk assessment in Yongjiang River basin in Zhejiang Province, China. *Sustainability*, *11*(10), 2817. https://doi.org/10.3390/su11102817
- Tong, X., Brandt, M., Hiernaux, P., Herrmann, S., Rasmussen, L. V., Rasmussen, K., Tian, F., Tagesson, T., Zhang, W., & Fensholt, R. (2020). The forgotten land use class: Mapping of fallow fields across the Sahel using Sentinel-2. *Remote Sensing of Environment*, 239, 111598. https://doi.org/10.1016/j.rse.2019.111598
- Traore, K., Sidibe, D. K., Coulibaly, H., & Bayala, J. (2017). Optimizing yield of improved varieties of millet and sorghum under highly variable rainfall conditions using contour ridges in Cinzana, Mali. *Agriculture and Food Security*, 6(1), 1–13. https://doi.org/10.1186/s40066-016-0086-0
- Twisa, S., & Buchroithner, M. F. (2019). Land-use and land-cover (LULC) change detection in Wami river basin, Tanzania. *Land*, 8(9), 136. https://doi.org/10.3390/land8090136
- UNEP. (2007). *Sudan: post-conflict environmental assessment.* https://postconflict.unep.ch/publications/UNEP_Sudan.pdf
- van Vliet, N., Reenberg, A., & Rasmussen, L. V. (2013). Scientific documentation of crop land changes in the Sahel: A half empty box of knowledge to support policy? *Journal of Arid Environments*, 95, 1–13. https://doi.org/10.1016/j.jaridenv.2013.03.010
- Verburg, P. H., Ritsema van Eck, J. R., de Nijs, T. C. M., Dijst, M. J., & Schot, P. (2004). Determinants of land-use change patterns in the Netherlands. *Environment and Planning B: Planning and Design*, 31(1), 125–150. https://doi.org/10.1068/b307
- Warrens, M. J. (2015). Properties of the quantity disagreement and the allocation disagreement. *International Journal of Remote Sensing*, *36*(5), 1439–1446. https://doi.org/10.1080/01431161.2015.1011794
- Wolf, J., Allice, I., & Bell, T. (2013). Values, climate change, and implications for adaptation: Evidence from two communities in Labrador, Canada. *Global Environmental Change*, 23(2), 548–562. https://doi.org/10.1016/j.gloenvcha.2012.11.007
- Wondie, M., Schneider, W., Melesse, A. M., & Teketay, D. (2011). Spatial and temporal land cover changes in the simen mountains national park, a world heritage site in Northwestern Ethiopia. *Remote Sensing*, *3*(4), 752–766. https://doi.org/10.3390/rs3040752
- Wu, F., Mo, C., & Dai, X. (2022). Analysis of the driving force of land use change based on geographic detection and simulation of future land use scenarios. *Sustainability*, 14(9), 5254. https://doi.org/10.3390/su14095254

- Yagoub, Y. E., Siddig, A. A. H., Musa, O. S., Bo, Z., Zhongqin, L., & Feiteng, W. (2017). Assessing the impacts of land use changes on vegetation cover in Eastern Sudan. *International Journal of Research in Agricultural Sciences*, 4(2), 70–75. https://www.researchgate.net/publication/315808477
- Yamane, Y. (1967). Mathematical formulae for sample size determination. In J. Mathemetics (Vol. 1, Issue 2, pp. 1–29).
- You, L., Ringler, C., Wood-Sichra, U., Robertson, R., Wood, S., Zhu, T., Nelson, G., Guo, Z., & Sun, Y. (2011). What is the irrigation potential for Africa? A combined biophysical and socioeconomic approach. *Food Policy*, *36*(6), 770–782. https://doi.org/10.1016/j.foodpol.2011.09.001
- Younis, A. Y. I., Bello, N. J., Togun, A. O., Hamad, G. E. A. I., Aïssata, S. D., Omer, S. O., & Nasrealdin, M. A. (2022). Farmers' perceptions of climate change and its impact on gum Talha (Acacia seyal var. seyal) production in Bahar Alarab locality, East Darfur State, Sudan. *Journal of Applied and Natural Science*, 14(2), 349–361. https://doi.org/10.31018/jans.v14i2.3392
- Yousif, L. A., & Babiker, E. H. (2015). Effect of conservation agriculture on sorghum yield in rainfed areas southern Gedarif state, Sudan. *Journal of Agricultural Science and Engineering*, *2*, 89–94. https://doi.org/10.46338/ijetae0920_04
- Zakaria, H. E. A. (2010). Integration of remote sensing and GIS in studying vegetation trends and conditions in the Gum Arabic belt in North Kordofan, Sudan. MSc. thesis. University of Hamburg, Germany.
- Zarroug, I. M. A., Elaagip, A., Gumaa, S. G., Ali, A. K., Ahmed, A., Siam, H. A. M.,
 Abdelgadir, D. M., Surakat, O. A., Olamiju, O. J., Boakye, D. A., Aziz, N., & Hashim, K. (2019). Notes on distribution of Simulium damnosum s. l. along Atbara River in Galabat sub-focus, eastern Sudan. *BMC Infectious Diseases*, 19(1), 477. https://doi.org/10.1186/s12879-019-4113-1
- Zewdie, A. (2014). Impacts of climate change on food security: A literature review in sub Saharan Africa. *Journal of Earth Science & Climatic Change*, 05(08), 8–11. https://doi.org/10.4172/2157-7617.1000225
- Zhang, Z., Liu, X., Wang, P., Shuai, J., Chen, Y., Song, X., & Tao, F. (2014). The heat deficit index depicts the responses of rice yield to climate change in the northeastern three provinces of China. *Regional Environmental Change*, 14(1), 27–38. https://doi.org/10.1007/s10113-013-0479-6
- Zhou, P., Huang, J., Pontius, R. G., & Hong, H. (2014). Land classification and change intensity analysis in a coastal watershed of Southeast China. *Sensors*, *14*(7), 11640–11658. https://doi.org/10.3390/s140711640
- Zoderer, B. M., Tasser, E., Erb, K. H., Lupo Stanghellini, P. S., & Tappeiner, U. (2016). Identifying and mapping the tourists' perception of cultural ecosystem services: A case

study from an Alpine region. *Land Use Policy*, *56*, 251–261. https://doi.org/10.1016/j.landusepol.2016.05.004

Zurqani, H. A., Post, C. J., Mikhailova, E. A., Schlautman, M. A., & Sharp, J. L. (2018). Geospatial analysis of land use change in the Savannah River Basin using Google Earth Engine. *International Journal of Applied Earth Observation and Geoinformation*, 69, 175– 185. https://doi.org/10.1016/j.jag.2017.12.006

Appendices

Appendix 1: Questionnaire

General Information

Enumerator's Name:			
Date of Interview:			
Village:	Locality:		
Geographic coordinates:			
Respondent's Name:		Phone No:	
Number of years the respondent is liv	ing in the village:		

A. General Questions

1. Demographic and Socioeconomic Characteristics

Sex of the	Head of the	Marital	Age	Education	Experience in	Main	Land tenure	Land size	Family size	Estimated
respondent	household	status	(years)	Code D	agriculture	sources of	Code F	(ha)		annual
Codes A	Codes B	Code C			(years)	income				income
						Code E				

Code D

1 Illiterate

2 Primary

3 Secondary

4 Graduate

Codes A 0=Male 1=Female Codes B

0=Male

1=Female

Codes C 1 Single 2 Married 3 Separated 4 Divorced 5 Widowed 6 Refused to answer

Codes ECool1 Farming (crop +10livestock)2 R2 Salaried employment3 S3 Self-employed (4 Sbusiness, trade,4 Shandicraft)4 Selling of forestproduce (e.g. charcoal,firewood, timber, poles)5 Other, specify......

Code F 1Owned 2 Rent in 3 Share in

4 Share out

B. Climate Change Perception

1. Do you observe any changes in the rainfall pattern in last 20 years?

Yes () No ()

- 2. If yes, what are the changes you observed? (Multiple choices are possible)
 - a. Decrease of rainfall amount ()
- b. Increase of rainfall amount ()
- c. Late onset of rainy season ()
- d. False onset of rainy season ()

- e. Early end of rainy season () f. Rainfall fluctuations/erratic ()
- g. Frequency of droughts ()

h. Frequency of floods ()

j. Other (specify):_

i. Increased length of the dry spells ()

- 3. Have you faced drought over the last 20 years?
 - Yes () No () Don't know ()
- 4. If yes, how do you describe the frequency of occurrence of drought for the last 5-10 years as compared to the past 30 years?

```
Increased ( ) Decreased ( ) Followed a similar trend ( )
```

- 5. Have you faced a flooding problem over the last 20 years?
 - Yes () No () Don't know ()
- 6. If yes, how do you describe the frequency of occurrence of flood in last 5-10 years as compared to the past 30 years?

Increased () Decreased () Followed a similar trend ()

Do you observe any changes of temperature during summer season?
 Yes () No () Don't know ()

- 8. If yes, what are the changes you observed?
 - a. Increase of daytime temperature () b. Decrease of daytime temperature ()
 - c. Increase of nighttime temperature () d. Decrease of nighttime temperature ()
 - e. No change ()
- 9. Do you observed any changes of temperature during winter season?
 Yes () No () Don't know ()
- 10. If yes, what are the changes you observed?
 - a. Increase of daytime temperature ()
 b. Decrease of daytime temperature ()
 c. Increase of nighttime temperature ()
 d. Decrease of nighttime temperature ()
 e. No change ()

C. Adaptation Options

1. Do you think climate change has already happened?
Yes () No () Don't know ()
2. If the answer to Q.1 is yes, how did you feel that on your farming activities?
a. Positively affected () How?
b. Negatively affected () How?
3. How do you evaluate the trend of crop production for the last 30 years?
a. Increasing () b. Decreasing () c. The same () d. I do not now ()
4. What alternative measures do you take when enough rain is not available for your farm?
5. According to your view, are those measures effective enough to save crops? Explain?

6. Does the local government provide any support in facing challenges related to climate change?

Yes () No ()

No	Adaptation options	Mark
a	Early cultivation	
b	Delayed cultivation	
с	Cultivation of short maturing varieties	
d	Intercropping	
e	Crop rotation	
f	Soil conservation	

7. What are the agricultural adaptation measures you have taken to cope with climate change?

g	Practice supplementary irrigation	
h	Use of technologies as fertilizers and pesticides	
j	Introduce water stress tolerant varieties (specify)	
k	Rainwater harvesting	
1	Cover the soil around the plant with straw, stone, plastic and/or crop residues to facilitate water infiltration and decreased water evaporation	
m	Mixing farming	
n	Use of underground water	
0	Others (specify)	

8. Are you interested in using adaptation measures? Yes () No ()

9. Why? _____

10. Do you have access to climate information? Yes () No ()

11. If your answer for Q 10 is yes through which channel do you receive the climate information?

a. Government extension officers ()
 b. Radio () c. Mobile phone () d. Personal contact or social group () e. Research institutions/NGOs
 f. Development agents () g. another channel, specify _______

12. Have you got advice in agricultural activities from extension service? Yes () No ()

13. If the answer to Q 12 is yes on which area the advice was given?

a. Crop husbandry () b. Crop diversification () c. Animal husbandry ()

- d. Marketing () f. Post-harvest () g. Climatic information ()
 - h. Others, specify____

D. Land use/ land cover Perception

- 1. Did you observe changes in cropped areas? No () Yes ()
- 2. If your answer for Q 1 is yes, has crop production area declined or increased over the past 20 years in your community?
 - a. Declined () b. Increased ()
- 3. Has crop production declined or increased over the past 20 years in your community?
 - a. Declined () b. Increased () c. No change ()
- 4. If you indicated that crop production has declined, which, in your opinion, are the main reasons for this decline in crop production? (**Check the one that applies**)

Soil infertility	Lack of agricultural inputs	Pests and diseases	Inadequate labour	
Unreliable rainfall	Fluctuating markets/prices	Lack of knowledge and skills	Limited land	
Lack of improved seed	Lack of money for inputs	Other		
-				

5. Do you know of any forests in your area?

	No () Yes () Name them:
6.	If yes, how do you think these forests came into existence?
	a. Natural () b. Man-made () c. Both ()
7.	What is the current status of the forests in your area?
	a. Declining () b. Constant () c. Increasing ()
8.	If the answer is declining is it due to?
	a. Settlements () b. Population growth () c. Agricultural expansion ()
	d. Cutting () e. Forest fires () f. Grazing ()
	g. Lack of law enforcement () h. others, specify

9. Do you think the population of your community has increased over the past 20 years?

No () Yes ()

10. If yes, what do you think have caused the population increase?

a. High fertility ()
b. Immigration ()
c. Both high fertility and immigration ()
b. Other, specify _______

11. Do you think that more land will be needed as your family grows?

No () Yes ()

12. If yes, how much extra land do you think you will need when you have a new family member?

a. 0.5 acres () b. 1 acre () c. 2 acres () d. > 2 acres () e. Don't know ()

13. What kind of land would you clear when your family size increases?

- a. Forest () b. Fallow land () c. Grazing land ()
- c. Other (specify) _____

14. What do you think are the causes of land-use and land-cover changes in your area (rank on a scale of 1 to 5; 1 = least important and 5 = most important).

Proximate cause				Rank		
	1	2	3	4	5	
Firewood						
Charcoal production						
Timber						
Construction						
Agriculture expansion						
Bush fires						
Settlements						

Others (Specify)						
		Rank				
Underlying Causes	1	2	3	4	5	
Poverty						
Population growth						
Lack of financial resources						
Lack of law enforcement						
Demand for timber						
Others (Specify)						

Appendix 2: Key informant interviews

A. General Information

Respondent's name:	
Respondent's email:	
Respondent's organizatio	on:
Respondent's sex:	
Village:	

B. background Information

- 1. Respondent's Role (e.g. farmer, village leader, extension officer)
- 2. Main livelihood(s) in village
- 3. How many years have you lived in this area?
- 4. What is your age?

C. Changes in Climate

1. What are some of the biggest changes you have observed over the last few years?

2. Please place a check mark next to the changes in weather, climate and extreme events that took place. Check all that apply

Increased rainfall	
Decreased rainfall	
Changes in timing of seasons	
Drought	
Loss of water source	
Flooding	
Heat waves/hotter days	
Cold spells	
Wildfires	
Changes in wind	
Erosion/landslides	
Other (please specify)	
None	

D. Impacts and Responses

1. How have the changes in weather you mentioned above impacted your main livelihood?

2. How have people in the area responded to these impacts?

3.	Which of these responses is working well and which isn't?
4.	How have the changes in climate you mentioned affected natural resources used by
	the community?

5. Please place a check mark next to the change in activities in response to changes in weather and climate that have mentioned above?

Crop practices	
Livestock practices	
Livelihood type	
Livelihood location	
Water management	
Disease/Pest management	
Natural resource use	
Natural habitat encroachment	
Land conversion	
Migration	
Other (please specify)	
None	

Appendix 3: Checklist for focus group discussions

Background information of respondent:

- Income level
- Household size
- Head of the household
- Farm size
- Rainfall and temperature variability
- Timelines and trend lines
- Current exposure sensitivities Climatic risks:
- Temperature changes
- Rainfall changes
- Droughts
- Perception(s) towards major climatic hazards
- Biophysical/Social/Economic
- Water and climate
- Historical data (when and how exposed)

Current adaptive/response strategies Present adaptive practices/ responses:

- Formal and Local response mechanisms
- How exposures are managed
- What constrains adaptive strategies

Land use and land cover change:

- Changes in cropped areas
- Crop production declined or increased over the past 20 years in the community
- If crop production has declined, what are the main reasons for this decline in crop production
- Forests in the area and from where these forests came into existence
- The current status of the forests in the area
- If the answer is the forests declining, what is it due to
- The status of the population in the community
- The causes of land-use and land-cover changes in the area